SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT
PURSUANT TO SECTION 13 OR 15(d) OF
THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest reported): **September 5, 2002**

ALLIANCE DATA SYSTEMS CORPORATION

(Exact name of registrant as specified in its chapter)

DELAWARE(State or Other Jurisdiction of Incorporation)

001-15749 (Commission File Number)

31-1429215 (IRS Employer Identification No.)

17655 WATERVIEW PARKWAY DALLAS, TEXAS 75252

(Address and Zip Code of Principal Executive Offices)

(972) 348-5100

(Registrant's telephone number, including area code)

NOT APPLICABLE

(Former name or former address, if changed since last report)

ITEM 2. ACQUISITION OR DISPOSITION OF ASSETS

On September 5, 2002, Alliance Data Systems Corporation ("Alliance") acquired Enlogix Group from Duke Energy Corporation for approximately \$13.5 million. Enlogix Group provides customer information system services to utilities in Canada. In the event that Enlogix Group enters into agreements with three potential customers within eight months of closing of the acquisition, the purchase price will increase by up to \$4.5 million. The purchase price was paid using Alliance's own funds. Alliance intends to continue to operate Enlogix Group as previously operated.

ITEM 7. FINANCIAL STATEMENTS, PRO FORMA FINANCIAL INFORMATION AND EXHIBITS

The following financial statements, pro forma information and exhibits are being provided as part of this report in accordance with the instructions to this item.

(a) FINANCIAL STATEMENTS OF BUSINESSES ACQUIRED.

The combined financial statements of Enlogix Group as of and for the years ended December 31, 2000 and 2001, together with the notes thereto, are incorporated herein by reference to Exhibit 99.1 of this Current Report on Form 8-K.

(b) PRO FORMA FINANCIAL INFORMATION.

The pro forma condensed financial information for Alliance and Enlogix Group, for the periods reflected therein, is set forth in Exhibit 99.2 of this Current Report on Form 8-K.

(c) EXHIBITS.

- 2.1 Purchase & Sale Agreement, dated September 5, 2002, among ADS Alliance Data Systems, Inc., Loyalty Management Group Canada, Inc. and Westcoast Energy, Inc. carrying on business as Duke Energy Gas Transmission.
- 23.1 Consent of Deloitte & Touche LLP.
- 99.1 Combined financial statements of Enlogix Group as of and for the years ended December 31, 2000 and 2001.
- 99.2 Pro forma condensed financial information for Alliance and Enlogix Group.

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ALLIANCE DATA SYSTEMS CORPORATION

Dated: September 5, 2002 By: /s/ EDWARD J. HEFFERNAN

Edward J. Heffernan Executive Vice President and Chief Financial Officer

QuickLinks

SIGNATURES

WESTCOAST ENERGY INC.

carrying on business as

DUKE ENERGY GAS TRANSMISSION

- and -

ADS ALLIANCE DATA SYSTEMS, INC.

- and -

LOYALTY MANAGEMENT GROUP CANADA INC.

PURCHASE AND SALE AGREEMENT

September 5, 2002

TABLE OF CONTENTS

				Page
ARTICL	E 1 IN	ΓERPRETA	ATION	1
	1.1	Definitio		1
	1.2	Schedule	es	11
	1.3	Headings	s and Table of Contents	11
	1.4	Gender a	nd Number	11
	1.5	Currency	T.	11
	1.6	Generally	y Accepted Accounting Principles	12
	1.7	Invalidity	y of Provisions	12
	1.8	Entire A	greement	12
	1.9	Waiver, A	Amendment	12
	1.10	Governir	ng Law	12
ARTICL	.E.2 PU	RCHASE A	AND SALE	12
THUTCH	2.1		ent to Purchase and Sell and Purchase Price	12
	2.2	U	of Purchase Price and Delivery of Share Certificates and Evidence of Transfer of the	
		-	d Partnership Interest	13
	2.3		on of Closing Date Balance Sheet	13
	2.4		Capital Adjustment	14
	2.5		Price Adjustments and Refunds	14
	2.6		on to Obtain Contracts	15
	2.7	Oracle A	rrangement	15
	2.8	Allocatio	on of Purchase Price Adjustments	16
A DELCE			ATTONS AND LIADD INTERS	4-
ARTICL			ATIONS AND WARRANTIES	17
	3.1	By the V		17
		3.1.1	Incorporation and Status of the Vendor	17
		3.1.2	Corporate Power of the Vendor and Due Authorization	17
		3.1.3 3.1.4	Incorporation and Status of the Corporation and the Subsidiary	17 17
		3.1.5	Formation and Status of Enlogix CIS L.P. Power and Capacity	18
		3.1.6	Capital of the Corporation and the Subsidiary	18
		3.1.7	Partnership Interests of Enlogix CIS L.P.	18
		3.1.8	Subsidiaries	18
		3.1.9	Reorganization	18
		3.1.10	No Obligations to Issue Securities	19
		3.1.11	Title to, and Right to Sell, Purchased Shares	19
		3.1.12	Title to, and Right to Sell, Purchased Partnership Interest	19
		3.1.13	Bankruptcy	19
		3.1.14	No Contravention	19
		3.1.15	Approvals and Consents of Vendor	20
		3.1.16	Financial Statements	20
		3.1.17	Liabilities	20
		3.1.18	Guarantees	20
		3.1.19	Indebtedness	20
		3.1.20	Absence of Unusual Transactions and Events	21

	3.1.22	Non-Arm's Length Transactions	22
	3.1.23	Intercompany Services	23
	3.1.24	Material Contracts	23
	3.1.25	Customers and Suppliers	23
	5.1 .2 5	Gustomers und Suppliers	
		i	
	2.4.20		
	3.1.26	Commitments Under the IBM Lease Agreement	23
	3.1.27	No Default Under Agreements	24
	3.1.28	Employment Matters	24
	3.1.29	Independent Contractors	25
	3.1.30	Collective Agreements	25
	3.1.31	Real Property	25
	3.1.32	Leased Premises	25
	3.1.33	Environmental Compliance	26
	3.1.34	Title and Sufficiency of Assets	26
	3.1.35	Assets in Good Condition	26
	3.1.36	Personal Property Leases	27
	3.1.37	Receivables	27
	3.1.38	Tax Matters, etc.	27
	3.1.39	Employee Plans	29
	3.1.40	Insurance	31
	3.1.41	Intellectual Property and Software	32
	3.1.42	Data Processing	36
	3.1.43	Permits and Registrations	36
	3.1.44	Bank Accounts/Powers of Attorney	37
	3.1.45	Compliance with Laws	37
		•	
	3.1.46	Litigation and Other Proceedings	37
	3.1.47	Corporate Records	37
	3.1.48	Vendor Resident of Canada	37
	3.1.49	Full Disclosure	37
3.2	-	urchasers	37
3.3	No Finde		39
3.4		of Representations, Warranties and Covenants	39
	3.4.1	Vendor's Representations and Warranties	39
	3.4.2	Purchasers' Representations and Warranties	40
	3.4.3	Covenants	40
A DELCT E 4			4.0
	CONDITION		40
4.1		ns for the Benefit of the Purchasers	40
	4.1.1	Closing Documents and Proceedings	40
	4.1.2	Delivery of Share Certificates and Evidence of Transfer of the Purchased Partnership	
		Interest	41
	4.1.3	Ancillary Agreements	41
	4.1.4	Release from the Vendor and its Affiliates	41
	4.1.5	No Litigation	41
	4.1.6	Competition Act	41
	4.1.7	Consents and Approvals	41
	4.1.8	Oracle Assignment	42
	4.1.9	Westcoast Oracle Certificate	42
4.2		ons for the Benefit of the Vendor	42
7.2	4.2.1	Payment of Purchase Price	42
	4.2.2	Closing Documents and Proceedings	42
	4.2.3	Ancillary Agreements	42
	4.2.4	No Litigation	42
	4.2.5	Competition Act	43
	4.2.6	Consents and Approvals	43
		ii	
		L COVENANTS AND AGREEMENTS OF THE PARTIES	43
5.1		e Action and Resignations	43
5.2	Bank Acc		43
5.3	Undertak	ing to Discharge Security Interests	43
5.4		nt of Loan	43
5.5		ion of Employees and Consulting Arrangements	44
5.6		g of Consents and Approvals	44
5.7		ion of Intercompany Services and Transfer of Assets and Licences	44
5.8		nents with respect to Support Obligations	45
5.9		rns for Closing Year	45
5.10			
		Not to Compete	46
5.11	Non-Solid		49
5.12	Confiden	Hallly	50

3.1.21

Certain Payments

	5.13	Remedies	50
	5.14	Reasonableness of Restrictions	51
ARTIC	CLE 6 IN	NDEMNIFICATION	51
	6.1	Indemnification by the Vendor	51
	6.2	Indemnification by the Purchasers	52
	6.3	Notice of Claim	53
	6.4	Defence of Third Party Claims	53
	6.5	Assistance for Third Party Claims	54
	6.6	Additional Rules and Procedures	54
	6.7	Tax Assessment Procedure	56
		6.7.1 Pre-Assessment Procedure	56
		6.7.2 Procedure on Assessment	56
		6.7.3 Objection/Appeal Procedure	56
	6.8	Indemnification Claims	56
	6.9	Duty to Mitigate	56
	6.10	Right of Indemnification Not Affected by Knowledge	56
ARTIC	CLE 7 C	LOSING	58
	7.1	Location and Time of the Closing	58
	7.2	Deliveries at the Closing	58
ARTIC	CLE 8 G	ENERAL MATTERS	59
	8.1	Confidentiality	59
	8.2	Public Notices	59
	8.3	Expenses	59
	8.4	Assignment	59
	8.5	Notices	59
	8.6	Time of Essence	60
	8.7	Further Assurances	61
	8.8	Counterparts	61

PURCHASE AND SALE AGREEMENT

iii

THIS AGREEMENT is made as of this 5th day of September, 2002,

BETWEEN:

WESTCOAST ENERGY INC., a corporation amalgamated under the laws of Canada, carrying on business as Duke Energy Gas Transmission

(the "Vendor")

– and –

ADS ALLIANCE DATA SYSTEMS, INC., a corporation incorporated under the laws of Delaware

("ADS")

- and -

LOYALTY MANAGEMENT GROUP CANADA INC., a corporation incorporated under the laws of Ontario

(individually, "LMGC", and collectively with ADS, the "Purchasers")

RECITALS:

- A. The Vendor wishes to sell all of the issued and outstanding shares in the capital of Enlogix Inc. and ADS wishes to purchase such shares, on and subject to the terms and conditions of this Agreement.
- B. The Vendor also wishes to sell all of its interest in Enlogix CIS L.P., a limited partnership formed under the laws of the Province of Alberta, and LMGC wishes to purchase such interest, on and subject to the terms and conditions of this Agreement.

NOW THEREFORE in consideration of the mutual covenants and agreements contained in this Agreement and other good and valuable consideration (the receipt and sufficiency of which are hereby acknowledged), the parties hereto agree as follows:

ARTICLE 1 INTERPRETATION

- 1.1.1 "**Act**" has the meaning attributed to such term in Section 4.1.6;
- 1.1.2 "Affiliate" means, with respect to any Person, any other Person who directly or indirectly controls, is controlled by, or is under direct or indirect common control with, such Person, and includes any Person in like relation to an Affiliate. A Person shall be deemed to control a Person if such Person possesses, directly or indirectly, the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise; and the term "controlled" shall have a similar meaning;
- 1.1.3 "Agreement" means this agreement and all exhibits and schedules attached to this agreement, in each case as they may be amended or supplemented from time to time, and the expressions "hereof", "herein", "hereto", "hereunder", "hereby" and similar expressions refer to this agreement; and unless otherwise indicated, references to Articles and sections are to Articles and sections in this Agreement;
- 1.1.4 "Ancillary Agreements" means the Human Resources Agreement, the Transition Services Agreement, the Union Gas Amendment and the Sublease;
- 1.1.5 "Applicable Law" means, with respect to any Person, property, transaction, event or other matter, any law, rule, statute, regulation, order, judgment, decree or treaty, any legislative, administrative and judicial promulgation, applicable regulatory policy or other requirement having the force of law (collectively, the "Law") relating or applicable to such Person, property, transaction, event or other matter. Applicable Law also includes, where appropriate, any interpretation of the Law (or any part thereof) by any Person having jurisdiction over it, or charged with its administration or interpretation;
- 1.1.6 "Applicable Tax Legislation" means any legislation pursuant to which Governmental Charges and Taxes are imposed by any Taxation Authority;
- 1.1.7 "Application Software" has the meaning attributed to such term in Section 1.1.32;
- 1.1.8 "Assessment" means an assessment, reassessment or any other formal claim for, or in respect of, Governmental Charges and Taxes which is made by any Taxation Authority with respect to any pre-Closing period;
- 1.1.9 "Assets" means all the properties, assets, interests and rights of each of the Purchased Entities, including the following:
 - (a) all rights and interests of the Purchased Entities to and in the Leased Premises and under the Premises Leases, including prepaid rents, security deposits and options to renew or purchase, rights of first refusal under the Premises Leases and all leasehold improvements owned by any of the Purchased Entities and forming part of the Leased Premises;
 - (b) the Personal Property;
 - (c) the Receivables;
 - (d) all rights and interests under or pursuant to all warranties, representations and guarantees, express, implied or otherwise, of or made by suppliers or others in connection with the Assets;
 - (e) the Intellectual Property;
 - (f) the Software;
 - (g) the BANNER Modifications;
 - (h) the Contracts;
 - (i) contracts of insurance;
 - (j) the Permits;
 - (k) the Books and Records;
 - (l) all prepaid charges, deposits, sums and fees paid by any of the Purchased Entities before the Time of Closing;
 - (m) all goodwill of each of the Purchased Entities, including the present telephone numbers, internet domain addresses and other communications numbers and addresses of each of the Purchased Entities; and
 - (n) all proceeds of any or all of the foregoing received or receivable after the Closing Date.

1.1.11	"BANNER Modifications" means all modifications, enhancements, corrections and improvements to, customizations of, and derivative				
	works based on, the SCT Software which have been developed by any of the Purchased Entities;				
1.1.12	"Benefit Plans" has the meaning attributed to such term in Section 1.1.34(c);				
1.1.13	"Books and Records" means all books, records, information, data, client lists, files and papers of each of the Purchased Entities including software manuals, sales and advertising materials, sales and purchases correspondence, trade association files, lists of present and former customers and suppliers, personnel, employment and other records, and the minute and share certificate books of each of the Purchased Entities, and all copies and recordings of the foregoing;				
1.1.14	"Business" means the provision of customer care and billing system outsourcing solutions for utilities and energy companies, through an application service provider and/or license model;				
1.1.15	"Business Day" means any day, other than Saturday, Sunday or any statutory holiday in the Province of Ontario or the State of Texas;				
1.1.16	"Certificate Withheld Amount" means a dollar amount to be calculated in accordance with Section 4.1.9.				
1.1.17	"Charge" means any security interest, lien, charge, hypothec, pledge, encumbrance, mortgage, option, adverse claim or title retention agreement of any nature or kind whatsoever;				
1.1.18	"CIS Software" means all Software used by the Purchased Entities in providing to any customer of the Business any services which a Purchased Entity is obligated to provide under any Contract with that customer;				
1.1.19	"Claims" means any and all claims, demands, actions, causes of action, damages, losses, costs, liabilities and expenses, including, without limitation, reasonable legal fees;				
1.1.20	"Closing" means the completion of the sale and purchase of the Purchased Interests pursuant to this Agreement at the Time of Closing;				
1.1.21	"Closing Date" means the date of this Agreement, being September 5, 2002;				
1.1.22	"Closing Date Balance Sheet" has the meaning attributed to such term in Section 2.3.1;				
1.1.23	"Code" means the Internal Revenue Code of 1986 (U.S.), as amended from time to time;				
1.1.24	"Commissioner" has the meaning attributed to such term in Section 4.1.6;				
1.1.25	"Confidential Information" has the meaning attributed to such term in Section 5.12;				
1.1.26	"Confidentiality Agreement" means the confidentiality agreement signed by the Vendor and the Purchaser, dated as of February 18, 2002;				
	3				
1.1.27	"Consulting Agreements" means: (a) the agreement to be entered into at or prior to the Closing between Anthony Haines and Enlogix CIS L.P. pursuant to which Anthony Haines will provide certain services to Enlogix CIS L.P. for purposes of concluding the contracts referred to in Section 2.5 and certain other related or ancillary purposes, which agreement shall be substantially in the form set out in Part A of Schedule 1.1.27; and (b) the agreement to be entered into at or prior to the Closing between Don Critchley and Enlogix CIS L.P. pursuant to which Don Critchley will provide certain services to Enlogix CIS L.P. for purposes of concluding the contracts referred to in Section 2.5 and certain other related or ancillary purposes, which agreement shall be substantially in the form set out in Part B of Schedule 1.1.27;				
1.1.28	"Contracts" means all contracts, agreements, leases and arrangements to which any of the Purchased Entities is a party or by which any of the Purchased Entities or any of the Assets or the Business is bound, including the Material Contracts, the Premises Leases and the Personal Property Leases;				
1.1.29	"Copyrights" has the meaning attributed to such term in Section 1.1.47(c);				
1.1.30	"Corporation" means Enlogix Inc., a corporation amalgamated under the laws of Canada;				
1.1.31	"Customer Oracle Software" has the meaning attributed to such term in Section 2.7.3;				
1.1.32	"Data Processing System" means the computer equipment and associated peripheral devices and the related operating systems and other				

"Employee Plans" means all written and material oral plans, arrangements, agreements, programs, grants, awards, policies or practices relating to some or all of the current or former directors, employees and officers of any of the Purchased Entities and their dependents under which any Purchased Entity has any actual or contingent liability, including those which provide for or relate to:

computer software subject to transferable licenses);

"Defending Party" has the meaning attributed to that term in Section 6.5;

1.1.33

1.1.34

bonus, retention, change of control, profit sharing or deferred profit sharing, performance compensation, deferred or incentive (a)

software owned, leased or licensed by any of the Purchased Entities (other than Software that is to be operated on such computer equipment and peripheral devices to perform specific user-oriented functions or applications ("Application Software") and off-the-shelf personal

- compensation, share compensation, share purchase or share option, share appreciation rights, phantom stock, employee loans, or any other compensation in addition to salary ("**Incentive Plans**");
- retirement or retirement savings, including, without limitation, pensions (registered or unregistered), supplemental pensions, registered
 retirement savings plans and retirement compensation arrangements ("Pension Plans"); or
- (c) insured or self-insured benefits for or relating to income continuation or other benefits during absence from work (including short term disability, long term disability and workers compensation or workplace safety insurance benefits), vacation, vacation pay, sick pay, hospitalization, health, welfare, medical, dental, life insurance, accident, death or survivor's benefits, supplementary employment insurance or other employment benefits ("Benefit Plans");

provided that nothing in the foregoing shall be construed to include statutory employee benefit plans to which any Purchased Entity is obliged to contribute;

1.1.35 "Environmental Laws" has the meaning attributed to such term in Section 3.1.33;

4

- 1.1.36 "ERISA Plan" means an Employee Plan that is an "employee benefit plan" as that term is defined in Section 3(3) of ERISA;
- 1.1.37 "Final Closing Working Capital Statement" has the meaning attributed to such term in Section 2.3.1;
- 1.1.38 "Financial Statements" means the annual audited balance sheets of each of the Purchased Entities, as at December 31, 2001 and the accompanying annual audited statements of earnings, retained earnings and changes in financial position, including the notes thereto, for the year then ended, and the interim unaudited balance sheets of each of the Purchased Entities as at July 31, 2002 and the accompanying interim unaudited statements of earnings, retained earnings and changes in financial position for the period ended July 31, 2002, copies of which are attached as Schedule 1.1.38;
- "Governmental Charges and Taxes" means all taxes, fees, levies, imposts, assessments, reassessments and other charges, including all income, sales, use, goods and services, value added, capital, capital gains, alternative, net worth, transfer, profits, withholding, payroll, employer health, excise, franchise, real property and personal property taxes, and any other taxes, customs duties, fees, assessments or similar charges in the nature of a tax including Canada Pension Plan and provincial pension plan contributions, employment insurance payments and workers' compensation or workplace safety insurance premiums, together with any instalments with respect thereto, and all related penalties, interest and fines, due and payable to any domestic or foreign government (federal, provincial, municipal or otherwise) or to any regulatory authority, agency, commission or board of any domestic or foreign government, or imposed by any court or any other law, regulation or rulemaking entity, whether disputed or not;
- 1.1.40 "**Human Resources Agreement**" means an agreement to be entered into at or prior to the Closing between the Vendor and the Purchasers providing for pension, employee benefits and employment matters as between the Vendor and the Purchasers substantially in the form set out in <u>Schedule 1.1.40</u>;
- 1.1.41 "IBM Lease Agreement" means the lease agreement between Enlogix CIS L.P. and IBM Canada Limited dated June 1, 2001;
- 1.1.42 "**Incentive Plans**" has the meaning attributed to such term in Section 1.1.34(a);
- 1.1.43 "including" means "including without limitation" and "includes" means "includes without limitation";
- 1.1.44 "**Indemnified Party**" means a person whom any party to this Agreement has agreed to indemnify under Articles 2 or 6;
- 1.1.45 "**Indemnifying Party**" means, in relation to an Indemnified Party, the party to this Agreement that has agreed to indemnify that Indemnified Party under Article 6;
- 1.1.46 "**Independent Auditor**" means PriceWaterhouseCoopers or such other nationally recognized accounting firm as may be agreed to by the Vendor and the Purchasers in writing;
- 1.1.47 "Intellectual Property" means:
 - (a) all Trade-Marks;
 - all inventions, patents, patent rights, patent applications (including all reissues, divisions, continuations, continuations-in-part and extensions of any patent or patent application), industrial designs and applications for registration of industrial designs used in the conduct of the Business (collectively, "Inventions");

- (c) all copyrights, registrations and applications for registration of copyrights used in the conduct of the Business (collectively, "Copyrights");
- (d) all rights and interests in and to processes, lab journals, notebooks, data, data bases, trade secrets, designs, know-how, product

formulae and information, engineering and other drawings and manuals, technology, blue prints, research and development reports, agency agreements, technical information, technical assistance, engineering data, design and engineering specifications, and similar materials recording or evidencing expertise or information used in the conduct of the Business (collectively, "**Know-How**");

- (e) all other intellectual and industrial property rights of each of the Purchased Entities, whether in Canada, the United States or elsewhere in the world that are used in the conduct of the Business; and
- (f) all rights and interests in licences from third parties of intellectual property listed in items (a) to (e) above;

provided that "**Intellectual Property**" excludes Software, but includes any Trade-Marks associated with Software, Inventions practiced or embodied in Software, Copyrights subsisting in Software and Know-How relating to Software;

- 1.1.48 "**Intercompany Services**" means the services and/or functions provided by or on behalf of the Vendor or its Affiliates (other than the Purchased Entities) to the Purchased Entities described in <u>Schedule 1.1.48</u>;
- 1.1.49 "**Inventions**" has the meaning attributed to such term in Section 1.1.47(b);
- 1.1.50 **"Know-How**" has the meaning attributed to such term in Section 1.1.47(d);
- 1.1.51 "Leased Premises" means all real property that is leased or occupied by any of the Purchased Entities under the Premises Leases;
- 1.1.52 "Liabilities" means all costs, expenses, charges, debts, liabilities, claims, demands and obligations, whether primary or secondary, direct or indirect, known or unknown, fixed, contingent, absolute or otherwise, including under or in respect of any contract, agreement, arrangement, lease, commitment or undertaking, Applicable Law and Governmental Charges and Taxes;
- 1.1.53 "Licensed Intellectual Property" means all Intellectual Property that is licensed to a Purchased Entity by a Person, other than another Purchased Entity;
- 1.1.54 "**Licensed Software**" means all Software that is licensed to a Purchased Entity by a Person other than another Purchased Entity, excluding BANNER Modifications, but including the Software identified in Parts C and E of Schedule 3.1.41;
- 1.1.55 "Limited Partnership Agreement" means the limited partnership agreement between Enlogix CIS Inc. (a company that has since been amalgamated with Enlogix Inc. to form the Corporation) and UEI Holdings Inc. (a company that has since been amalgamated with various entities to form the Vendor), dated June 1, 1998, as amended on December 20, 1999, and January 17, 2002;
- 1.1.56 "Material Adverse Effect", in respect of any of the Purchased Entities, means a material adverse effect on (a) the business operations, results of operations, affairs, financial or other condition, assets, properties or liabilities of any of the Purchased Entities, taken as a whole, or (b) the ability of the Purchased Entities to continue operating the Business in the manner it was conducted in the last twelve (12) months;

- 1.1.57 "Material Contract" means any contract, agreement or commitment (whether oral or written) to which any of the Purchased Entities is a party or by which any of the Purchased Entities or any of the Assets or the Business is bound that:
 - (a) expires or may be renewed at the option of any Person other than any of the Purchased Entities so as to expire more than two years after the date of this Agreement;
 - (b) is a trust indenture, mortgage, promissory note, loan agreement or other Contract for the borrowing of money, and any currency exchange, commodities or other hedging arrangement or any leasing transaction of the type required to be capitalized in accordance with generally accepted accounting principles (each, a "**Loan**") (excluding Loans having a principal amount of less than \$20,000 provided that the aggregate principal amount of all Loans does not exceed \$175,000);
 - (c) involves or may reasonably be expected to involve the payment to or by any of the Purchased Entities of more than \$175,000 over the term of the agreement (excluding any Contracts of employment disclosed in <u>Schedule 3.1.28</u> and contracts of indefinite hire terminable without cause on reasonable notice in accordance with Applicable Law);
 - is a confidentiality, secrecy or non-disclosure Contract (excluding, for greater certainty, service or customer Contracts that include an ordinary course non-disclosure clause);
 - (e) prohibits or restricts any of the Purchased Entities or their shareholders or general or limited partners, as applicable, from engaging in any line of business, compete with any other Person, operate its assets or otherwise conduct its business in any jurisdiction (excluding any Contracts that will not continue to be binding upon the Purchased Entities or its shareholders or general or limited partners from and after Closing);
 - is a Contract with a Person with whom any of the Purchased Entities does not deal at arm's length within the meaning of the *Income Tax Act* (Canada);
 - (g) is an agreement of guarantee, support, indemnification, assumption or endorsement of, or a similar commitment with respect to, the obligations, liabilities (whether accrued, absolute, contingent or otherwise) or indebtedness of any other Person (excluding (i) any such arrangements that are solely between the Purchased Entities, and (ii) any such arrangements relating to any indebtedness, liabilities or obligations (whether actual or potential) of less than \$20,000 provided that the aggregate amount of such indebtedness, liabilities or obligations does not exceed \$175,000);

- (h) is a Contract for Intercompany Services;
- (i) is a Contract made out of the Ordinary Course; or
- (j) is otherwise material to the operation of the Business;
- 1.1.58 "Net Working Capital" means the aggregate of: (i) current assets as shown on the Closing Date Balance Sheet, including cash and Receivables; minus (ii) current liabilities as shown on the Closing Date Balance Sheet, including accounts payable and accrued liabilities;
- 1.1.59 "**Oracle**" means Oracle Corporation Canada Inc.;
- 1.1.60 "**Oracle Assignment**" means an assignment and certification of non-possession document duly executed by the Vendor as assignor, by Enlogix CIS L.P. as assignee and by Oracle, which document shall be substantially in the form attached as <u>Schedule 4.1.8</u>;
- 1.1.61 "**Oracle Assignment Date**" means the date that is the earlier of: (a) sixty (60) days following the Closing Date; and (b) the date upon which the Oracle Assignment, as executed by the Vendor, Oracle and Enlogix CIS L.P., is delivered to the Purchasers;

- 1.1.62 "Oracle Software" means any software which Oracle or its Affiliates have made or make commercially available to third parties;
- 1.1.63 "Ordinary Course" means, with respect to an action taken by a Person, that such action is:
 - (a) consistent with the past practices of such Person and is taken in the ordinary course of the normal operations of such Person; and
 - (b) not required to be authorized by the board of directors, or executed or authorized in writing by the President, Chief Financial Officer or Chief Operating Officer, of such Person (or by any Person or group of Persons exercising similar authority);
- 1.1.64 "Partnership Interest Purchase Price" has the meaning attributed to that term in Section 2.1.2;
- 1.1.65 **"Pension Plans"** has the meaning attributed to such term in Section 1.1.34(b);
- 1.1.66 "**Permits**" has the meaning attributed to such term in Section 3.1.45;
- 1.1.67 "Person" means any individual, partnership, limited partnership, joint venture, syndicate, sole proprietorship, company or corporation with or without share capital, unincorporated association, trust, trustee, executor, administrator or other legal personal representative, regulatory body or agency, government or governmental agency, authority or entity however designated or constituted;
- 1.1.68 "Personal Property" means all machinery, equipment, furniture, motor vehicles and other chattels owned or leased by any of the Purchased Entities (including those in possession of third parties);
- 1.1.69 "Personal Property Leases" means all chattel leases, equipment leases, rental agreements, conditional sales contracts and other similar agreements relating to the Personal Property;
- 1.1.70 "**Premises Leases**" means all the leases, agreements to lease, subleases, licence agreements and occupancy or other agreements relating to the Leased Premises;
- 1.1.71 "**Prime Rate**" means, at any time, the rate of interest expressed as annual rate, established by Canadian Imperial Bank of Commerce in the City of Toronto, Ontario, Canada, from time to time as its reference rate of interest to determine the interest rates it will charge for loans in Canadian dollars to Canadian customers;
- 1.1.72 **"Proprietary Intellectual Property**" means all Intellectual Property used in the conduct of the Business other than Licensed Intellectual Property, and includes the Intellectual Property identified in Part A-1 of <u>Schedule 3.1.41</u>;
- 1.1.73 "**Proprietary Software**" means all Software identified in Parts D and H of <u>Schedule 3.1.41</u> used in the conduct of the Business, other than Licensed Software and BANNER Modifications;
- 1.1.74 "Purchase Price" means collectively, the Share Purchase Price and the Partnership Interest Purchase Price;
- 1.1.75 "Purchased Entities" means collectively the Corporation, the Subsidiary and Enlogix CIS L.P. and their successors;
- 1.1.76 "Purchased Interests" means collectively the Purchased Shares and the Purchased Partnership Interest;
- 1.1.77 **"Purchased Partnership Interest"** has the meaning attributed to that term in Section 2.1.2;
- 1.1.78 "Purchased Shares" has the meaning attributed to such term in Section 2.1.1;

"Purchasers' Counsel" means the firm of Blake, Cassels & Graydon LLP of Toronto, Ontario, or such other counsel as the Purchaser may 1.1.79 appoint with respect to this Agreement and the matters contemplated hereby; 1.1.80 "Receivables" means all accounts receivable, trade accounts and book debts of each of the Purchased Entities; "Related to the Business" means, directly or indirectly, used in, arising from or relating in any manner to the Business; 1.1.81 1.1.82 "**Reorganization**" has the meaning attributed to such term in Section 3.1.9; 1.1.83 "SCT" means (i) SCT Utility Systems, Inc., (ii) SCT Technologies Canada Inc., (iii) any Affiliate of either (i) or (ii); or (iv) any successor or assign of (i), (ii) or (iii); 1.1.84 "SCT Development Agreement" means the Software Development Services Agreement between Enlogix CIS L.P. and SCT Utility Systems, Inc. effective as of December 22, 2000; 1.1.85 "SCT Software" means all Software which has been licensed to or otherwise made available to a Purchased Entity by SCT prior to the Closing Date; 1.1.86 "Share Purchase Price" has the meaning attributed to that term in Section 2.1.1; 1.1.87 "Software" means all computer software programs, operating systems, and applications, firmware or software of any nature used in the conduct of the Business, whether operational, under development or inactive, including all object code, source code, technical manuals, user manuals and other documentation thereof, whether in machine-readable from, programming language or any other language or symbols, and whether stored, encoded, recorded or written on disk, tape, film, memory device, paper or other media of any nature, provided that "Software" excludes Intellectual Property; "Statements Date" means July 31, 2002; 1.1.88 1.1.89 "Sublease" means an agreement to be entered into at or prior to Closing between Union Gas Limited and Enlogix CIS L.P. setting out the terms of the sublease between Union Gas Limited, as sublessor, and Enlogix CIS L.P., as sublessee, relating to the office space on the 9th, 10th and 11th floors of the building located at the municipal address of 200 Yorkland Boulevard, Toronto, Ontario, which agreement shall be substantially in the form attached as Schedule 1.1.89; 1.1.90 "subsidiaries" has the meaning attributed to such term in the Canada Business Corporations Act as the same may be amended from time to time and any successor legislation thereto; 1.1.91 "Subsidiary" means Enlogix USA Inc., a corporation incorporated under the laws of the State of Delaware; 1.1.92 "**Support Obligations**" has the meaning attributed to that term in Section 5.8.2; 1.1.93 "Taxation Authority" means any government agency or authority which is entitled to impose Governmental Charges and Taxes or to administer any Applicable Tax Legislation and shall, for greater certainty, include any government authority, the federal government of Canada and of any country other than Canada and any provincial, state, county, municipal or other governmental subdivision within Canada or within any country other than Canada; 1.1.94 "Tax Returns" means all returns, reports, declarations, statements, bills, schedules or written information of, or in respect of, Governmental Charges and Taxes which are required to be filed with, or supplied to, any Taxation Authority; 9 1.1.95 "Territory" means Canada and the United States of America, including all of their respective provinces, states and territories; 1.1.96 "**Third Party**" has the meaning attributed to that term in Section 6.6.6; 1.1.97 "**Third Party Claim**" has the meaning attributed to such term in Section 6.3; "Time of Closing" means 10:00 a.m., Toronto time, on the Closing Date or such other time on the Closing Date as may be agreed upon in 1.1.98 writing by the parties; 1.1.99 "to the Vendor's knowledge" means the knowledge of any one of Morna Creedon, Greg Ebel, Anthony Haines, Don Critchley, Mark Board and John McKinnon, after: (a) in the case of Morna Creedon and Greg Ebel, due enquiry of all senior management of the Purchased Entities (comprising Anthony Haines, Don Critchley, Mark Board and John McKinnon) and after having reviewed such books, records and information in the possession of the Vendor relating to any of the Purchased Entities, the Assets, the Business and/or the Purchased Interests; and (b) in the case of Anthony Haines, Don Critchley, Mark Board and John McKinnon, due enquiry of all senior management of the Purchased Entities, and any employees or independent contractors that directly report to a member of the senior management team of the Purchased Entities, that are responsible for the applicable area, and after having reviewed such books, records, and information in the possession of any of the Purchased Entities relating to any of the Purchased Entities, the Assets, the Business and/or the Purchased Interests; 1.1.100 "Trade-Marks" means all registered or unregistered trade-marks, trade names, business names, brand names, brands, designs, logos, identifying indicia and service marks of each of the Purchased Entities used in the conduct of the Business including any goodwill attaching

to any such trade-marks and all registrations and applications relating thereto;

- "Transition Services Agreement" means an agreement to be entered into at or prior to the Closing between the Purchasers, the Vendor, the Purchased Entities and Union Gas Limited pursuant to which the Vendor and Union Gas Limited will provide certain services to the Purchased Entities on a transitional basis, which agreement shall be substantially in the form set out in Schedule 1.1.101;
 "Union Gas Amendment" means an agreement to be entered into at or prior to Closing between Union Gas Limited and Enlogix CIS L.P. providing for certain amendments to the master services agreement between Union Gas Limited and Enlogix CIS L.P. dated January 1, 2002, which agreement shall be substantially in the form attached as Schedule 1.1.102;
 - 1.1.103 "**Vendor's Counsel**" means the firm of Torys LLP of Toronto, Ontario, or such other counsel as the Vendor may appoint with respect to this Agreement and the matters contemplated hereby;
 - 1.1.104 "WARN Act" means the Worker Adjustment and Retraining Notification Act of 1988 (U.S.), as amended from time to time;
 - 1.1.105 "Westcoast Certificate Date" means the date that is the earlier of: (a) sixty (60) days following the Closing Date; and (b) the date upon which the Westcoast Oracle Certificate is delivered to the Purchasers; and
 - 1.1.106 "Westcoast Oracle Certificate" means a certificate of the Vendor, on its own behalf and on behalf of its affiliates Union Gas Limited and Pacific Northern Gas Ltd., duly executed by an officer of the Vendor and acknowledged by and agreed to by an officer of Union Gas Limited and Pacific Northern Gas Ltd., respectively, substantially in the form of Schedule 1.1.106;

1.2 Schedules

The following are the schedules attached to this Agreement:

Schedule 1.1.27	_	Consulting Agreements
Schedule 1.1.38	-	Financial Statements
Schedule 1.1.40	-	Human Resources Agreement
Schedule 1.1.48	-	Intercompany Services
Schedule 1.1.89	-	Sublease
Schedule 1.1.101	-	Transition Services Agreement
Schedule 1.1.102	-	Union Gas Amendment
Schedule 1.1.106	-	Westcoast Oracle Certificate
Schedule 3.1.3	-	Information on Corporation and Subsidiary
Schedule 3.1.6	-	Capital of the Corporation and Subsidiary
Schedule 3.1.7	_	Partnership Interests of Enlogix CIS L.P.
Schedule 3.1.9	_	Reorganization
Schedule 3.1.15	_	Approvals and Consents of the Vendor
Schedule 3.1.17	_	Liabilities
Schedule 3.1.20	_	Unusual Transactions and Events
Schedule 3.1.21	_	Certain Payments
Schedule 3.1.22	_	Non-Arm's Length Transactions
Schedule 3.1.24	_	Material Contracts
Schedule 3.1.25	_	Customers and Suppliers
Schedule 3.1.27	_	Defaults Under Agreements
Schedule 3.1.28	_	Employment Matters
Schedule 3.1.29	_	Independent Contractors
Schedule 3.1.32	_	Leased Premises
Schedule 3.1.34	-	Assets
Schedule 3.1.36	-	Personal Property Leases
Schedule 3.1.39	_	Employee Plans
Schedule 3.1.40	_	Insurance
Schedule 3.1.41	_	Intellectual Property
Schedule 3.1.43	-	Permits
Schedule 3.1.44	-	Bank Accounts/Powers of Attorney
Schedule 3.1.46	_	Litigation
Schedule 4.1.8	_	Oracle Assignment
Schedule 5.8.2	_	Support Obligations

1.3 Headings and Table of Contents

The inclusion of headings and a table of contents in this Agreement are for convenience of reference only and shall not affect the construction or interpretation hereof.

1.4 Gender and Number

In this Agreement, unless the context otherwise requires, words importing the singular include the plural and *vice versa* and words importing gender include all genders.

1.5 <u>Currency</u>

Except where otherwise expressly provided, all amounts in this Agreement are stated and shall be paid in the lawful currency of the United States of America.

1.6 Generally Accepted Accounting Principles

In this Agreement, except to the extent otherwise expressly provided, references to "generally accepted accounting principles" mean, for all principles stated in the Handbook of the Canadian Institute of Chartered Accountants, such principles so stated, consistently applied.

1.7 Invalidity of Provisions

Each of the provisions contained in this Agreement is distinct and severable and a declaration of invalidity or unenforceability of any such provision or part thereof by a court of competent jurisdiction shall not affect the validity or enforceability of any other provision hereof. To the extent permitted by applicable law, the parties waive any provision of law which renders any provision of this Agreement invalid or unenforceable in any respect.

1.8 Entire Agreement

This Agreement and the Ancillary Agreements constitute the entire agreement between the parties pertaining to the subject matter of this Agreement. There are no warranties, conditions, or representations (including any that may be implied by statute) and there are no agreements in connection with such subject matter except as specifically set forth or referred to in this Agreement or any Ancillary Agreement. No reliance is placed on any warranty, representation, opinion, advice or assertion of fact made either prior to, contemporaneous with, or after entering into this Agreement, or any amendment or supplement thereto, by any party to this Agreement or its directors, officers, employees or agents, except to the extent that the same has been reduced to writing and included as a term of this Agreement or any of the Ancillary Agreements, and none of the parties to this Agreement has been induced to enter into this Agreement or any of the Ancillary Agreements, or any amendment or supplement thereto by reason of any such warranty, representation, opinion, advice or assertion of fact. Accordingly, there shall be no liability, either in tort or in contract, assessed in relation to any such warranty, representation, opinion, advice or assertion of fact, except to the extent contemplated above.

1.9 Waiver, Amendment

Except as expressly provided in this Agreement, no amendment or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby. No waiver of any provision of this Agreement shall constitute a waiver of any other provision nor shall any waiver of any provision of this Agreement constitute a continuing waiver unless otherwise expressly provided.

1.10 Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the Province of Ontario and the laws of Canada applicable therein. The parties hereto hereby irrevocably attorn and submit to the non-exclusive jurisdiction of the Courts of the Province of Ontario.

ARTICLE 2 PURCHASE AND SALE

2.1 Agreement to Purchase and Sell and Purchase Price

- 2.1.1 Subject to the terms of this Agreement, at the Closing the Vendor hereby sells, and ADS hereby purchases, 3,749 common shares in the capital of the Corporation (the "**Purchased Shares**"), constituting all of the issued and outstanding common shares in the capital of the Corporation, for an aggregate price (the "**Share Purchase Price**") of \$382,500, subject to adjustment pursuant to Sections 2.4, 2.5 and 2.7.
- 2.1.2 Subject to the terms of this Agreement, at the Closing the Vendor hereby sells and LMGC hereby purchases, 100% of the Vendor's limited partnership interest in Enlogix CIS L.P. (the "**Purchased Partnership Interest**") for an aggregate purchase price (the "**Partnership Interest Purchase Price**") of \$13,117,500, subject to adjustment pursuant to Sections 2.4, 2.5 and 2.7.

12

2.2 Payment of Purchase Price and Delivery of Share Certificates and Evidence of Transfer of the Purchased Partnership Interest

- 2.2.1 The Share Purchase Price less one percent (1%) of the Assignment Withheld Amount and one percent (1%) of the Certificate Withheld Amount shall be paid in full at the Closing by ADS by delivery of a negotiable cheque certified by a Canadian chartered bank or Canadian trust company or an official bank draft drawn on a Canadian chartered bank or a wire transfer of funds or as otherwise agreed to by the Vendor and ADS in writing, in the amount of the Share Purchase Price, payable to or to the order of the Vendor against delivery by the Vendor of share certificates representing the Purchased Shares duly endorsed in blank for transfer or accompanied by duly signed powers of attorney for transfer in blank.
- 2.2.2 The Partnership Interest Purchase Price less ninety-nine percent (99%) of the Assignment Withheld Amount and ninety-nine percent (99%) of the Certificate Withheld Amount shall be paid in full at the Closing by LMGC by delivery of a negotiable cheque certified by a Canadian chartered bank or Canadian trust company or an official bank draft drawn on a Canadian chartered bank or a wire transfer of funds or as otherwise agreed to by the Vendor and LMGC in writing, in the amount equal to the Partnership Interest Purchase Price minus the Assignment Withheld Amount and the Certificate Withheld Amount, payable to or to the order of the Vendor against delivery by the Vendor of evidence satisfactory to LMGC of the transfer of the Purchased Partnership Interest.
- 2.2.3 If applicable, the Assignment Withheld Amount shall be paid in accordance with Section 2.7.1 and the Certificate Withheld Amount shall be paid in accordance with Section 2.7.2.

2.3 Preparation of Closing Date Balance Sheet

- 2.3.1 As soon as practicable after the Time of Closing, and in any event within 60 days after the Closing Date, the Purchasers shall prepare and deliver (such delivery to be from the Vice-President, Finance) to the Vendor an unaudited combined balance sheet of the Purchased Entities as at the Closing Date (the "Closing Date Balance Sheet"), together with a statement (the "Final Closing Working Capital Statement") setting out the Net Working Capital as at the Closing Date, all prepared in accordance with generally accepted accounting principles applied on a basis consistent with the Financial Statements.
- 2.3.2 The Closing Date Balance Sheet and the Final Closing Working Capital Statement shall be final and binding upon the Parties unless before the fifteenth Business Day after the date on which the Closing Date Balance Sheet and the Final Closing Working Capital Statement are delivered to the Vendor, the Vendor gives to the Purchasers a notice of objection to any matter stated in the Closing Date Balance Sheet or the Final Working Capital Statement.
- 2.3.3 If the Vendor gives a notice of objection pursuant to Section 2.3.2, then the Vendor and the Purchasers shall consult with each other with respect to the objection. If they are unable to reach agreement within 10 Business Days after the notice of objection has been given, then the dispute shall be referred to the Independent Auditor. The Independent Auditor shall be requested to deliver its determination of the amount of Net Working Capital within 20 Business Days following referral of the matter to the Independent Auditor and the decision of the Independent Auditor as to the Net Working Capital shall be final and binding on both parties. The fees and expenses of the Independent Auditor shall be borne fifty percent (50%) by the Vendor and fifty percent (50%) by the Purchasers.

2.4 Working Capital Adjustment

Within five (5) Business Days of the agreement of the Parties or the final determinations of the Independent Auditor on the Final Closing Working Capital Statement, as the case may be, (i) if the Net Working Capital as at the Closing Date exceeds negative Cdn\$1,834,594, then the Purchase Price shall be increased by the U.S. dollar equivalent of such difference and the Purchasers shall pay to the Vendor by certified cheque, bank draft or wire transfer an amount equal to such difference, and (ii) if the Net Working Capital as at the Closing Date is less than negative Cdn\$1,834,594, then the Purchase Price shall be decreased by the U.S. dollar equivalent of such difference and the Vendor shall pay to the Purchasers by certified cheque, bank draft or wire transfer an amount equal to such difference. For purposes of this Section 2.4, the "U.S. dollar equivalent" shall be the amount of United States dollars that, where clause (i) applies, the Purchasers could purchase, and where clause (ii) applies the Vendor could purchase, with the amount of Canadian dollars determined in accordance with this section, from a major financial institution at the U.S.\$/Cdn.\$ rate of exchange quoted by the institution at the United States dollars are purchased.

2.5 Purchase Price Adjustments and Refunds

- 2.5.1 In the event that, within the eight (8) month period following the Closing Date, Enlogix CIS L.P. enters into a contract with Manitoba Hydro pursuant to which Enlogix CIS L.P. agrees to sublicence certain CIS Software to Manitoba Hydro then the Purchase Price shall be increased by \$500,000, payable within fifteen (15) days after the date on which the contract between Enlogix CIS L.P. and Manitoba Hydro is executed by both parties thereto.
- 2.5.2 In the event that, within the eight (8) month period following the Closing Date, Enlogix CIS L.P. or one of its Affiliates enters into a contract with El Paso Electric Company pursuant to which Enlogix CIS L.P. or one of its Affiliates agrees to provide customer care and billing system services to El Paso Electric Company through an application service provider model, then the Purchase Price shall be increased by \$2,000,000, payable within 15 days after the date on which the contract between El Paso Electric Company and Enlogix CIS L.P. (or one of its Affiliates) is executed by both parties thereto.
- 2.5.3 In the event that, within the eight (8) month period following the Closing Date, Enlogix CIS L.P. or one of its Affiliates enters into a contract with Orlando Utilities Commission pursuant to which Enlogix CIS L.P. or one of its Affiliates agrees to provide customer care and billing system services to Orlando Utilities Commission through an application service provider model, then the Purchase Price shall be increased by \$2,000,000, payable within 15 days after the date on which the contract between Enlogix CIS L.P. (or one of its Affiliates) and Orlando Utilities Commission is executed by both parties thereto.
- 2.5.4 In the event that the aggregate Purchase Price (as finally adjusted in accordance with the other terms of this Agreement and translated into Canadian dollars based on the Bank of Canada noon spot rate in effect on the date each payment is made) paid by ADS and LMGC exceeds the undepreciated capital cost to Enlogix CIS L.P. of its depreciable property of all classes for Canadian federal and provincial income tax purposes finally delivered by the Vendor pursuant to Section 3.1.38(f), the Vendor shall be obligated to pay to LMGC (as to 99%) and ADS (as to 1%), as a partial refund of the Purchase Price, an amount equal to 35.5% of such excess. Such amount shall be due the earlier of 30 days after the end of the eight (8) month period following the Closing Date and the date that the final adjustment to the Purchase Price is paid pursuant to Section 2.5.1, 2.5.2 or 2.5.3.

14

2.6 Obligation to Obtain Contracts

- 2.6.1 The Purchasers, in consultation and cooperation with the Vendor, shall use commercially reasonable efforts to negotiate in good faith and settle, as soon as practicable following the Closing Date, the contracts referred to in Section 2.5 and promptly thereafter to cause the relevant Purchase Entity to enter into the relevant contract.
- 2.6.2 In connection with their obligations under Section 2.6.1, the Purchasers shall:
 - (a) inform the Vendor on a regular basis, and when requested by the Vendor, as to the status of the negotiations under Section 2.6.1;
 - (b) provide to the Vendor on a timely basis copies of all correspondence, draft agreements, memoranda and other materials relevant to the status of such negotiations;
 - (c) permit the Vendor a reasonable opportunity to attend any negotiations concerning the contracts, and provide timely advance notice of any

(d) in the context of acting in a commercially reasonable manner, not take any action or position that would reasonably be likely to adversely affect the Purchased Entities entering into any of the contracts in Section 2.5 provided that in no event shall the Purchasers be obligated to enter into a contract that is not on commercially reasonable terms.

2.7 Oracle Arrangement

- 2.7.1 In the event Oracle Assignment is delivered to the Purchasers within a sixty (60) day period following the Closing Date, the Assignment Withheld Amount shall be payable by the Purchasers to the Vendor within three (3) Business Days following the delivery of the Oracle Assignment. In the event that the Oracle Assignment is not delivered to the Purchasers within the sixty (60) day period following the Closing Date, then the Partnership Interest Purchase Price shall be reduced by ninety-nine percent (99%) of the Assignment Withheld Amount, the Share Purchase Price shall be reduced by one percent (1%) of the Assignment Withheld Amount, and the Purchasers shall have no obligation to pay the Assignment Withheld Amount to the Vendor.
- 2.7.2 In the event the Westcoast Oracle Certificate is delivered to the Purchasers within a sixty (60) day period following the Closing Date, the Certificate Withheld Amount shall be payable by the Purchasers to the Vendor within three (3) Business Days following the delivery of the Westcoast Oracle Certificate. In the event that the Westcoast Oracle Certificate is not delivered to the Purchasers within the sixty (60) day period following the Closing Date, then the Partnership Interest Purchase Price shall be reduced by ninety-nine percent (99%) of the Certificate Withheld Amount, the Share Purchase Price shall be reduced by one percent (1%) of the Certificate Withheld Amount, and the Purchasers shall have no obligation to pay the Certificate Withheld Amount to the Vendor.

15

- 2.7.3 The Vendor shall indemnify and save each of the Purchasers and their respective directors, officers, employees, agents, representatives and Affiliates (including, without limitation, the Purchased Entities), harmless from and against any and all Claims which may be made or brought against an Indemnified Party of which it may suffer or incur directly or indirectly as a result of, in respect of or arising out of: (i) any Liability arising from the use by a Purchased Entity of any Oracle Software up to and including the Oracle Assignment Date; and (ii) any Liability arising from a failure or an omission by Enlogix CIS L.P. to obtain from Oracle or from a customer of Enlogix CIS L.P. on the date hereof all necessary agreements, licences, consents, authorizations or permissions in respect of any Oracle Software which the applicable customer has furnished to Enlogix CIS L.P. prior to the Closing Date ("Customer Oracle Software") to enable Enlogix CIS L.P. to provide the required services to such customer during the lesser of (i) the current term of the applicable agreement between Enlogix CIS L.P. and such customer, and (ii) twenty-eight (28) months following the Closing Date.
- 2.7.4 In addition to the indemnification obligations of the Vendor set forth in Section 2.7.3, the Vendor agrees that if, during the lesser of (i) the current term of the applicable agreement between Enlogix CIS L.P. and its customer and (ii) twenty-eight (28) months following the Closing Date, Oracle determines, or such customer determines acting reasonably, that additional licences or licences of a different type are required from Oracle by or for Enlogix CIS L.P. and/or such customer in order for Enlogix CIS L.P. to continue to provide the services then being provided to such customer and as contemplated by the applicable agreement between Enlogix CIS L.P. and such customer, Enlogix CIS L.P. shall be entitled, after notifying the Vendor in advance, to purchase or otherwise procure all such licences and the Vendor shall promptly reimburse Enlogix CIS L.P. for its costs and expenses associated therewith.
- 2.7.5 In no event shall the aggregate amount of the indemnification obligations of the Vendor set forth in Section 2.7.3(ii) and the reimbursement obligations of the Vendor set forth in Section 2.7.4 exceed \$675,000; to the extent that Oracle or such customer determines pursuant to Section 2.7.4 that named-user licences are required from Oracle by or for Enlogix CIS L.P. or the customers identified below, in no event shall the reimbursement obligations of the Vendor set forth in Section 2.7.4 in respect of such named user licences for each such customer exceed the number of licences identified per customer below; and in no event shall the reimbursement obligations of the Vendor set forth in Section 2.7.4 exceed the amounts identified below in respect of the customers identified below:

Customer	_ D	ollar Amount	Number of Licences	
Enmax	\$	382,000	200	
Centra Gas B.C.	\$	117,500	50	
Union Energy	\$	175,500	100	

2.8 Allocation of Purchase Price Adjustments

Any adjustments to the Purchase Price that are made pursuant to Section 2.4 or 2.5 of this Agreement shall be allocated between the Share Purchase Price and the Partnership Interest Purchase Price on the following basis: (a) one-percent (1%) to the Share Purchase Price; and (b) ninety-nine percent (99%) to the Partnership Interest Purchase Price.

The Vendor represents and warrants to each of the Purchasers as follows and acknowledges that the Purchasers are relying upon the following representations and warranties in connection with their purchase of the Purchased Interests:

3.1.1 Incorporation and Status of the Vendor

The Vendor is duly incorporated and validly existing under the laws of its jurisdiction of incorporation. No act or proceeding has been taken by or against the Vendor in connection with the dissolution, liquidation, winding-up, bankruptcy or reorganization of the Vendor.

3.1.2 Corporate Power of the Vendor and Due Authorization

The Vendor has the corporate power and capacity to own and dispose of the Purchased Interests and to enter into, and to perform its obligations under, this Agreement. Each of this Agreement, the Ancillary Agreements and any other agreements, contracts and instruments required by this Agreement to be delivered by the Vendor at Closing has been duly authorized by the Vendor. This Agreement has been duly executed and delivered by the Vendor and is a valid and binding obligation of the Vendor, enforceable in accordance with its terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies. At the Time of Closing, the Ancillary Agreements to which the Vendor is a party will be duly executed and delivered by the Vendor and will be valid and binding obligations of the Vendor, enforceable in accordance with their respective terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies. The Vendor is not an insolvent person within the meaning of the *Bankruptcy and Insolvency Act* (Canada) and will not become an insolvent person as a result of the Closing.

3.1.3 Incorporation and Status of the Corporation and the Subsidiary

The information contained in <u>Schedule 3.1.3</u> sets out the name and jurisdiction of incorporation, and the directors and officers, of each of the Corporation and the Subsidiary. Each of the Corporation and the Subsidiary is duly formed and organized, and is validly existing and up-to-date in the filing of all corporate and similar returns under the laws of its jurisdiction of incorporation or amalgamation, as applicable. Each of the Corporation and the Subsidiary is a subsisting company in good standing in each jurisdiction in which it carries on the Business with all powers, licenses, registrations, permits and rights that it requires to carry on the Business in such jurisdictions.

3.1.4 Formation and Status of Enlogix CIS L.P.

Enlogix CIS L.P. is a limited partnership duly formed and validly existing under the laws of the Province of Alberta. Enlogix CIS L.P. is a subsisting limited partnership in good standing in each jurisdiction in which it carries on the Business with all powers, licenses, registrations, permits and rights that it requires to carry on the Business in such jurisdiction.

17

3.1.5 **Power and Capacity**

Each of the Corporation and the Subsidiary (and in respect of Enlogix CIS L.P., the general partner(s) thereof as set out in <u>Schedule 3.1.7</u>) has full corporate power and capacity to own, lease and operate its assets, properties and business and to carry on the Business as the same has been owned, leased, operated and conducted during the last twelve (12) months.

The limited partner of Enlogix CIS L.P. (i) has full corporate power and capacity to own the Purchased Partnership Interest, and (ii) has not taken part in the control of the Business or Enlogix CIS L.P.

3.1.6 Capital of the Corporation and the Subsidiary

Schedule 3.1.6 sets out particulars of all of the authorized, issued and outstanding shares of the Corporation and the Subsidiary, the names of the Persons who are the beneficial owners of such shares and, if such beneficial owners are not the registered owners of such shares, the names of the Persons shown on the securities register of the Corporation as the holder of any such shares, and the number and class of shares held by such Persons. All the shares indicated on such Schedule as being issued and outstanding have been validly issued and are outstanding as fully paid and non-assessable shares.

3.1.7 Partnership Interests of Enlogix CIS L.P.

Schedule 3.1.7 sets out the particulars of all of the issued and outstanding partnership interests in Enlogix CIS L.P. and the names of the general partner and the limited partners of Enlogix CIS L.P.

3.1.8 Subsidiaries

The Subsidiary is the only subsidiary of the Corporation. The Subsidiary and Enlogix CIS L.P. have no subsidiaries. Except for the shares of the Subsidiary and the partnership interest in Enlogix CIS L.P. held by the Corporation, none of the Purchased Entities holds or owns, directly or indirectly: (i) any shares or other securities of any other body corporate; or (ii) any partnership interest or other interest of any kind in any corporation, partnership, joint venture, co-ownership or other legal entity; nor is any of the Purchased Entities a party to any agreement, obligation, contract or commitment, whether written, oral or implied to acquire any such shares, securities or interest.

3.1.9 **Reorganization**

Part A of <u>Schedule 3.1.9</u> sets out a complete description of the reorganization (the "**Reorganization**") involving the Purchased Entities that was completed in March 2002 and that resulted in the corporate structure depicted in the diagram set out in Part B of <u>Schedule 3.1.9</u>. The Reorganization was completed in compliance with Applicable Laws and the terms and conditions of all Contracts. All consents, approvals,

authorizations, filings, registrations and notices required under Applicable Law or pursuant to any contract or other obligation to which any of the Affiliates of the Vendor involved in the Reorganization is a party or by which any of such Persons is bound were obtained. The Reorganization has not created and did not create any Liabilities (including Liabilities for Governmental Charges and Taxes) which the Purchased Entities would not otherwise have, or be reasonably expected to have, if the Reorganization had not been completed other than those Liabilities (a) that have been fully paid, satisfied and discharged prior to the date hereof or (b) for which an accrual has been reflected in the Financial Statements.

18

3.1.10 **No Obligations to Issue Securities**

Except pursuant to the Incentive Plans, there are no agreements, subscriptions, options, warrants, rights of conversion, calls, commitments, plans or other rights of any kind outstanding pursuant to which (a) any of the Purchased Entities is, or may become, obligated to issue any shares, partnership units or partnership interests or any securities convertible or exchangeable, directly or indirectly, into any shares, partnership units or partnership interests of the Purchased Entities, or (b) any Person has the right to purchase or otherwise acquire any shares or other securities or any interest of or in any of the Purchased Entities.

3.1.11 Title to, and Right to Sell, Purchased Shares

The Vendor is the sole registered and beneficial owner of the Purchased Shares with good and legal title thereto, free and clear of all Charges. There are no agreements to which the Vendor is a party, which in any way limit or restrict the transfer to ADS of any of the Purchased Shares, other than share transfer restrictions contained in the Corporation's articles (which will be complied with on or prior to Closing), and there are no shareholders agreements, pooling agreements, voting trusts or other agreements or understandings with respect to the voting of the Purchased Shares or any of them. No Person other than ADS has any agreement, option, right or privilege capable of becoming an agreement for the purchase from the Vendor of any of the Purchased Shares or the purchase from the Corporation of any of the issued and outstanding shares of the Subsidiary.

3.1.12 Title to, and Right to Sell, Purchased Partnership Interest

The Vendor is the sole legal and beneficial owner of the Purchased Partnership Interest with good and legal title thereto, free and clear of all Charges. Except for the Limited Partnership Agreement, there are no agreements to which the Vendor is a party, which in any way limit or restrict the transfer to LMGC of any of the Purchased Partnership Interest and there are no agreements or understandings with respect to the voting of the Purchased Partnership Interest. No Person other than LMGC has any agreement, option, right or privilege capable of becoming an agreement for the purchase from the Vendor of any of the Purchased Partnership Interest.

3.1.13 **Bankruptcy**

None of the Purchased Entities is an insolvent person within the meaning of the *Bankruptcy and Insolvency Act* (Canada) nor has made an assignment in favour of its creditors nor a proposal in bankruptcy to its creditors or any class thereof nor had any petition for a receiving order presented in respect of it. None of the Purchased Entities has initiated proceedings with respect to a compromise or arrangement with its creditors or for its winding-up, liquidation or dissolution. No receiver has been appointed in respect of any of the Purchased Entities or any of the Assets and no execution or distress has been levied upon any of the Assets.

3.1.14 **No Contravention**

Neither the execution, delivery and performance of this Agreement and the Ancillary Agreements nor the sale of the Purchased Interests by the Vendor or the Purchased Entities, as applicable, constitute or will result in any of the following:

(a) a default, breach or violation, or an event that, with notice or lapse of time or both, would be a default, breach or violation, of any of the terms, conditions or provisions of the articles, by-laws, constating documents or other organizational or partnership documents of the Vendor or the Purchased Entities, as applicable;

19

- (b) an event which, pursuant to the terms of any Contract or Permit, causes any right or interest of any of the Purchased Entities to come to an end or be amended in any way that is detrimental to the Business or entitles any other Person to terminate or amend any such right or interest (other than third party consents under Contracts listed in Schedule 3.1.15);
- (c) the creation or imposition of any Charge on any Asset; or
- (d) the violation of any Applicable Law by any of the Purchased Entities.

3.1.15 Approvals and Consents of Vendor

Except as set out in <u>Schedule 3.1.15</u>: (i) no authorization, consent or approval of, or filing with or notice to, any governmental authority, regulatory body or court, and (ii) no authorization, consent or approval of, or filing with or notice to, any Person under Applicable Law or pursuant to any Material Contract, any Contract for Licensed Software listed in <u>Schedule 3.1.41</u>, or any other material obligation to which any of the Purchased Entities is a party or by which any of the Purchased Entities is bound, or which is applicable to any of the Assets, is required in connection with the execution, delivery or performance of this Agreement and the Ancillary Agreements by the Vendor or the sale of any of the Purchased Interests hereunder.

3.1.16 Financial Statements

The Financial Statements have been prepared in accordance with generally accepted accounting principles (subject to usual year-end adjustments) throughout the periods indicated and fairly, completely and accurately present the financial position of the Purchased Entities and the results of its operations as of the dates and throughout the periods indicated.

3.1.17 Liabilities

None of the Purchased Entities has any outstanding liabilities of a financial nature, whether accrued, absolute, contingent or otherwise, other than:

- (a) those set out in Schedule 3.1.17;
- (b) those reflected or reserved against in the Financial Statements;
- (c) liabilities in respect of trade or business obligations incurred after the Statements Date in the Ordinary Course of the Business, none of which has or will have, in the aggregate, a Material Adverse Effect.

3.1.18 Guarantees

None of the Purchased Entities is a party to or bound by any agreement of guarantee, support, indemnification, assumption, or endorsement of, or any other similar commitment with respect to the obligations, liabilities (contingent or otherwise) or indebtedness of any Person other than any of the Purchased Entities.

3.1.19 **Indebtedness**

Except as set out in the Financial Statements, none of the Purchased Entities has incurred or is subject to any outstanding loans, bonds, debentures, notes, mortgages or other indebtedness and none of the Purchased Entities has agreed to create or issue any such loans, bonds, debentures, notes, mortgages or other indebtedness (other than trade payables incurred in the Ordinary Course of the Business).

20

3.1.20 Absence of Unusual Transactions and Events

Except as set out in Schedule 3.1.20, since the Statements Date, none of the Purchased Entities has:

- (a) amended its articles, by-laws, constating documents on other organizational or partnership documents (including the partnership agreement), as applicable;
- (b) amalgamated, merged on consolidated with, or acquired all or substantially all of the shares or assets of any Person;
- (c) declared or made any payment of any dividend or other distribution to any shareholder or partner and has not redeemed, purchased or otherwise acquired any shares or other interest (including equity or debt);
- (d) issued or sold any shares or other securities (including any partnership interest) or issued, sold or granted any option, warrant or right to purchase any shares or other securities (including any partnership interest);
- (e) distributed to the holders of any Purchased Shares or Purchased Partnership Interest (or any other interest as the case may be), any property or assets of any of the Purchased Entities;
- (f) transferred, leased, licensed, sold, assigned or otherwise disposed of any of the Assets reflected on the balance sheet forming part of the Financial Statements other than in the Ordinary Course of the Business or that individually or in the aggregate have a net book value equal to or greater than \$50,000;
- (g) changed any accounting or costing systems or methods;
- (h) suffered any extraordinary loss or cancelled or waived any debt, claim or other right;
- (i) paid or satisfied any liabilities of a financial nature other than current liabilities disclosed in the Financial Statements and current liabilities incurred since the Statements Date in the Ordinary Course of the Business;
- (j) granted or paid or agreed to pay or set aside any bonus, profit sharing, retirement, insurance, severance or other benefit, increased the compensation paid (other than for normal merit and cost of living increases), made or granted any improvements or other amendments to the terms of the Employee Plans (except as required by Applicable Law) or made loans or advances to a director, officer or employee of any of the Purchased Entities;
- (k) amended any contract of employment with a director, officer or employee of any of the Purchased Entities;
- (1) mortgaged, pledged, granted a security interest in or otherwise encumbered any of the Assets;
- (m) entered into or terminated any Material Contracts or modified or varied the terms of any Material Contract or waived any right or

- remedy available to any of the Purchased Entities thereunder;
- (n) entered into any transaction that was not in the Ordinary Course of the Business;
- (o) made any capital expenditure that exceeds the amount allocated for capital expenditures in the 2002 budget of the Purchased Entities that was previously provided by the Vendor to the Purchasers;

- (p) terminated, cancelled or modified in any material respect or received notice or a request for termination, cancellation or modification in any material respect of any Material Contract;
- (q) made or suffered any change or changes which have had, or could reasonably be expected to have, a Material Adverse Effect on any of the Purchased Entities;
- (r) suffered or incurred any damage, destruction or loss, whether or not covered by insurance, which has had, or could reasonably be expected to have, a Material Adverse Effect on any of the Purchased Entities; or
- (s) authorized or agreed or otherwise become committed to do any of the foregoing.

3.1.21 **Certain Payments**

Except as set out in <u>Schedule 3.1.21</u>, none of the Purchased Entities or, to the Vendor's knowledge, any director, officer, agent, employee or independent contractor of any of the Purchased Entities has directly or indirectly (a) made any gift (other than gifts that are immaterial in value), bribe, payoff, influence payment, kick-back, or other payment to any Person, private or public, regardless of form, whether in money, property or services (i) to obtain favourable treatment in securing business, (ii) to pay for favourable treatment for business secured, (iii) to obtain special concessions or for special concessions already obtained, for or in respect of any of the Purchased Entities or any of their respective Affiliates, or (iv) in violation of any Applicable Law, or (b) established or maintained any fund or asset that has not been recorded in the Books and Records of the Purchased Entities.

3.1.22 **Non-Arm's Length Transactions**

None of the Purchased Entities has made any payment or loan to, or borrowed any monies from or is otherwise indebted to or in respect of, any officer, director, employee, shareholder, Affiliate or any other Person with whom the Purchased Entity is not dealing at arm's length (within the meaning of the *Income Tax Act* (Canada)) or any Affiliate of any of the foregoing, except for the Receivables and trade payables between the Vendor and its Affiliates and the Purchased Entities incurred in the Ordinary Course of the Business in connection with services performed that are described in Part A of <u>Schedule 3.1.22</u>, and compensation paid to officers, directors and employees in the Ordinary Course of the Business, or as disclosed in the Financial Statements or Part B of Schedule 3.1.22.

Except for up to five percent (5%) of the outstanding publicly-traded shares or other securities of any class of any issuer listed on a recognized stock exchange or as disclosed in Schedule 3.1.22, neither the Vendor nor any Affiliate of the Vendor and, to the Vendor's knowledge, no director, officer or employee of any of the Purchased Entities, the Vendor or any Affiliate of the Vendor: (i) has any pecuniary interest in any supplier or customer of, or service provider to, any of the Purchased Entities or in any other business with which any of the Purchased Entities is in competition or; (ii) has any interest in any property or assets used by any of the Purchased Entities or pertaining to the Business.

Except as disclosed in <u>Schedule 3.1.22</u>, and except for contracts of employment, none of the Purchased Entities is a party to or bound by any contract or agreement with any officer, director, employee, shareholder, partner or any other Person with whom the Purchased Entity is not dealing at arm's length (within the meaning of the *Income Tax Act* (Canada)) or any Affiliate of any of the foregoing.

22

3.1.23 **Intercompany Services**

Except for the Intercompany Services, there are no services and/or functions provided to any of the Purchased Entities by the Vendor or by any Affiliate of the Vendor (other than any of the Purchased Entities). None of the Purchased Entities provides services or functions to the Vendor or any Affiliate of the Vendor (other than pursuant to the Material Contracts listed on <u>Schedule 3.1.24</u>).

3.1.24 **Material Contracts**

Part A of <u>Schedule 3.1.24</u> lists all of the written Material Contracts, and summarizes the terms of all oral Material Contracts, to which each of the Purchased Entities is a party (except for contracts of employment disclosed in <u>Schedule 3.1.28</u>). Except as disclosed in Part B of <u>Schedule 3.1.24</u>, the Vendor has delivered to the Purchasers complete and correct copies of the Material Contracts (and in the case of any oral contracts, complete and correct descriptions of all material terms and conditions), each as amended or modified to the date hereof.

3.1.25 **Customers and Suppliers**

(a) Part A of Schedule 3.1.25 lists all of the customers of each of the Purchased Entities for the twelve (12) month period ending

immediately before the date of this Agreement. Part B of Schedule 3.1.25 lists (i) all of the suppliers of software and/or telecommunications services of each of the Purchased Entities for the twelve (12) month period ending immediately before the date of this Agreement, (ii) all suppliers that provided goods or services to any of the Purchased Entities during the twelve (12) month period ending immediately before the date of this Agreement and invoiced such Purchased Entities, in the aggregate, \$100,000 or more during such period.

(b) Subject to Part C of Schedule 3.1.25, the Vendor has not received notice of, nor, to the Vendor's knowledge, is there, any intention on the part of any such customer or supplier to cease doing business with any of the Purchased Entities or to modify or change in any material manner any existing arrangement with any of the Purchased Entities for the purchase or supply of any products or services. The relationships of each of the Purchased Entities with its customers and suppliers are satisfactory and there are no unresolved disputes of a material nature with any such customer or supplier. All outstanding proposals or bids by any of the Purchased Entities for new business have been made in a manner consistent with the past practices of the Purchased Entities in respect of proposals and bids made to Persons with whom the Purchased Entities deal at arm's length (within the meaning of the *Income Tax Act* (Canada)). All regulatory and other consents, approvals, notices and filings that are or were required pursuant to any and all Contracts between any of the Purchased Entities and its customers and suppliers have been obtained, given or filed, as applicable.

3.1.26 Commitments Under the IBM Lease Agreement

The commitment to purchase equipment from IBM Canada Limited as described in the Volume Offering Transaction Document forming part of the IBM Lease Agreement has been fully satisfied by Enlogix CIS L.P. and there are no additional agreements or commitments to purchase any additional equipment from IBM Canada Limited.

23

3.1.27 **No Default Under Agreements**

Except as disclosed in <u>Schedule 3.1.27</u>, none of the Purchased Entities have received notice of any default, and none of the Purchased Entities are in default under any Contract and there has not occurred any event which, with the lapse of time or giving of notice or both, would constitute a default under any Contract by any of the Purchased Entities or, to the Vendor's knowledge, any other party to the Contract. Each Contract is in full force and effect and the Purchased Entity that is a party thereto is entitled to the full benefit of each Contract in accordance with its terms.

3.1.28 Employment Matters

- (a) Schedule 3.1.28 lists all of the employees of the Purchased Entities as of the date of this Agreement and the date of birth, position, status (including whether on short or long term disability), length of service, date of hire, employment location, and annual salary and other compensation (other than as covered by Employee Plans) of each of them respectively. Except as set out in Schedule 3.1.28, none of the Purchased Entities is a party to nor bound by any contracts in respect of any director, officer or employee, or former director, officer or employee, including:
 - (i) any contract for the employment or statutorily required re-employment of any director, officer or employee, other than contracts of indefinite hire terminable without cause on reasonable notice in accordance with Applicable Law; or
 - (ii) any contract for the employment or statutorily required re-employment of any director, officer or employee providing for severance or termination (including on a change of control of the applicable Purchased Entity) or any other payment on a change of control of the applicable Purchased Entities or upon the completion of the transactions contemplated hereunder.
- (b) None of the Purchased Entities have paid any bonus, fee, distribution, remuneration or other compensation to any of its employees (other than salaries, wages, bonuses or incentive payments paid or payable to employees of any of the Purchased Entities in the Ordinary Course of the Business or in accordance with then current compensation levels and practices, the current ones of which are set out in Schedules 3.1.28 and 3.1.39).
- (c) All salaries, wages, vacation pay, bonuses, commissions and other remuneration from employment for or in respect of the present and former employees of the Purchased Entities have been paid or accrued by the Purchased Entities in the Financial Statements in accordance with generally accepted accounting principles.

24

3.1.29 **Independent Contractors**

- (a) Schedule 3.1.29 lists all of the independent contractors currently engaged by the Purchased Entities and the position, type of contract and compensation of each of them respectively, as well as the effective date of the Contract between each such independent contractor and the applicable Purchased Entity. None of the Contracts with the independent contractors listed in Schedule 3.1.29 provide for any payments on termination, on a change of control of the applicable Purchased Entity, or upon the completion of the transactions contemplated hereunder. None of the Contracts between the independent contractors listed in Schedule 3.1.29 and the Purchased Entities are subject to the implication that they are only terminable on reasonable notice.
- (b) The Vendor has provided to the Purchasers true, correct and complete copies of all Contracts with each of the independent contractors listed in <u>Schedule 3.1.29</u>, and all amendments and restatements thereof as of the date of this Agreement. The rights, liabilities and obligations of each of the Purchased Entities with respect to each of the independent contractors listed in <u>Schedule 3.1.29</u> (including the rights to amend or terminate such arrangements) are fully disclosed in the Contracts provided in accordance with this Section 3.1.29(b).

(c) All fees, consideration and other payments in respect of the independent contractors listed in <u>Schedule 3.1.29</u> have been paid or accrued by the Purchased Entities in the Financial Statements in accordance with generally accepted accounting principles.

3.1.30 Collective Agreements

None of the Purchased Entities is a party to or bound by any Contract with any labour union, trade union, council of trade unions, employee association, employer's association, employee bargaining agent or affiliated bargaining agent. No trade union, employee association or other entity has acquired any bargaining rights by either certification or voluntary recognition with respect to any of the employees of any of the Purchased Entities and, to the Vendor's knowledge, there are no applications or discussions involving any of the Purchased Entities regarding such certification or recognition. To the Vendor's knowledge, there are no union-organizing activities involving the employees of any of the Purchased Entities.

3.1.31 **Real Property**

None of the Purchased Entities owns any real property or has any interests therein, other than under the Premises Leases.

3.1.32 **Leased Premises**

Schedule 3.1.32 lists all the Premises Leases. Each Premises Lease is in full force and effect, unamended by oral or written agreement, and the Purchased Entities are entitled to the full benefit of such Premises Lease in accordance with the terms thereof. None of the Purchased Entities or the Vendor has received notice of any default, and none of the Purchased Entities is in default, under any Premises Lease and there has not occurred any event which, with a lapse of time or giving of notice, or both, would constitute a default under any Premises Lease. There are no disputes between any of the Purchased Entities and any landlord under any of the Premises Leases. The Premises Leases have been delivered to the Purchaser on or before the Closing Date and the documents so delivered are complete in all respects and include all amendments, extensions or renewals pertaining thereto. None of the Premises Leases have been assigned by any of the Purchased Entities in favour of any Person. The current uses of the Leased Premises comply with Applicable Law.

25

3.1.33 **Environmental Compliance**

Each of the Purchased Entities is in material compliance with all federal, provincial, state, municipal, local and foreign environmental laws which are applicable to the operation of the Business (collectively, the "Environmental Laws") including any such environmental laws relating to pollution or protection of human health or the environment, or a discharge, emission, spill or other release of any contaminant (as defined in the *Environmental Protection Act* (Ontario) and any other applicable local or foreign legislation existing as of the date hereof that is applicable to any of the Purchased Entities, the Assets or the Business). None of the Purchased Entities have received any written communication, whether from a governmental authority, private party, employee or otherwise, that alleges that any of the Purchased Entities is not in compliance with any Environmental Laws. The Business and the Assets are not the subject of any governmental or regulatory remedial or control action or order or, to the Vendor's knowledge, any investigation or evaluation concerning non-compliance or potential non-compliance with Environmental Laws. Since the date of lease by the applicable Purchased Entity, the Leased Premises have not been and are not now used as a landfill or waste disposal site, nor, to the Vendor's knowledge, has any hazardous substance or contaminant ever been deposited in or disposed of on the Leased Premises, nor, to the Vendor's knowledge, has there ever been any release, emission, spill or discharge of any contaminant which would give rise to any action or claim relating to violation of any such Environmental Laws or other requirements.

3.1.34 Title and Sufficiency of Assets

Each of the Purchased Entities has good and marketable title to all of the Assets (other than Assets which by their nature are not capable of being owned or "off-the-shelf" computer software), free and clear of all Charges, other than Charges listed on Schedule 3.1.34. The Assets (i) are sufficient to permit the continued operation of the Business in substantially the same manner as conducted in the year ended on the date of this Agreement, and (ii) together with any assets that are identified in the 2002 Outlook and Five Year Forecast of the Purchased Entities previously delivered to the Purchasers by the Vendor as assets that must be acquired, are sufficient to permit the performance of any services relating to or in connection with all outstanding proposals or bids for new business by any of the Purchased Entities. Except as set out in Schedule 3.1.34, there are no assets owned by any of the Purchased Entities which have a net book value in excess of \$25,000. Schedule 3.1.34 sets out a complete and accurate list of all locations where the tangible Assets are situate to the extent that such tangible Assets are not located in the Leased Premises, including a brief description of the tangible Assets situate at each location. There is no agreement, option or other right or privilege outstanding in favour of any Person for the purchase from any of the Purchased Entities of the Business or any of the Assets out of the Ordinary Course.

3.1.35 **Assets in Good Condition**

To the Vendor's knowledge, all of the Personal Property of the Purchased Entities is in good operating condition and in a state of good maintenance and repair, in all material respects, having regard to the use to which the assets are put and the age thereof.

26

3.1.36 **Personal Property Leases**

<u>Schedule 3.1.36</u> lists all Personal Property Leases which involve a payment by any Purchased Entity in the future of more than \$25,000, provided that the aggregate amounts payable under those Personal Property Leases not listed on <u>Schedule 3.1.36</u> does not exceed \$175,000.

Each Personal Property Lease is in full force and effect and the Purchased Entities are entitled to the full benefit of each Personal Property Lease in accordance with its terms. None of the Purchased Entities have received notice of any default, and none of the Purchased Entities are in default, under the Personal Property Leases and there has not occurred any event which, with a lapse of time or giving of notice, or both, would constitute a default under any Personal Property Lease. The Personal Property Leases have been delivered to the Purchaser on or before the Closing Date and the documents so delivered are complete in all respects and include all amendments, extensions or renewals pertaining thereto.

3.1.37 Receivables

The Receivables are valid obligations which arose in the Ordinary Course of the Business and, except for reserves for doubtful accounts which have been established in accordance with generally accepted accounting principles consistent with prior periods and reflected in the Financial Statements, are collectible in the Ordinary Course of the Business, in the aggregate, at their full face value. Except for Receivables that are owing from the Vendor or any of its Affiliates that were incurred in the Ordinary Course of the Business in connection with services performed, none of the Receivables is due from an Affiliate of any of the Purchased Entities.

3.1.38 Tax Matters, etc.

(a) Each of the Purchased Entities has prepared and filed on time all Tax Returns and other documents of every nature required to be filed by or on behalf of it prior to the date hereof in all applicable jurisdictions in respect of any Governmental Charges and Taxes or in respect of any other provision in any domestic or foreign, federal, provincial, municipal, state, territorial or other taxing statute for all fiscal periods ending prior to the date hereof, and will continue to do so in respect of such Tax Returns or other documents due on or before the Closing Date. All such Tax Returns and other documents are correct and complete in all material respects, and no material fact has been omitted therefrom. No extension of time in which to file any such Tax Returns or other documents is in effect. Each of the Purchased Entities has paid in full all Governmental Charges and Taxes shown on such Tax Returns, or any assessments or reassessments in respect of such Tax Returns and all other Governmental Charges and Taxes required to be paid, on or prior to the date hereof and has made adequate provision in the Financial Statements in accordance with generally accepted accounting principles for the payment of all Governmental Charges and Taxes, and all professional fees related thereto, payable in respect of the Business or Assets or otherwise for all fiscal periods ending on or before the Statements Date.

27

- Canadian federal and provincial income tax assessments have been issued to the Corporation covering all past periods up to and including the fiscal year ended 2000 and such assessments, if any amounts were owing in respect thereof, have been paid in full, and as the Corporation was incorporated in 1998, the 1998 fiscal year and all subsequent fiscal years remain open for reassessment of additional taxes, interest or penalties. Assessments for all other applicable Governmental Charges and Taxes have been issued and any amounts owing thereunder have been paid in full, and as the Corporation was incorporated in 1998, the 1998 fiscal year and all subsequent fiscal years remains open for reassessment of additional Governmental Charges and Taxes. There are no proceedings (as defined in Section 3.1.46) in progress, pending or threatened against any of the Purchased Entities in respect of any Governmental Charges and Taxes and, in particular, there are no currently outstanding reassessments, written enquiries or issues which have been issued to, or raised in respect of and communicated to, any of the Purchased Entities by any governmental authority relating to any Governmental Charges and Taxes. No governmental body has challenged, disputed or questioned any of the Purchased Entities in respect of Governmental Charges and Taxes or of any returns, filings or other reports filed under any statute providing for Governmental Charges and Taxes. None of the Purchased Entities are negotiating any draft assessment or reassessment with any governmental body. None of the Purchased Entities or the Vendor has received any indication from any governmental body that an assessment or reassessment of any of the Purchased Entities is proposed in respect of any Governmental Charges and Taxes, regardless of its merits. None of the Purchased Entities have executed or filed with any governmental body any agreement or waiver extending the period for assessment, reassessment, or collection of any Governmental Charges and Taxes.
- (c) Each of the Purchased Entities has withheld or collected from each payment made to any of its present or former employees, officers and directors, all persons who are non-residents of Canada for purposes of the *Income Tax Act* (Canada) and any other applicable payees, all amounts required by Applicable Law to be withheld or collected, and has remitted such withheld or collected amounts within the prescribed periods to the appropriate governmental authority. Each of the Purchased Entities has charged, collected and remitted on a timely basis all Governmental Charges and Taxes as required under Applicable Law on any sale, supply or delivery whatsoever, made by any of the Purchased Entities.
- (d) None of the Purchased Entities will at any time be deemed to have a capital gain pursuant to subsection 80.03(2) of the *Income Tax Act* (Canada) or any analogous provincial provision as a result of any transaction or event taking place in any taxation year ending on or before the Closing Date.
- (e) There are no circumstances existing which could result in the application of Sections 78 or 160 of the *Income Tax Act* (Canada) or any equivalent provincial provision to any of the Purchased Entities.

28

(f) The maximum undepreciated capital cost to Enlogix CIS L.P. of its depreciable property of all classes expected to be delivered by the Vendor for Canadian federal and provincial income tax purposes will be Cdn\$28,000,000 but in no case will it be less than the lesser of (i) Cdn\$26,000,000, or (ii) the aggregate Purchase Price as finally adjusted in accordance with the terms of this Agreement and translated into Canadian dollars based on the Bank of Canada noon spot rate in effect on the date each payment is made.

- (g) None of the Purchased Entities are or have even been a United States Real Property Holding Corporation as defined in Section 897(c) (2) of the Code.
- (h) The Subsidiary is not delinquent in the payment of any Governmental Charges and Taxes and there are no Governmental Charges or Tax liens upon any property or assets of the Subsidiary except liens for current Governmental Charges and Taxes not yet due; the Subsidiary is not, or has it ever been, a party to any agreement to share federal, state or local income tax liability with any corporation with which it has or will file a consolidated tax return for periods ending on or prior to the Closing Date.
- (i) All Governmental Charges and Taxes which are required by the laws of the United States, any state, municipality, subdivision or instrumentality of the United States or any foreign country or any other taxing authority, to be withheld or collected by the Subsidiary have been duly withheld or collected and, to the extent required, have been paid to the proper governmental authorities or properly deposited as required by Applicable Laws. No deficiency for any tax or claim for additional Governmental Charges and Taxes relating to or affecting in any manner the Subsidiary has been proposed, asserted, or assessed against the Vendor or the Subsidiary.

3.1.39 **Employee Plans**

All Employee Plans are described in <u>Schedule 3.1.39</u>. Except as set out in <u>Schedule 3.1.39</u>:

- (a) Each Employee Plan which is required to be registered pursuant to its terms or by Applicable Law is and has been registered, and each Employee Plan has been qualified, established and administered and, if applicable, invested and funded in accordance in all material respects with Applicable Law and the terms of the applicable Employee Plan. Each ERISA Plan that is intended to be a qualified Pension Plan under Section 401 of the Code has received, or is eligible to receive within the remedial amendment period described in Section 401(b) of the Code, a favourable determination letter from the Internal Revenue Service, and its related trust has been determined to be exempt from tax under Section 501(a) of the Code and, to the Vendor's knowledge, nothing has occurred since the date of such letter to materially adversely affect such qualification or exemption. To the Vendor's knowledge, no fact or circumstance exists that could reasonably be expected to adversely affect the registered or other tax-exempt status of an Employee Plan.
- (b) There are no outstanding defaults or violations by any party to any Employee Plan and no Governmental Charges and Taxes, penalties or fees are owing or exigible under any of the Employee Plans (except such fees as provided for or contemplated in the Employee Plans). All reports and the documents required to be filed by any of the Employee Plans with any governmental body or distributed to Employee Plan participants or beneficiaries under Applicable Law have been timely filed or distributed.
- (c) None of the Purchased Entities have incurred any liability under Title IV of ERISA since the effective date of ERISA that has not been satisfied in full, and no material liability currently exists under any ERISA Plan other than liability for benefits payable in the Ordinary Course and premiums due the Pension Benefit Guaranty Corporation, which premiums have been paid when due. To the Vendor's knowledge, the Pension Benefit Guaranty Corporation has not instituted proceedings to terminate any of the ERISA Plans.

- (d) The consummation of the transactions contemplated herein will not result in any improvement or acceleration of benefits under any Employee Plan or any increase in funding or contribution obligations of a Purchased Entity under any Employee Plan, except with respect to any vesting in respect of the Employee Plans described in Section 2.6 of the Human Resources Agreement and any statutory vesting or "grow-in" entitlements or statutorily accelerated funding obligations arising in respect of any partial wind-up of one or both of the Vendor's Pension Plans as defined in the Human Resources Agreement.
- (e) No Employee Plan, nor any related trust or other funding medium thereunder, is subject to any pending investigation, examination or other proceeding, action or claim initiated by any governmental agency or instrumentality, or by any other party (other than routine claims for benefits), and, to the Vendor's knowledge, there exists no state of facts which after notice or lapse of time or both could reasonably be expected to give rise to any such investigation, examination or other proceeding, action or claim or to affect the registration of any Employee Plan required to be registered.
- (f) None of the Purchased Entities have engaged in, nor have any knowledge of anyone engaging in any transaction or act or failed to act in any manner that would subject the Vendor, or any of the Purchased Entities to any material liability for a breach of fiduciary duty under ERISA. None of the Purchased Entities have engaged in any transaction in violation of Section 406(a) or (b) of ERISA or Section 4975(d) of the Code or that would result in a civil penalty being imposed on any Related Entity under subsections (i) or (l) of Section 502 of ERISA.
- All contributions and premiums required to be made by the Purchased Entities or the Vendor under the terms of each Employee Plan or by Applicable Law have been made in a timely fashion in accordance with the terms of the Employee Plans and Applicable Law and are properly recorded on the books of the Purchased Entities or the Vendor, as applicable. All contributions made or required to be made under any Employee Plan for which any of the Purchased Entities has claimed an income tax deduction meet the requirements for deductibility under the applicable tax legislation. None of the Purchased Entities have, and as of the Time of Closing will have, any liability (other than liabilities accruing after the Time of Closing and liabilities for which the Vendor has retained responsibility pursuant to the Human Resources Agreement) with respect to any of the Employee Plans. No ERISA Plan or any trust established thereunder has incurred any "accumulated funding deficiency" (as defined in Section 302 of ERISA and Section 412 of the Code), whether or not waived, as of the last day of the most recent fiscal year of each respective ERISA Plan ended prior to the date of this Agreement. Contributions or premiums in respect of the period prior to August 1, 2002 will either be paid when due by any of the Purchased Entities prior to the Time of Closing or will be accrued as liabilities in the Financial Statements.
- (h) The Pension Plans (other than the Vendor's Supplemental Retirement Plan, as defined in the Human Resources Agreement) have been funded on both an ongoing and solvency basis in accordance with the relevant terms of each plan and Applicable Law and in

- (i) No amendments or improvements have been made or promised respecting any Pension Plan since the date of the last actuarial report relating to such plan or where there is no actuarial report relating to a Pension Plan since the Statements Date, or respecting any Benefit Plan or Incentive Plan since the Statements Date, and no amendments or improvements to an Employee Plan will be made or promised by any of the Purchased Entities before the Time of Closing.
- (j) There have been no withdrawals of, or applications made to withdraw, surplus from any Benefit Plan or Pension Plan, and all transfers of assets respecting any Benefit Plan or Pension Plan, have been undertaken and completed in accordance with the relevant terms of each plan and Applicable Law.
- (k) The Vendor has furnished or made available to the Purchasers true, correct and complete copies of all the Employee Plans as amended as of the date hereof together with all related documentation including funding agreements, actuarial reports, annual reports (if required under Applicable Law, including ERISA), funding and financial information returns and statements, the most recent determination letter received from the Internal Revenue Service with respect to each ERISA Plan that is intended to be a qualified Pension Plan under Section 401 of the Code, and plan summaries, booklets and personnel manuals. No material changes have occurred to the Employee Plans or are expected to occur which would affect the actuarial reports or financial statements required to be provided to the Purchaser pursuant to this Section 3.1.39, other than changes in general economic conditions.
- (l) No advance tax rulings affecting the rights or obligations of the Purchased Entities have been sought or received in respect of the Employee Plans. No ERISA Plan is a "multi-employer pension plan" as such term is defined in Section 3(37) of ERISA, or a plan covered by Title IV of ERISA or subject to the funding requirements of Section 412 of the Code. No Employee Plan is a "multiple employer plan" within the meaning of Section 413(c) of the Code. No event has occurred with respect to any of the Purchased Entities in connection with which any of the Purchased Entities would be subject to any material Liability or Charge with respect to any ERISA Plan.
- (m) All employee data necessary to administer each Employee Plan has been provided, or will be provided within five (5) Business Days following the Closing Date, by the Vendor to the Purchaser and, to the Vendor's knowledge, is true and correct.
- (n) No insurance policy or any other contract or agreement affecting any Employee Plan requires or permits a retroactive increase in premiums or payments due thereunder. To the Vendor's knowledge, the level of insurance reserves under each insured Employee Plan is reasonable and sufficient to provide for all incurred but unreported claims.
- (o) None of the Employee Plans provides benefits to retired employees or to the beneficiaries or dependents of retired employees.
- (p) None of the Purchased Entities have during the past ninety (90) days taken any action which would require any compliance under the WARN Act, including the termination or laying off of any employees, or any other action that could constitute a "plant closing" or "mass layoff", as those terms are defined by the WARN Act.

3.1.40 Insurance

Particulars of the policies of insurance maintained by each of the Purchased Entities as at the date of this Agreement are set out in <u>Schedule 3.1.40</u>.

31

3.1.41 Intellectual Property and Software

- (a) Trade-Marks. Part A-1 of Schedule 3.1.41 contains a complete and accurate listing of all registrations and applications for registration of Trade-Marks which are recorded in the name of a Purchased Entity and in each case, the name of the Purchased Entity under whose name the registration or application is recorded. Part A-2 of Schedule 3.1.41 contains a complete and accurate listing of all Trade-Marks which have been licensed for use by a Purchased Entity pursuant to an agreement in writing by a Person other than a Purchased Entity and in each case, the name of the licensor of the Trade-Mark. To the Vendor's knowledge, no Person has granted an oral licence to a Purchased Entity in respect of the use of any Trade-Mark. All registrations and applications for registration listed in Part A-1 of Schedule 3.1.41 has been rejected or opposed. To the Vendor's knowledge, the Purchased Entity identified in Part A-1 of Schedule 3.1.41 was and is entitled to secure the registration for each corresponding Trade-Mark identified in Part A-1 of Schedule 3.1.41. To the Vendor's knowledge, each of the registrations identified in Part A-1 of Schedule 3.1.41 is not invalid. Each Purchased Entity identified in Part A-1 of Schedule 3.1.41. No Purchased Entity has granted to any Person any rights or interests in or to any of the Trade-Marks. To the Vendor's knowledge, no Person has violated or infringed upon any rights or interests of a Purchased Entity in or to the Trade-Marks listed in Parts A-1 and A-2 of Schedule 3.1.41.
- (b) <u>Patents.</u> No Purchased Entity has any pending patent or industrial design application or any patent or industrial design registration. No Person has granted a licence in writing to a Purchased Entity in respect of any specific patent or industrial design application or any

- specific patent or industrial design registration. To the Vendor's knowledge, no Person has granted an oral licence to a Purchased Entity in respect of any specific patent or industrial design application or any specific patent or industrial design registration.
- (c) <u>Copyrights</u>. No Purchased Entity has any pending application for copyright or any copyright registration. With the exception of any licences granted to a Purchased Entity in respect of Software, no Person has granted to a Purchased Entity a licence in writing in respect of any specific Copyrights. With the exception of any licences granted by a Purchased Entity in respect of Software, no Purchased Entity has granted a licence in writing in respect of any specific Copyrights to any other Person.
- (d) Know-How. With the exception of any licences granted to a Purchased Entity in respect of Software, no Person has granted to a Purchased Entity a licence in writing in respect of any specific Know-How. With the exception of any licences granted by a Purchased Entity in respect of Software, no Purchased Entity has granted a licence in writing in respect of any specific Know-How to any other Person.
- (e) <u>SCT Development Agreement Customizations</u>. Part B of <u>Schedule 3.1.41</u> is a complete and accurate listing of all Customizations (as defined in the SCT Development Agreement) developed by Enlogix CIS L.P. as part of the Services (as defined in the SCT Development Agreement) and all Tasks (as defined in the SCT Development Agreement) in respect of which Services (as defined in the SCT Development Agreement) are yet to be performed by Enlogix CIS L.P. None of the Customizations listed in Part B of <u>Schedule 3.1.41</u> and none of the Customizations resulting from any Tasks listed in Part B of <u>Schedule 3.1.41</u> form part of the CIS Software. The Purchased Entities have not disclosed any such Customizations to any Person other than SCT.

- (f) BANNER Modifications. To the Vendor's knowledge, the Purchased Entities have all necessary rights and interests, including licences, to develop modifications to, enhancements to, corrections to, improvements to, customizations of and derivative works based on the SCT Software. The Purchased Entities either own all rights, title and interests, including intellectual and industrial property rights, in and to all BANNER Modifications, other than the Customizations identified in Part B of Schedule 3.1.41, or have all necessary rights and interests, including licences, to use all such BANNER Modifications in accordance with the Software License Agreement between SCT Utility Systems, Inc. and Westcoast Energy Inc., effective as of April 29, 1996, as assigned to Enlogix CIS L.P. Other than any general descriptions of the BANNER Modifications provided in a Purchased Entity's marketing material, and other than any disclosures of any BANNER Modifications which have been made by one or more Purchased Entities to contractors who have been retained by a Purchased Entity to perform services in respect of Software and, in the case of contractors currently engaged by a Purchased Entity, who have agreed in writing to maintain the confidentiality of any such BANNER Modifications, and other than furnishing object code versions, and in the case of Manitoba Hydro and Centra Gas Manitoba Inc., furnishing a source code version under a software sublicense agreement with Enlogix CIS L.P., made as of January 28, 2002, of any such BANNER Modifications and associated user documentation to customers of the Business, the Purchased Entities have not disclosed to any Person, including SCT, and, subject to a possible obligation to disclose to SCT, are under no obligation or duty to disclose, to any Person, any BANNER Modifications, other than any Customizations (as defined in the SCT Development Agreement) developed by or to be developed by Enlogix CIS L.P. for SCT as part of the Services (as defined in the SCT Development Agreement).
- CIS Software. Part C of Schedule 3.1.41 contains a complete and accurate listing of all Licensed Software which forms part of the CIS Software, and in each case, the name of the applicable licensor and the name of the Purchased Entity having rights in respect thereof. Part D of Schedule 3.1.41 contains a complete and accurate listing of all Proprietary Software which forms part of the CIS Software. Together, the Licensed Software identified in Part C of Schedule 3.1.41, the BANNER Modifications and the Proprietary Software identified in Part D of Schedule 3.1.41 comprise the CIS Software. Subject to: (i) any rights a licensor identified in Part C of Schedule 3.1.41 may have to the corresponding Licensed Software identified in Part C of Schedule 3.1.41; (ii) any rights SCT may have in or to BANNER Modifications; and (iii) any rights Manitoba Hydro and Centra Gas Manitoba Inc. may have under a software sublicense agreement with Enlogix CIS L.P., made as of January 28, 2002, no Person other than a Purchased Entity has the right to grant a license to or the right to permit other Persons to access the CIS Software. Other than disclosures made by one or more Purchased Entities to contractors who have been retained by a Purchased Entity to perform services in respect of any Proprietary Software identified in Part D of Schedule 3.1.41 and, in the case of contractors currently engaged by a Purchased Entity, who have agreed in writing to maintain the confidentiality of any such Proprietary Software, and other than furnishing object code versions, and in the case of Manitoba Hydro and Centra Gas Manitoba Inc., furnishing a source code version under a software sublicense agreement with Enlogix CIS L.P., made as of January 28, 2002, of any such Proprietary Software and associated user documentation to customers of the Business, the Purchased Entities have not disclosed, and are under no obligation or duty to disclose, to any Person any Proprietary Software identified in Part D of Schedule 3.1.41.

33

(h) <u>Licensed Software</u>. Part E of <u>Schedule 3.1.41</u> contains a complete and accurate listing of all Licensed Software other than: (i) the Licensed Software identified in Part C of <u>Schedule 3.1.41</u>; and (ii) off-the-shelf personal computer software subject to transferable licences, and in each case, the name of the applicable licensor. Subject to any rights or interests which a Person other than the licensor of any Licensed Software may have in the use of the Licensed Software, the Purchased Entities have all necessary rights and interests, including licences, to use all Licensed Software, excluding any software licensed from Oracle Corporation Canada Inc., in substantially the same manner as used at any time since December 22, 2000 or since the date when such rights and interests in the specific Licensed Software were first acquired by the Purchased Entities. The Purchased Entities maintain and have maintained sufficient licences in respect of all Licensed Software, excluding any software licensed from Oracle Corporation Canada Inc., and are not in underpayment of any fees owed to any licensor of any Licensed Software, excluding any software licensed from Oracle Corporation Canada Inc.. The Purchased Entities use and have used all Licensed Software in accordance with the provisions of the applicable licence therefor.

- (i) <u>Source Code</u>. Part F of <u>Schedule 3.1.41</u> contains a complete and accurate listing of all Licensed Software, other than Licensed Software in Part C, the source code for which is in the possession of or has otherwise been made available to one or more Purchased Entities and in each case, the name of the Person who furnished or otherwise made such source code available to the Purchased Entities. Except as disclosed in Part G of <u>Schedule 3.1.41</u>, there are no source code escrow agreements to which a Purchased Entity is a party relating to any of the Software.
- (j) Proprietary Software. Part H of Schedule 3.1.41 contains a complete and accurate listing of all Proprietary Software other than the Proprietary Software identified in Part D of Schedule 3.1.41. To the Vendor's knowledge, no Person other than a Purchased Entity has any right or interest, including intellectual and/or industrial property rights, in or to any Proprietary Software in Part H of Schedule 3.1.41 and subject to any rights SCT may have in or to Banner Modifications and any rights Manitoba Hydro and Centra Gas Manitoba Inc. may have under a software sublicense agreement with Enlogix CIS L.P., made as of January 28, 2002, and to the Vendor's knowledge, no Person other than a Purchased Entity has any right or interest, including intellectual and/or industrial property rights, in or to any Proprietary Software in Part D of Schedule 3.1.41. With respect to all Proprietary Software and BANNER Modifications: (i) at least one Purchased Entity maintains machine-readable, master-reproducible copies, source code listings, and technical documentation for all releases or versions thereof currently used by a Purchased Entity to provide a service under a Contract to a customer; (ii) in each case, the machine-readable copy substantially conforms to the corresponding source code listing; and (iii) in each case, it can be maintained and modified by competent programmers familiar with the applicable language, hardware and operating systems.

- (k) To the Vendor's knowledge, the Purchased Entities have all necessary rights and interests, including licenses, to use the Intellectual Property and the Software in substantially the same manner as conducted in the year ended on the date of this Agreement. The use of the Intellectual Property and the Software in the Business does not: (i) breach or result in any non-fulfillment of any covenant or agreement that a Purchased Entity has with any other Person; (ii) breach any express licence or permission granted to a Purchased Entity by any other Person; (iii) to the Vendor's knowledge, breach any implied licence or permission granted to a Purchased Entity by any other Person; (iv) breach or violate any duty or obligation of confidentiality owed by a Purchased Entity to any other Person or any intellectual and/or industrial property of any other Person. The use of the Intellectual Property and the Software in the Business has not: (v) breached or resulted in any non-fulfillment of any covenant or agreement that a Purchased Entity has with any other Person; (vi) breached any express licence granted to a Purchased Entity by any other Person; (viii) breached any implied licence granted to a Purchased Entity by any other Person; (viii) breached or violated any duty or obligation of confidentiality owed by a Purchased Entity to any other Person; (vix) to the Vendor's knowledge, infringed, violated or misappropriated any intellectual and/or industrial property right of any other Person or any intellectual and/or industrial property of any other Person.
- (1) Other than in respect of Licensed Software and the Licensed Intellectual Property, and with the exception of the obligations of Enlogix CIS L.P. under the SCT Development Agreement, none of the Purchased Entities is subject to any obligation or arrangement pursuant to which any other Person has or shall have any rights, title or interests in or to any Intellectual Property or Software which is currently being developed or may in the future be developed by or for any Purchased Entity.
- (m) None of the Purchased Entities has received any written notice challenging a Purchased Entity's ownership of or right to use any Intellectual Property or Software or challenging the validity of any of the registrations or applications identified in Part A-1 of <u>Schedule 3.1.41</u>.
- (n) To the extent that any Person who is currently engaged by a Purchased Entity, other than a regular full-time employee of a Purchased Entity contributed to the conception, reduction to practise, invention, creation or development of any Proprietary Intellectual Property, Proprietary Software or BANNER Modifications, such Person has irrevocably assigned to such Purchased Entity in writing all intellectual and industrial property rights in such Person's contribution to any such Proprietary Intellectual Property, Proprietary Software or BANNER Modifications.
- (o) The Purchased Entities have used commercially reasonable efforts to maintain the confidentiality of any Proprietary Intellectual Property or Proprietary Software the value of which to the Purchased Entities is contingent upon maintenance of the confidentiality thereof.

- (p) Except as disclosed in Part I of <u>Schedule 3.1.41</u>, to the Vendor's knowledge, no Person: (i) has breached or is in breach of, or has failed or is failing to fulfill any covenant or agreement with a Purchased Entity in respect of the Intellectual Property or the Software; (ii) has breached or is in breach of any express or implied licence granted by a Purchased Entity; (iii) has breached or is breaching or has violated or is violating any duty or obligation of confidentiality owed to a Purchased Entity; or (iv) has infringed, violated or misappropriated or is infringing, violating or misappropriating any intellectual and/or industrial property right of a Purchased Entity or any intellectual and/or industrial property of a Purchased Entity.
- (q) The Proprietary Intellectual Property and the Proprietary Software are not subject to any security interest, lien, charge, hypotec, pledge, encumbrance, mortgage or option. There are no security interests, liens, charges, hypotecs, pledges, encumbrances, mortgages or options against any Purchased Entity's rights or interests in the Licensed Software or the Licensed Intellectual Property. All amounts, including royalties, due and payable by each of the Purchased Entities to any licensors of any Licensed Intellectual Property or Licensed Software have been paid in full or accrued in the Financial Statements.

The Data Processing System (i) adequately meets the data processing needs of the Business and operations of each of the Purchased Entities as conducted during the last twelve (12) months, and (ii) together with any assets that would form part of the Data Processing System and are identified in the 2002 Outlook and Five Year Forecast of the Purchased Entities previously delivered to the Purchasers by the Vendor as assets that must be acquired, adequately meets the data processing needs arising out of or in connection with the performance of any services contemplated by all outstanding proposals and bids for new business by any of the Purchased Entities, in each case including the need to run the Application Software to process data for customers of each of the Purchased Entities. Each of the Purchased Entities has arranged for back-up data processing services adequate to meet its data processing needs in the event the Data Processing System or any of its components is rendered temporarily or permanently inoperative as a result of a natural or other disaster.

3.1.43 **Permits and Registrations**

Each of the Purchased Entities holds all applicable permits, licences, approvals, consents, authorizations, filings, registrations, certificates, franchises and other indicia of authority which are Related to the Business or are required in order to own the Assets or carry on the Business consistent with the manner in which it was conducted during the last twelve (12) months (collectively, the "Permits"). All the Permits are listed in Schedule 3.1.43 (including any applicable expiry dates) and are in full force and effect; each of the Purchased Entities is in compliance in all material respects with all the terms and conditions relating to the Permits; and there are no proceedings (as defined in Section 3.1.46) in progress, pending or, to the Vendor's knowledge, threatened which may result in revocation, cancellation, suspension, rescission or any adverse modification of any of the Permits. Except as set out in Schedule 3.1.15, neither the terms and conditions relating to the Permits nor the legislation or regulations pursuant to which the same were issued require that any consent or approval of, or filing with or notice to, any governmental agency or regulatory body or other Person be made to assure the continued holding by the Purchased Entities of the Permits after completion of the transaction contemplated by this Agreement. The Corporation is registered under Part IX of the *Excise Tax Act* (Canada).

36

3.1.44 Bank Accounts/Powers of Attorney

Schedule 3.1.44 lists:

- (a) the name of each bank, trust company and other financial institution in which any of the Purchased Entities holds accounts or safety deposit boxes and the names of every Person authorized to draw thereon or to have access thereto; and
- (b) the name of each Person holding a general or special power of attorney from any of the Purchased Entities and a summary of the terms thereof.

3.1.45 **Compliance with Laws**

Each of the Purchased Entities is in compliance in all material respects with all Applicable Laws of each jurisdiction in which the Business is carried on.

3.1.46 Litigation and Other Proceedings

Except as set out in <u>Schedule 3.1.46</u>, there is no court, administrative, regulatory or similar proceeding (whether civil, quasi-criminal or criminal); arbitration or other dispute settlement procedure; investigation or inquiry by any governmental, administrative, regulatory or similar body; or any similar action, suit, claim, application, complaint, investigation or proceeding (collectively "**proceedings**") pending or, to the Vendor's knowledge, threatened, against or involving any of the Purchased Entities or affecting the Business or the transactions contemplated by this Agreement, and, to the Vendor's knowledge, there is no factual or legal basis which could give rise to any such action, suit, proceeding, claim, application, complaint or investigation.

3.1.47 **Corporate Records**

The corporate or partnership records, as applicable, and minute books of each of the Purchased Entities contain complete and accurate copies of its articles, by-laws, shareholders and/or partnership agreements, every written resolution of directors, partners and shareholders, and minutes of all meetings of directors, partners and committees thereof and shareholders, as applicable, held since its date of formation, and all such meetings were duly called and held. The share certificate books, register of shareholders, record of partners, register of transfers and register of directors and officers, as applicable, of each of the Purchased Entities are complete and accurate.

3.1.48 **Vendor Resident of Canada**

The Vendor is not a non-resident of Canada for purposes of Section 116 of the Income Tax Act (Canada).

3.1.49 Full Disclosure

There is no misstatement of a material fact, or failure to disclose any material matter or fact, which renders any representations and warranties contained in this Article 3 inaccurate, untrue or misleading in any material respect.

3.2 By the Purchasers

3.2.1 ADS represents and warrants to the Vendor as follows and acknowledges that the Vendor is relying upon the following representations and

warranties in connection with its sale of the Purchased Shares:

(a) Incorporation and Status of ADS

ADS is duly incorporated and validly existing under the laws of its jurisdiction of incorporation. No act or proceeding has been taken by or against ADS in connection with the dissolution, liquidation, winding-up, bankruptcy or reorganization of ADS.

37

(b) Corporate Power of ADS and Due Authorization

ADS has the corporate power and capacity to enter into, and to perform its obligations under, this Agreement. Each of this Agreement, the Ancillary Agreements and any other agreements, contracts and instruments required by this Agreement to be delivered by ADS at Closing has been duly authorized by ADS. This Agreement has been duly executed and delivered by ADS and is a valid and binding obligation of ADS, enforceable in accordance with its terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies. At the Time of Closing, the Ancillary Agreements will be duly executed and delivered by ADS and will be valid and binding obligations of ADS, enforceable in accordance with their respective terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies.

(c) Consents and Approvals of ADS

Except for the required notification and approval under the *Competition Act* (Canada) described in Section 4.1.6, no authorization, consent, or approval of, or filing with or notice to, any governmental authority, regulatory body or court is required in connection with the execution, delivery or performance of this Agreement by ADS, or the purchase of the Purchased Shares hereunder.

3.2.2 LMGC represents and warrants to the Vendor as follows and acknowledges that the Vendor is relying upon the following representations and warranties in connection with its sale of the Purchased Partnership Interest:

(a) Incorporation and Status of LMGC

LMGC is duly incorporated and validly existing under the laws of its jurisdiction of incorporation. No act or proceeding has been taken by or against LMGC in connection with the dissolution, liquidation, winding-up, bankruptcy or reorganization of LMGC.

(b) Corporate Power of LMGC and Due Authorization

LMGC has the corporate power and capacity to enter into, and to perform its obligations under, this Agreement. Each of this Agreement, the Ancillary Agreements and any other agreements, contracts and instruments required by this Agreement to be delivered by LMGC at Closing has been duly authorized by LMGC. This Agreement has been duly executed and delivered by LMGC and is a valid and binding obligation of LMGC, enforceable in accordance with its terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies. At the Time of Closing, the Ancillary Agreements will be duly executed and delivered by LMGC and will be valid and binding obligations of LMGC, enforceable in accordance with their respective terms, subject to the usual exceptions as to bankruptcy and the availability of equitable remedies.

(c) Consents and Approvals of LMGC

Except for the required notification and approval under the *Competition Act* (Canada) described in Section 4.1.6, no authorization, consent, or approval of, or filing with or notice to, any governmental authority, regulatory body or court is required in connection with the execution, delivery or performance of this Agreement by LMGC, or the purchase of the Partnership Interest hereunder.

38

3.3 No Finder's Fees

Each of the Vendor, on one hand, and the Purchasers, on the other hand, represents and warrants to the other that such party has not taken, and agrees that it will not take, any action that would cause the other party or the Purchased Entities to become liable to any claim or demand for a brokerage commission, finder's fee or other similar payment. For greater certainty, the Vendor acknowledges that all fees or other amounts payable to Yorkton Securities Inc. in connection with the purchase and sale of the Purchased Interests or any other matters contemplated by this Agreement shall be at the sole expense of the Vendor and not any of the Purchased Entities or the Purchasers.

3.4 Survival of Representations, Warranties and Covenants

3.4.1 Vendor's Representations and Warranties

The representations and warranties of the Vendor contained in this Agreement (including without limitation those representations and warranties set out in Section 3.1) and in all certificates and documents delivered pursuant to or contemplated by this Agreement shall survive the Closing and shall survive for a period of twenty (20) months from the Closing Date, and notwithstanding the Closing and any inspection or inquiries made by or on behalf of the Purchasers, shall continue in full force and effect for the benefit of the Purchasers, after which time the Vendor shall be released from all obligations in respect of such representations and warranties except with respect to any Claims asserted by

either of the Purchasers in writing (setting out in reasonable detail the nature of the Claim and the approximate amount of such Claim) before the expiration of such period, provided, however, that:

- (a) there shall be no time limit on the representations and warranties set out in Sections 3.1.6 to 3.1.8 and 3.1.10 to 3.1.12 inclusive;
- (b) the representations and warranties set out in Section 3.1.38 (Tax Matters), and the representations and warranties set out in Section 3.1.9 (Reorganization) to the extent they relate to Governmental Charges and Taxes, shall continue in full force and effect after Closing for the benefit of the Purchasers in respect of each taxation year of each of the Purchased Entities to which it relates until the expiration of the applicable time period within which Canada Customs and Revenue Agency (or any other applicable taxing authority) may issue a reassessment in respect of that taxation year under the *Income Tax Act* (Canada) (or any other applicable Governmental Charges and Taxes legislation); and
- (c) no claim for breach of representation or warranty shall be valid unless the party against whom such claim is made has been given notice thereof before the date on which the applicable representation or warranty shall have terminated in accordance with the foregoing.

39

3.4.2 **Purchasers' Representations and Warranties**

The representations and warranties of the Purchasers contained in this Agreement (including without limitation those representations and warranties set out in Section 3.2) and in all certificates and documents delivered pursuant to or contemplated by this Agreement shall survive the Closing for a period of twenty (20) months from the Closing Date, and notwithstanding the Closing, shall continue in full force and effect for the benefit of the Vendor, after which time each of the Purchasers shall be released from all obligations in respect of such representations and warranties except with respect to any Claims asserted by the Vendor in writing (setting out in reasonable detail the nature of the Claim and the approximate amount of such Claim) before the expiration of such period, provided, however that:

- (a) there shall be no time limit on the representations and warranties set out in Sections 3.2.1 and 3.2.2; and
- (b) no claim for breach of representation or warranty shall be valid unless the party against whom such claim is made has been given notice thereof before the date on which the applicable representation or warranty shall have terminated in accordance with the foregoing.

3.4.3 Covenants

The covenants of each of the Purchasers and the Vendor set forth in this Agreement shall survive the completion of the purchase and sale of the Purchased Interests herein provided for and, notwithstanding such completion, shall continue in full force and effect for the benefit of the other party in accordance with the terms thereof.

ARTICLE 4 CONDITIONS

4.1 Conditions for the Benefit of the Purchasers

The obligation of the Purchasers to complete the purchase of the Purchased Interests pursuant to this Agreement is subject to the satisfaction of, or compliance with, at or prior to the Time of Closing, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Purchasers); and the Vendor agrees with the Purchasers to take all such actions, steps and proceedings as are reasonably within its control as may be necessary to ensure that the following conditions are fulfilled at or before the Time of Closing:

4.1.1 Closing Documents and Proceedings

All documents relating to the authorization and completion of the transactions contemplated by this Agreement to be executed and delivered (including without limitation those listed in Section 7.2), and all actions and proceedings to be taken, at or prior to the Time of Closing in connection with the performance by the Vendor of its obligations under this Agreement shall be reasonably satisfactory to the Purchasers and to the Purchasers' Counsel and each of the Purchasers shall have received copies of all such documents and evidence that all such actions and proceedings have been taken as it may reasonably request, in form and substance satisfactory to the Purchasers and the Purchasers' Counsel, acting reasonably.

40

4.1.2 Delivery of Share Certificates and Evidence of Transfer of the Purchased Partnership Interest

The Vendor shall have delivered to ADS share certificates representing the Purchased Shares duly endorsed for transfer and to LMGC evidence satisfactory to LMGC of the transfer of the Purchased Partnership Interest.

4.1.3 **Ancillary Agreements**

The Vendor and the Purchased Entities, as applicable, shall have executed and delivered, or caused to be executed and delivered, to each of the Purchasers, as applicable, the Ancillary Agreements.

4.1.4 Release from the Vendor and its Affiliates

The Vendor shall provide on its own behalf and on behalf of its Affiliates (other than the Purchased Entities and Union Gas Limited), and shall cause Union Gas Limited to provide, to each of the Purchased Entities a full and final release of all claims relating to all matters prior to the Closing, except for claims arising under this Agreement or any Ancillary Agreement and claims arising in respect of the trade payables owing to the Vendor or any of its Affiliates that were incurred in the Ordinary Course of the Business in connection with services performed.

4.1.5 **No Litigation**

There shall be no litigation or proceedings:

- (a) pending or threatened against any of the Purchased Entities, any of the parties hereto or against any of their respective Affiliates or any of their respective directors or officers, or involving the assets and properties of any of them, for the purpose of enjoining, preventing or restraining the completion of the transactions contemplated by this Agreement; or
- (b) pending or threatened against any of the Purchased Entities, any of the parties hereto or against any of their respective Affiliates or any of their respective directors or officers which:
 - (i) in the result, could adversely affect the right of the Purchasers to acquire or retain any of the Purchased Interests; or
 - (ii) if successful, would have a Material Adverse Effect on the Purchased Entities.

4.1.6 Competition Act

Notification shall have been made pursuant to Part IX of the *Competition Act* (Canada) (the "**Act**") and the applicable waiting period shall have expired and either: (i) an advance ruling certificate pursuant to Section 102 of the Act to the effect that the Director of Investigation and Research under the Act does not at that time intend to make application to the Competition Tribunal under Section 92 of the Act in respect of the transactions contemplated in this Agreement has been obtained; or (ii) written notification from the Commissioner appointed under the Act (the "**Commissioner**") that he does not intend to make an application under Section 92 of the Act in respect of the proposed transaction (a "no-action letter") has been obtained.

4.1.7 Consents and Approvals

The consents and approvals required under the Master Agreement between Enlogix CIS L.P. and Telus Enterprise Solutions Inc. dated August 1, 2000 shall have been obtained.

41

4.1.8 **Oracle Assignment**

The Vendor shall have delivered, or caused to be delivered, to the Purchasers the Oracle Assignment. In the event that at the Time of Closing, the Vendor has not delivered, or caused to be delivered, to the Purchasers the Oracle Assignment, the Assignment Withheld Amount shall be \$200,000.00; otherwise, the Assignment Withheld Amount shall be \$0.00.

4.1.9 **Westcoast Oracle Certificate**

The Vendor shall have delivered, or caused to be delivered, to the Purchasers the Westcoast Oracle Certificate. In the event that at the Time of Closing, the Vendor has not delivered, or caused to be delivered, to the Purchasers the Westcoast Oracle Certificate, the Certificate Withheld Amount shall be \$625,000.00; otherwise, the Certificate Withheld Amount shall be \$0.00.

4.2 Conditions for the Benefit of the Vendor

The obligation of the Vendor to complete the sale of the Purchased Interests hereunder is subject to the satisfaction of, or compliance with, at or prior to the Time of Closing, each of the following conditions (each of which is acknowledged to be for the exclusive benefit of the Vendor); and each of the Purchasers agrees with the Vendor to take all such actions, steps and proceedings as are reasonably within its control as may be necessary to ensure that the following conditions are fulfilled at or before the Time of Closing:

4.2.1 Payment of Purchase Price

ADS shall have paid the portion of the Share Purchase Price payable on Closing in accordance with Section 2.2.1 and LMGC shall have paid the portion of the Partnership Interest Purchase Price payable on Closing in accordance with Section 2.2.2.

4.2.2 Closing Documents and Proceedings

All documents relating to the authorization and completion of the transactions contemplated by this Agreement to be executed and delivered (including without limitation those listed in Section 7.2), and all actions and proceedings to be taken, at or prior to the Time of Closing in connection with the performance by each of the Purchasers of its obligations under this Agreement shall be reasonably satisfactory to the

Vendor and to the Vendor's Counsel and the Vendor shall have received copies of all such documents and evidence that all such actions and proceedings have been taken as it may reasonably request, in form and substance satisfactory to the Vendor and the Vendor's Counsel, acting reasonably.

4.2.3 **Ancillary Agreements**

Each of the Purchasers, as applicable, shall have executed and delivered to the Vendor and the Purchased Entities, as applicable, the Ancillary Agreements.

4.2.4 **No Litigation**

There shall be no litigation or proceedings pending or threatened against any of the Purchased Entities, any of the parties hereto, or against any of their respective Affiliates, or any of their respective directors or officers, or involving the assets or properties of any of them, for the purpose of enjoining, preventing or restraining or prohibiting the completion of the transactions contemplated by this Agreement.

42

4.2.5 **Competition Act**

Notification shall have been made pursuant to Part IX of the Act and the applicable waiting period shall have expired and either: (i) an advance ruling certificate pursuant to Section 102 of the Act to the effect that the Director of Investigation and Research under that Act does not at that time intend to make application to the Competition Tribunal under Section 92 of the Act in respect of the transactions contemplated in this Agreement has been obtained; or (ii) written notification from the Commissioner that he does not intend to make an application under section 92 of the Act in respect of the proposed transaction (a "no-action letter") has been obtained.

4.2.6 Consents and Approvals

The consents and approvals required under the Master Agreement between Enlogix CIS L.P. and Telus Enterprise Solutions Inc. dated August 1, 2000 shall have been obtained.

ARTICLE 5 ADDITIONAL COVENANTS AND AGREEMENTS OF THE PARTIES

5.1 Corporate Action and Resignations

At or prior to the Time of Closing, the Vendor shall cause all necessary action or corporate action, as applicable, to be taken for the purpose of approving the transfer of the Purchased Interests to the Purchasers. At or prior to such time, the Vendor shall also obtain (a) the resignations of each of the directors and officers of the Purchased Entities, and (b) releases from, each of the directors and officers of the Purchased Entities other than Anthony Haines, Don Critchley, John McKinnon, Mark Board and Janet Fraser.

5.2 Bank Accounts

At or prior to the Time of Closing, the Vendor shall: (a) take, or cause to be taken, all actions necessary to terminate the involvement of the bank accounts described in Schedule 3.1.44 in the Vendor's cash concentration account program effective as of the Time of Closing; and (b) ensure that sufficient funds are remaining in the bank accounts described in Schedule 3.1.44 to satisfy any and all cheques, money orders, bank drafts, electronic funds transfers or other commitments to pay that are outstanding as of the Time of Closing and that can be withdrawn from such bank accounts by the payee or its agent within 15 days of the Closing Date without any further action on the part of any of the Purchased Entities. For greater certainty, the parties acknowledge and agree that the funds remaining in the bank accounts described in Schedule 3.1.44 as required by this Section 5.2 shall be included in the calculation of the value of the current assets of the Purchased Entities for purposes of Sections 2.3 and 2.4.

5.3 <u>Undertaking to Discharge Security Interests</u>

The Vendor undertakes, at the expense of the Vendor and with the co-operation of, and assistance from, the Purchasers, to obtain a discharge of the security interest registered against the Corporation and its Assets by Jaguar Credit Canada, a Division of Primus AFS under the registration number 20010306 1754 1531 8237 as soon as reasonably practicable after Closing, but in no event shall such discharge be obtained any later than thirty (30) days following the Closing Date.

5.4 Repayment of Loan

At or prior to the Time of Closing, the Vendor shall cause the loan made by Enlogix CIS L.P. to Anthony Haines and Lea Campbell to be repaid and satisfied in full.

43

5.5 <u>Termination of Employees and Consulting Arrangements</u>

At or prior to the Time of Closing: (a) the Vendor shall cause the Purchased Entities to terminate the employment of Anthony Haines, Don Critchley and John McKinnon in accordance with the terms and conditions of the Human Resources Agreement; and (b) the Consulting Agreements shall be executed by each

of the parties thereto. The Purchasers agree to notify the Vendor in the event that the Purchasers or Enlogix CIS L.P. decides to (a) terminate either or both of the Consulting Agreements, or (b) amend the Consulting Agreement to reflect a material adverse change in the terms and conditions thereof.

5.6 Obtaining of Consents and Approvals

If any of the consents and approvals described in <u>Schedule 3.1.15</u> (other than the consents and approvals described in Sections 4.1.6, 4.1.7, 4.2.5 and 4.2.6) have not been obtained as of the Closing Date, (a) the Purchasers shall continue to pursue such consents after the Closing Date; and (b) the Vendor shall cooperate and provide such assistance as is reasonably requested by the Purchasers from time to time in obtaining such consents.

5.7 Termination of Intercompany Services and Transfer of Assets and Licences

- 5.7.1 At the Time of Closing, the parties agree that, except as set out in the Ancillary Agreements, the Intercompany Services shall terminate effective immediately.
- 5.7.2 The Purchasers shall cause the Purchased Entities to provide reasonable access, in a manner that is not disruptive to the Business and any other operations of the Purchased Entities, to the Canadian Employees and to the United States Employees (both as defined in the Human Resources Agreement) at such times as are necessary for the purpose of facilitating payment of bonuses and other compensation payable pursuant to the Incentive Plans including the Enlogix Retention and Sales Incentive Plan (as defined in the Human Resources Agreement), which access shall be provided until such time as such bonuses and other compensation are paid out in accordance with the terms of such Incentive Plans.
- 5.7.3 The Vendor undertakes, at the expense of the Vendor and with the co-operation of, and assistance from, the Purchasers, to transfer to Enlogix CIS L.P. the Microsoft software licences for desktop applications currently used by or made available to Enlogix CIS L.P. as soon as reasonably practicable after the Closing Date, but in no event shall such transfer take place any later than one hundred and twenty (120) days following the Closing Date. The Vendor shall deliver to the Purchasers a copy of the form that will be submitted to the licensor of such Microsoft software licences requesting that such Mircosoft software licences be transferred to Enlogix CIS L.P.
- 5.7.4 (a) The Vendor agrees that, to the extent there are any assets (other than contracts, licenses and other agreements with third party service providers (the "Third Party Contracts")) that are used by the Purchased Entities and necessary to carry on the Business as carried on by the Purchased Entities immediately prior to the Closing, that are not owned by the Purchased Entities or to which the Purchased Entities do not have the right to use or enjoy the benefit of following the Closing and which are identified to the Vendor as assets that meet the criteria described in this Section 5.7.4(a) within one (1) year following the Closing Date, the Vendor shall, as soon as reasonably practicable after such asset has been identified, transfer such assets to the Purchased Entities or enter into any reasonable arrangement designed to provide the ownership, use or benefit of such assets to or for the benefit of the Purchased Entities.

44

- (b) The Vendor agrees that, to the extent there are any Third Party Contracts, including any support, service, maintenance and administration agreements, equipment leases or software licenses, that are used by the Purchased Entities and necessary to carry on the Business as carried on by the Purchased Entities immediately prior to the Closing, that are not owned by or in the name of the Purchased Entities or to which the Purchased Entities do not have the right to use or enjoy the benefit of following the Closing and which are identified to the Vendor as Third Party Contracts that meet the criteria described in this Section 5.7.4(b) within one (1) year following the Closing Date, the Vendor shall, as soon as reasonably practicable after such Third Party Contract has been identified, transfer or assign such Third Party Contracts to the Purchased Entities to the extent that they relate to the Purchased Entities in accordance with the terms of such Third Party Contracts; if the Vendor is unable to transfer or assign any such Third Party Contract, it shall use commercially reasonable efforts to assist and facilitate the Purchased Entities to enter into a reasonable alternative arrangement designed to provide substantially similar rights and benefits to or for the benefit of the Purchase Entities on substantially similar terms and conditions (including pricing applicable for a period similar to the remaining term of the applicable Third Party Contract); and, if the Purchased Entities are unable to enter into such alternative arrangement, the Vendor shall enter into any reasonable arrangement designed to provide the benefits of such Third Party Contract to or for the benefit of the Purchased Entities.
- (c) The Parties acknowledge and agree that nothing in this Section 5.7.4 shall: (i) qualify, or place any limitation on, any representation or warranty made by the Vendor under this Agreement; or (ii) limit or prejudice the right of the Purchasers to make any Claim for a breach of any representation or warranty made by the Vendor under this Agreement.

5.8 Arrangements with respect to Support Obligations

- 5.8.1 From and after the Time of Closing, ADS hereby assumes and agrees to pay and discharge when due, and promptly upon request of the Vendor, reimburse the Vendor and indemnify and hold the Vendor harmless with respect to any Liabilities incurred by the Vendor from and after the Time of Closing pursuant to the following Support Obligations:
 - (a) the guarantee dated February 23, 1998 given by the Vendor in favour of the City of Calgary and assigned to Enmax Encompass Inc. as of January 1, 2001 in respect of continued services to be provided under the CIS Business Services Agreement dated February 20, 1998 between the City of Calgary and the Corporation; and
 - (b) the guarantee in favour of Duke-Weeks Realty Limited Partnership in respect of the Subsidiary's lease obligations under a lease dated May 21, 2001 between Duke-Weeks Realty Limited Partnership and the Subsidiary providing for the lease of 3655 Brookside Parkway, Alpharetta, Georgia, United States of America, 30022;

provided that (a) such Liabilities relate to, or are in connection with, matters that occurred after the Time of Closing; and (b) ADS' assumption and indemnification obligations in respect of a Support Obligation described in this Section 5.8.1 shall terminate effective immediately without

The Vendor may, at its expense, with such cooperation and assistance from ADS as the Vendor may reasonably request (which cooperation and assistance will be provided at the Vendor's expense), effect the complete and unconditional release of the Vendor and its Affiliates (other than the Purchased Entities) from any and all liabilities and obligations in respect of the guarantees, financial support and other obligations listed in Schedule 5.8.2 (collectively, the "Support Obligations"). For purposes of this Section 5.8.2, the Purchasers shall not be required to pay any monies or give any other consideration, guarantees or financial support in connection with the releases obtained by the Vendor and its Affiliates under this Section 5.8.2 other than providing an acknowledgement to Enmax Encompass Inc. and/or Duke-Weeks Realty Limited Partnership, as applicable, that ADS has assumed all of the liabilities and obligations described in Section 5.8.1.

5.9 Tax Returns for Closing Year

The Vendor shall after the Closing Date prepare all Tax Returns and other returns related to Governmental Charges and Taxes required to be filed by the Purchased Entities for the taxation year ending as a consequence of the Closing, and for any fiscal period beginning before the Closing (including for greater certainty the taxation year of the Corporation ended March 13, 2002 due to an acquisition of control). Such returns shall be prepared in a manner consistent with prior returns filed by the Purchased Entities. Such returns shall be provided to the Purchasers for review at least 45 days prior to their due date (except for the returns due September 13, 2002). In addition, the Vendor may prepare amended income tax returns for the Purchased Entities for years ended prior to Closing for the purpose of delivering the undepreciated capital cost required by Section 3.1.38(f). Provided that the Purchasers are satisfied that such returns are properly completed, they shall cause the Purchased Entities to sign and file such returns in a timely manner. If the Purchasers are not satisfied, the Purchasers and Vendor agree to appoint one of the four large accounting firms to advise them on the proper manner of completion of the returns. The third party cost of such a review shall be shared equally by the Vendor and the Purchasers. The Purchasers shall make available all documents and records and access to personnel of the Purchased Entities necessary to enable the Vendor to prepare such returns, and the Vendor shall co-operate with the Purchasers and the Purchased Entities and make available all documents and records in the Vendor's possession and access to personnel of the Vendor necessary to resolve matters related to Governmental Charges and Taxes with respect to the pre-Closing periods of the Purchased Entities. The Vendor shall pay all Governmental Charges and Taxes that relate to any taxation or fiscal year ending as a consequence of the Closing or before Closing. The parties agree to request approval from the Canada Customs and Revenue Agency and all relevant provincial tax authorities to have a fiscal year end of Enlogix CIS L.P. immediately prior to the Closing Date. If such approval is not received, the Vendor shall be allocated 99% of the taxable income or loss of Enlogix CIS L.P. for the period from the first day of its current fiscal year (i.e. January 1, 2002) to the Closing Date (with any capital cost allowance claimed for the year being allocated between such period and the remainder of the year in the manner agreed by the Vendor and the Purchasers and consistent with the requirement of Section 3.1.38(f)), notwithstanding that the Vendor may no longer be a partner at the end of its current fiscal year or any subsequent fiscal year, and the parties hereto and the Purchased Entities agree to amend the agreement governing Enlogix CIS L.P. and to enter into any other agreements as may be necessary or desirable to ensure this result. The Vendor shall pay the amount of any Governmental Charges and Taxes relating to the pre-Closing portion of a fiscal period of any of the Purchased Entities (including Enlogix CIS L.P.) beginning before but ending after the Closing. Also in the event such approval is not received, the Vendor shall reimburse the Purchased Entities for 50% of the third party costs associated with preparation of tax statements or returns for Enlogix CIS L.P. for such period, and the completion of the allocation of taxable income or loss described above.

46

5.10 Covenant Not to Compete

- 5.10.1 In order to ensure that the Purchasers receive the expected benefits of acquiring the Purchased Interests, for a period of three (3) years after the Closing Date, the Vendor shall not, and shall ensure that its Affiliates do not, individually or in partnership or jointly or in conjunction with any other Person, as principal, agent, shareholder, consultant, trustee or in any other manner whatsoever, whether directly or indirectly, invest in, own, manage, operate, carry on or be engaged or concerned with or have a financial interest in or advise, lend money to, guarantee the debts or obligations of, or permit its name or any part thereof to be used or employed by or associated with, any Person which is directly or indirectly through an Affiliate or otherwise, engaged in the business of providing billing system or customer care outsourcing solutions for utilities and energy companies in any manner whatsoever (a "Competitive Business") in all or any part of the Territory.
- 5.10.2 Notwithstanding anything to the contrary in this Section 5.10, the Vendor and its respective Affiliates shall not be prohibited from:
 - (a) engaging in a Competitive Business for the sole purpose of providing services to itself or to Duke Energy Corporation or any of its Affiliates (other than Union Gas Limited or Pacific Northern Gas Ltd.);
 - (b) holding shares or securities listed on a recognized stock exchange or traded in an over-the-counter or other recognized market of any Person provided that such holdings do not exceed 5% of the outstanding shares or securities or principal amount of such securities, as applicable, so listed or traded;
 - (c) acquiring assets or securities of a Person where the business conducted by the Person (and its Affiliates, if any) includes a Competitive Business, provided that:
 - (i) the annual gross revenues derived from such Competitive Business when acquired constitute less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) for the financial year of the Person ending immediately before the date of acquisition;
 - (ii) if the annual gross revenues derived from such Competitive Business component of such Person when acquired constitute 10% or

more of the annual gross revenues of such Person (and its Affiliates, if any), the Vendor or its Affiliate, as applicable, disposes of the Competitive Business, subject to Section 5.10.3, within the later to occur of:

- (A) 15 months following the Trigger Date (as defined in Section 5.10.3) (or such longer period as may be reasonably required to obtain any approvals or consents that may be required by Applicable Law, provided that in no event shall such longer period exceed 20 months following the Trigger Date); and
- (B) such longer period as may be reasonably required to dispose of such portion of the Competitive Business component in order to enable the Vendor or its Affiliate, as applicable, to dispose of such Competitive Business component on a commercially reasonable basis, but in no event to exceed 20 months following the Trigger Date;

47

- (iii) if the annual gross revenues derived from such Competitive Business component of such Person when acquired constitute less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) but thereafter constitute more than 10% of the annual gross revenues of such Person (and its Affiliates, if any) determined as of the most recent fiscal quarter end, the Vendor or its Affiliate, as applicable, disposes of the Competitive Business, subject to Section 5.10.3, within the later to occur of:
 - (A) 15 months following the Trigger Date (or such longer period as may be reasonably required to obtain any approvals or consents that may be required by Applicable Law, provided that in no event shall such longer period exceed 20 months following the Trigger Date); and
 - (B) such longer period as may be reasonably required to dispose of such portion of the Competitive Business component in order to enable the Vendor or its Affiliate, as applicable, to dispose of such Competitive Business component on a commercially reasonable basis, but in no event to exceed 20 months following the Trigger Date;
- (d) being acquired, by way of acquisition, merger or other combination or consolidation, by a Person, a component of which is engaged in a Competitive Business, provided that:
 - the annual gross revenues derived from such Competitive Business component of such Person or the resulting combined or consolidated Person constitutes less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) on the date of, but after giving effect to, such acquisition, merger, consolidation or combination; or
 - (ii) if the annual gross revenues derived from such Competitive Business component of such Person constitutes 10% or more of the annual gross revenues of such Person (and its Affiliates, if any) on the date of, but after giving effect to, such acquisition, merger, combination or consolidation, the Vendor or its Affiliate, as applicable, disposes of that portion of the Competitive Business component necessary to ensure that the Competitive Business component of such Person constitutes less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) on the date of such disposition within the later to occur of:
 - (A) 18 months following the date of such acquisition, merger, combination or consolidation (or such longer period as may be reasonably required to obtain any approvals or consents that may be required by Applicable Law, provided that in no event shall such longer period exceed 24 months following the date of such acquisition, merger, combination or consolidation); and
 - (B) such longer period as may be reasonably required to dispose of such portion of the Competitive Business component in order to enable the Vendor or its Affiliate, as applicable, to dispose of such Competitive Business component on a commercially reasonable basis, but in no event to exceed 24 months following the date of such acquisition, merger, combination or consolidation:

- (iii) if the annual gross revenues derived from such Competitive Business component of such Person constitutes less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) on the date of, but after giving effect to, such acquisition, merger, combination or consolidation but thereafter constitutes 10% or more of the annual gross revenues of such Person (and its Affiliates, if any) determined as of the most recent fiscal quarter end, the Vendor or its Affiliate, as applicable, disposes of that portion of the Competitive Business component necessary to ensure that the Competitive Business component of such Person constitutes less than 10% of the annual gross revenues of such Person (and its Affiliates, if any) on the date on which the obligation to sell is triggered under this Section 5.10.2(d)(iii) within the later to occur of:
 - (A) 18 months following the date on which the obligation to sell is triggered under this Section 5.10.2(d)(iii) (or such longer period as may be reasonably required to obtain any approvals or consents that may be required by Applicable Law, provided that in no event shall such longer period exceed 24 months following the date on which the obligation to sell is triggered under this Section 5.10.2(d)(iii)); and
 - (B) such longer period as may be reasonably required to dispose of such portion of the Competitive Business component in order to enable the Vendor or its Affiliate, as applicable, to dispose of such Competitive Business component on a commercially reasonable basis, but in no event to exceed 24 months following the date on which the obligation to sell is triggered under this Section 5.10.2(d)(iii);
- (e) in the case of an Affiliate of the Vendor where the Vendor or Duke Energy Corporation does not exercise legal control, engaging in a

- (f) entering into joint ventures, partnerships or other business relationships with Persons who carry on a Competitive Business so long as such joint venture, partnership or business relationship is not involved in a Competitive Business; or
- (g) carrying on its business in the Ordinary Course with Persons who carry on a Competitive Business where the business in the Ordinary Course (which, for greater certainty, includes any of the matters included in Section 5.10.1) is unrelated to the Competitive Business.
- 5.10.3 Promptly following an acquisition under Section 5.10.2(c), the Vendor or its Affiliate, as applicable, will give written notice to the Purchasers of the acquisition. If the Purchasers express an interest in purchasing the Competitive Business component within 10 days following receipt of the notice, the parties will negotiate in good faith for 30 days following the Purchaser's expression of interest for the purpose of selling the Competitive Business component (the end of such 30-day period being the "**Trigger Date**").
- 5.10.4 Notwithstanding anything to the contrary in this Section 5.10, Duke Energy Corporation and its Affiliates shall not be prohibited from:
 - (a) providing, in North Carolina and South Carolina, the services that they currently provide to their customers in North Carolina and South Carolina, which services, the Vendor acknowledges and agrees, do not include billing or invoicing system services that are provided to utilities or energy companies; or
 - (b) engaging in a Competitive Business in North Carolina and South Carolina, provided that Duke Energy Corporation does not enter into any arrangement to provide services to (i) Progress Energy or Scana Corporation or any successor thereof, or (ii) any other utility or energy company in North Carolina or South Carolina with a comparable number of customer accounts, as part of such Competitive Business.

5.11 Non-Solicitation

- 5.11.1 The Vendor shall not, and shall ensure that its Affiliates do not, at any time within the three (3) year period following the Closing Date, in any manner whatsoever, hire or offer to hire or entice away or in any other manner persuade any employee, to discontinue any such Person's relationship with any of the Purchased Entities or the Purchasers, except that nothing in this Section 5.11.1 will prevent the Vendor or any of its Affiliates from offering to hire or soliciting any employer, consultant or independent contractor through advertising in a newspaper or periodical or trade publication of general circulation, or employing any such employee, consultant or independent contractor who responds to the solicitation or who contacts the Vendor or any of its Affiliates on his or her own initiative.
- 5.11.2 The Vendor shall not, and shall ensure that its Affiliates do not, at any time within the three (3) year period following the Closing Date, in any manner whatsoever, entice away or persuade a Customer to cease or refrain from doing business with, or reduce the amount of business which any such Customer has historically done, with the Purchased Entities. For purposes of this Section 5.11.2, "Customer" means any Person that (A) as of the relevant date is a customer of the Business or has received a proposal or bid for new business from any of the Purchased Entities, or (B) was a customer of the Business or received a proposal or bid for new business from any of the Purchased Entities within the immediately preceding twelve (12) month period.

50

5.12 Confidentiality

The Vendor acknowledges that it has had access to confidential and proprietary information and trade secrets ("Confidential Information") relating to the Purchased Entities and the Business, including financial information and information relating to the present and contemplated services, customer lists, prospects, marketing, and confidential information relating to customers and consumers associated with the Business, the disclosure of any of which to competitors of the Business or to the general public would be detrimental to the best interest of the Purchasers. The Vendor acknowledges and agrees with the Purchasers that the right to maintain the confidentiality of such Confidential Information, and the right to preserve the goodwill of the Business, constitute proprietary rights which the Purchasers are entitled to protect. Accordingly, the Vendor covenants and agrees with each of the Purchasers that it will not at any time hereafter disclose to any Person, or use for its own benefit, any of such Confidential Information (other than information which is in the public domain at the time of such disclosure or information which subsequently comes into the public domain without breach by the Vendor under this Agreement or any of the Ancillary Agreements).

Nothing in this Section 5.12 shall preclude the Vendor from disclosing, without liability, the Confidential Information if (and to the extent that) such disclosure is required by Applicable Law. If the Vendor is required to make such disclosure, it shall:

- 5.12.1 to the extent not prohibited by law, give the Purchasers and the Purchased Entities prompt notice of the requirement so that the Purchasers and the Purchased Entities may have the opportunity to seek an appropriate protective order or pursue such legal action, remedy or assurance as it deems necessary to preserve the confidentiality of the Confidential Information;
- 5.12.2 refrain from opposing, and co-operate at the Purchaser's expense in the obtaining of, any action by the Purchasers or the Purchased Entities to obtain a protective order or other remedy or assurance;
- 5.12.3 take all reasonable steps (after consultation with the Purchasers and at the Purchasers' request and expense) to preserve the confidential nature of the Confidential Information, including requesting that the Confidential Information not be released to third parties or the public and disclosing only that portion of the Confidential Information that is, in the opinion of its counsel, legally compelled to disclose; and

to the extent not prohibited by law, provide the Purchasers and the Purchased Entities with as much prior notice as is reasonably possible of the fact and proposed content of any disclosure and co-operate in ensuring consistency in disclosure and communications.

5.13 Remedies

The Vendor acknowledges that a breach by it of any of the covenants contained in Sections 5.10, 5.11 and 5.12 of this Agreement would result in damages to the Purchasers and that the Purchasers may not be adequately compensated for such damages by monetary award alone. Accordingly, the Vendor agrees that in the event of any such breach, in addition to any other remedies available at law or otherwise, each of the Purchasers shall be entitled as a matter of right to apply to a court of competent jurisdiction for relief by way of injunction, restraining order, decree or otherwise as may be appropriate to ensure compliance by the Vendor with such covenants. Any remedy expressly set out in this Agreement shall be in addition to and not inclusive of or dependent upon the exercise of any other remedy available at law or otherwise.

51

5.14 Reasonableness of Restrictions

The Vendor agrees that all restrictions set out in Sections 5.10, 5.11 and 5.12 of this Agreement are necessary and fundamental to the protection of the respective businesses of the Purchased Entities and the Purchasers and are reasonable and valid. All defences to the strict enforcement of such provisions against the Vendor are hereby waived.

5.15 Consent to Sublease

The Vendor undertakes to and in favour of the Purchasers to use commercially reasonable efforts to obtain the consent of the landlord under the Head Lease (as defined in the Sublease) to the Sublease and the waiver of the landlord's right of first offer pursuant to Section 12.04 of the Head Lease in the form previously provided to the landlord within ninety (90) days following the Closing Date. In the event that, within the ninety (90) day period following the Closing Date, the landlord under the Head Lease; (a) does not provide such consent to the Sublease, or a consent to the Sublease that is otherwise in form and substance satisfactory to the Purchasers and the Purchasers' Counsel, acting reasonably; (b) indicates that it will not provide such a consent; (c) does not waive its right of first offer pursuant to Section 12.04 of the Head Lease; (d) terminates the Head Lease in accordance with its terms; or (e) requests that Enlogix CIS L.P. vacate the Sublease Premises (as defined in the Sublease), Enlogix CIS L.P. shall have the right to terminate the Sublease without penalty upon written notice to Union Gas Limited and the Vendor and the Vendor shall indemnify Enlogix CIS L.P. for any and all Liability that Enlogix CIS L.P. may incur arising or resulting from, or relating to, the failure to obtain that unconditional consent of the landlord under the Head Lease to the Sublease and the waiver of the landlord's right of first offer pursuant to Section 12.04 of the Head Lease in accordance with this Section 5.1.15, including all costs and expenses incurred by Enlogix CIS L.P. in connection with moving their operations to new premises (which new premises will be comparable in size, location and building quality to the Subleased Premises) and any excess rent or other payments required to be paid by Enlogix CIS L.P. in connection with any new premises. In the event that Enlogix CIS L.P. exercises its right to terminate the Subleased Premises.

ARTICLE 6 INDEMNIFICATION

6.1 Indemnification by the Vendor

The Vendor shall indemnify and save each of the Purchasers and their respective directors, officers, employees, agents, representatives and Affiliates (including, without limitation, the Purchased Entities), harmless from and against any and all Claims which may be made or brought against an Indemnified Party or which it may suffer or incur directly or indirectly as a result of, in respect of or arising out of:

- 6.1.1 any incorrectness in or breach of any representation or warranty of the Vendor contained in this Agreement or under any other agreement, certificate or instrument executed and delivered pursuant to this Agreement;
- any breach of or any non-fulfilment of any covenant or agreement on the part of the Vendor under this Agreement, including under any indemnity provisions, or under any other agreement, certificate or instrument executed and delivered pursuant to this Agreement;

- any and all Liabilities relating to or arising in connection with the Reorganization including (i) any and all Liabilities, indebtedness or debts (including without limitation, Liabilities for any and all Governmental Charges and Taxes) of any nature whatsoever, incurred or arising as a result of the Reorganization, including any Liabilities which, but for the Reorganization, would not have been a liability, obligation, indebtedness or debt of any of the Purchased Entities and any other expenses related thereto; and (ii) any and all Liabilities attributable to any step of the Reorganization, the result of which is to cause any of the Purchased Entities to be in breach of any obligation, duty or responsibility to another Person, and (iii) any obligation to pay any employee, officer, director or shareholder of any of the Purchased Entities a bonus or similar payment;
- 6.1.4 Enlogix CIS L.P.'s obligations under Section 12 "Intellectual Property Right Infringement Indemnification" of the SCT Development Agreement, except to the extent Enlogix CIS L.P.'s obligations under such Section 12 result from or arise out of the use of any work product created by Enlogix CIS L.P. to SCT after the Closing Date.
- 6.1.5 any income, sales or other Governmental Charges and Taxes (including interest and penalties in respect thereof) payable by or collectible by or from any of the Purchased Entities (i) in respect of any period ending on or before the Closing Date, or the pre-Closing portion of any period

beginning before but ending after the Closing Date, and not accrued in the Financial Statements or (ii) as a result of any assessment or reassessment of Governmental Charges and Taxes by any taxing authority after the Closing Date relating to any period ending on or prior to the Closing Date, or to the pre-Closing portion of any period beginning before but ending after the Closing Date; for greater certainty, this indemnity shall cover Governmental Charges and Taxes with respect to the income of Enlogix CIS LP earned by it prior to the Closing Date (computed on the assumption that Enlogix CIS LP had a fiscal year end for purposes of Governmental Charges and Taxes on the Closing Date) regardless of whether such income is allocated to the Corporation, to the Purchasers or their respective Affiliates or to another Person and regardless of whether such income is included in a taxation year ending on the Closing Date nor any prior or subsequent taxation year, but excludes any liability for tax under the *Land Transfer Tax Act* (Ontario), *Retail Sales Tax Act* (Ontario) or *Excise Tax Act* (Canada) arising by reason of the transfer of the Purchased Entities to the Purchasers pursuant to this Agreement other than retail sales tax payable by virtue of a failure by the Vendor or the Purchased Entities to pay retail sales taxes for which they were liable in a pre-Closing period;

- 6.1.6 any Liability that any of the Purchased Entities may have to any of its employees in respect of any period or partial period up to and including the Closing Date, other than in respect of salaries, wages, vacation pay, short term incentive plan payments, and other compensation that has been accrued in the Ordinary Course;
- 6.1.7 any Liability associated with the litigation disclosed in <u>Schedule 3.1.46</u>;
- 6.1.8 any Liability that any of the Purchased Entities may have to Anthony Haines or Don Critchley in their capacity as directors or officers, or former directors and officers, of the Purchased Entities;
- 6.1.9 the security interest registered against the Corporation and its Assets by Jaguar Credit Canada, a Division of Primus AFS under the registration number 20010306 1754 1531 8237 or any motor vehicles leased (by or on behalf of) or otherwise provided to Anthony Haines in connection therewith; and
- 6.1.10 any and all acts, suits, proceedings, demands, assessments, judgments, legal fees, costs and expenses incident to any of the foregoing.

53

6.2 Indemnification by the Purchasers

The Purchasers shall jointly and severally indemnify and save the Vendor and its directors, officers, employees, agents, representatives and Affiliates harmless from and against any and all Claims which may be made or brought against an Indemnified Party or which it may suffer or incur directly or indirectly as a result of, in respect of or arising out of:

- 6.2.1 any incorrectness in or breach of any representation or warranty of the Purchasers contained in this Agreement or in any other agreement, certificate or instrument executed and delivered pursuant to this Agreement;
- 6.2.2 any breach or non-fulfilment of any covenant or agreement on the part of the Purchasers under this Agreement or under any other agreement, certificate or instrument executed and delivered pursuant to this Agreement; and
- 6.2.3 any and all acts, suits, proceedings, demands, assessments, judgments, legal fees, costs and expenses incident to any of the foregoing.

6.3 Notice of Claim

If an Indemnified Party receives notice of the commencement of any proceeding (as defined in Section 3.1.46) against it in respect of a Claim for which indemnification is provided for pursuant to this Article 6, the Indemnified Party shall give written notice to the Indemnifying Party of the Claim within a reasonable period of time, provided that in the event of a failure to give such notice, such failure shall not preclude the Indemnified Party to obtain such indemnity, except to the extent that the Indemnifying Party demonstrates that the defence of such proceeding is prejudiced by the Indemnifying Party's failure to give such notice. Such notice shall specify whether the Claim arises as a result of a claim by a Person other than the Indemnified Party (a "Third Party Claim"), and shall also specify with reasonable particularity (to the extent that the information is available):

- 6.3.1 the factual basis for the Claim; and
- 6.3.2 the amount of the Claim, or, if an amount is not then determinable, an approximate and reasonable estimate of the likely amount of the Claim.

Any indemnity payments described in this Section 6.3 shall include interest on such amounts at the Prime Rate, from the date on which the event giving rise to the Claim by the Indemnified Party occurred to the date of payment and shall be paid in full by the Indemnifying Party within thirty (30) days after the determination of the amount of the indemnity payment.

54

6.4 Defence of Third Party Claims

The Indemnifying Party shall have the right, at its own expense, to participate in or assume control of the negotiation, settlement or defence of a Third Party Claim by giving notice to that effect to the Indemnified Party not later than thirty (30) days after receiving notice of the Third Party Claim, provided the Indemnifying Party acknowledges in writing its obligation to indemnify the Indemnified Party in accordance with the terms contained herein in respect of the Third Party Claim. The Indemnifying Party's right to do so shall be subject to the rights of any insurer in respect of that Third Party Claim. The Indemnifying Party agrees to pay all of its own expenses of participating in or assuming such defence. The Indemnified Party shall cooperate in good faith in the defence of each Third Party Claim, but at the expense of the Indemnifying Party with

respect to any out-of-pocket expenses incurred, if the defence has been assumed by the Indemnifying Party and may participate in such defence assisted by counsel of its own choice at its own expense. Until the Indemnified Party receives notice that the Indemnifying Party has elected to assume the defence of such Third Party Claim, the Indemnified Party may assume such defence, assisted by counsel of its own choosing and the Indemnifying Party shall be liable for all reasonable costs and expenses paid or incurred in connection therewith and any Losses suffered or incurred by the Indemnified Party with respect to such Third Party Claim. If an Indemnifying Party elects to assume the defence of any Third Party Claim as provided in this Section 6.4.1, the Indemnifying Party shall not be liable for any legal expenses subsequently incurred by the Indemnified Party in connection with the defence of such Third Party Claim.

6.4.2 If the Indemnifying Party, having elected to assume control as contemplated in Section 6.4.1 thereafter fails to defend any such Third Party Claim within a reasonable time, the Indemnified Party shall be entitled to assume such control and the Indemnifying Party shall (i) be liable for all losses suffered or incurred by the Indemnified Party in connection with such Third Party Claim; and (ii) be bound by the results obtained by the Indemnified Party with respect to such Third Party Claim.

6.5 Assistance for Third Party Claims

The Indemnifying Party and the Indemnified Party shall use all reasonable efforts to make available to the party which is undertaking and controlling the defence of any Third Party Claim (the "**Defending Party**"):

- 6.5.1 those employees whose assistance, testimony or presence is necessary to assist the Defending Party in evaluating and in defending any Third Party Claim; and
- all documents, records and other materials in the possession of such party reasonably required by the Defending Party for its use in defending any Third Party Claim, and shall otherwise co-operate with the Defending Party. The Indemnifying Party shall be responsible for all reasonable expenses associated with making such documents, records and materials available and for all reasonable expenses of any employees made available by the Indemnified Party to the Indemnifying Party hereunder, which expenses shall not exceed the actual out-of-pocket cost to the Indemnified Party associated with such employees (excluding normal salaries, wages and benefits).

55

6.6 Additional Rules and Procedures

The obligation of the parties to indemnify each other pursuant to this Article 6 shall also be subject to the following:

- Any Claim arising as a result of a breach of a representation or warranty contained in Article 3 shall be made not later than the date on which, pursuant to Section 3.4, such representation or warranty terminated.
- 6.6.2 Notwithstanding anything contained in this Agreement to the contrary, an Indemnifying Party's obligation to indemnify an Indemnified Party in respect of a breach of a representation or warranty under this Agreement shall not apply unless and until the accumulated aggregate amount of Claims in respect of which it is required to indemnify exceed \$200,000, following which all such accumulated Claims and all further Claims of such Indemnified Party shall be recoverable as provided in this Agreement.
- The Indemnifying Party shall not be responsible for any losses, costs, damages, expenses and liabilities to the extent that (i) the Indemnified Party is otherwise covered for such losses, costs, damages, expenses or liabilities under policies of insurance maintained by the Indemnified Party in the absence of any indemnity hereunder (save and except for losses, costs, damages, expenses and liabilities relating to any deductible applicable under such policies of insurance or any increased premiums payable in connection with such policies of insurance or replacements therefor); (ii) the Indemnified Party chooses to make a claim for such losses, costs, damages, expenses or liabilities under such policies of insurance; and (iii) insurance proceeds are received in respect of such losses, costs, damages, expenses or liabilities.
- 6.6.4 Notwithstanding anything contained in this Agreement to the contrary, in no event shall the aggregate amount of indemnification obligations of the Vendor in respect of breaches of representations or warranties under this Agreement exceed 100% of the Purchase Price.
- Notwithstanding anything to the contrary contained in this Agreement, it is the explicit intent of each party hereto that the Vendor has made and is making no representation or warranty whatsoever, express or implied, written or oral, including any implied representation or warranty as to condition, merchantability, usage, suitability or fitness for any particular purpose with respect to the Purchased Interests, the Purchased Entities or any of the assets of the Purchased Entities, or any part thereof, except those representations and warranties contained in this Agreement. In particular, and without in any way limiting the foregoing, other than those representations and warranties contained in this Agreement, the Vendor makes no representation or warranty to the Purchasers with respect to any financial projection or forecast relating to the Purchased Entities
- In the event that any Third Party Claim is of a nature such that the Indemnified Party is required by Applicable Law to make a payment to any Person (a "Third Party") with respect to such Third Party Claim before the completion of settlement negotiations or related legal proceedings, the Indemnified Party may make such payment and the Indemnifying Party shall, forthwith after demand by the Indemnified Party, reimburse the Indemnified Party for any such payment. If the amount of any liability of the Indemnified Party under the Third Party Claim in respect of which such a payment was made, as finally determined, is less than the amount which was paid by the Indemnifying Party to the Indemnified Party, the Indemnified Party shall, forthwith after receipt of the difference from the Third Party, pay the amount of such difference to the Indemnifying Party.

- 6.6.7 The Indemnified Party shall not permit any right of appeal in respect of any Third Party Claim to terminate without giving the Indemnifying Party notice thereof and an opportunity to contest such Third Party Claim.
- 6.6.8 Notwithstanding Section 6.4, the Indemnifying Party shall not settle any Third Party Claim or conduct any related legal or administrative proceeding in a manner which would, in the opinion of the Indemnified Party, acting reasonably, have a material adverse impact on the Indemnified Party, without obtaining the prior written consent of the Indemnified Party, such consent not to be unreasonably withheld or delayed.

57

6.7 Tax Assessment Procedure

6.7.1 **Pre-Assessment Procedure**

If any of the Purchased Entities receives a notice or other communication, in writing, from any Taxation Authority after Closing to the effect that such Taxation Authority proposes to issue an Assessment to the Purchased Entities with respect to any period commencing prior to the Closing Date, the Purchased Entities shall within a reasonable period of time forward a copy of such notice to the Vendor, or otherwise notify the Vendor of the contents of such communication, and the Vendor shall have the right, at their own expense, including all legal fees, costs and expenses, to require the Purchased Entities to make representations reasonably satisfactory to the Purchasers with respect to such proposed Assessment through counsel retained for this purpose by the Vendor, acting reasonably. The Purchased Entities shall be obligated to cooperate fully at all times with the Vendor and counsel so appointed, and to act in a timely manner, and in good faith, to make available all documents and records and access to personnel of the Purchased Entities necessary to enable the Vendor to make representations with respect to, or otherwise contest, the proposed Assessment. If the Vendor, having received notice of, or other communication in respect of, a proposed Assessment, as contemplated herein, decides not to require the Purchased Entities to make representations with respect to such proposed Assessment, the Vendor shall promptly so inform the Purchased Entities and thereafter the Purchased Entities may do so on their own behalf, and at their own expense. Any decision by the Purchased Entities not to make representations or any lack of success by the Purchased Entities in such action shall not, in such event, relieve the Vendor of their obligation to indemnify the Purchased Entities pursuant to the terms and conditions of this Article, except that such indemnity shall not, in such instance, extend to legal and other costs incurred by the Purchased Entities in connection with such proposed Assessment or Action at any time pri

6.7.2 **Procedure on Assessment**

If any Taxation Authority should at any time issue an Assessment to the Purchased Entities, the Vendor shall pay the amount due (including any interest and penalties) in relation to the Assessment to the Purchased Entities within 10 days before the Purchased Entities are required to pay the Taxation Authority the amount specified in such Assessment or within 10 days after the Purchased Entities have forwarded to the Vendor a claim for indemnity.

58

6.7.3 **Objection/Appeal Procedure**

The Purchased Entities shall, within 30 days after the receipt of an Assessment, send a copy of such Assessment to the Vendor. The failure of the Purchased Entities to notify the Vendor within 30 days after receipt of an Assessment will preclude the Purchasers from claiming indemnity with respect to such Assessment. The Vendor shall be entitled to undertake the carriage of any objection or appeal from such Assessment. The Vendor shall be entitled to retain their own counsel for that purpose and shall be responsible for all expenses, including legal fees and other costs associated therewith. The Purchased Entities shall join in and pursue any objection to or appeal of any such Assessment in a timely manner and in good faith, shall cooperate fully, at all times, with the Vendor and counsel that the Vendor retain and shall make available to the Vendor and such counsel all documents and records, and provide access to personnel, which may reasonably be necessary to enable the Vendor to contest such Assessment. The Purchased Entities shall cooperate with the Vendor to the extent reasonably necessary if the Vendor wish to object to or appeal any such Assessment. The Vendor shall be responsible for all expenses, including legal fees and costs, incurred by the Purchased Entities in so cooperating. If such objection or appeal is successful in whole or in part, the Purchased Entities shall pay to the Vendor the amount of any refund received in respect of such Assessment relating to pre-Closing years together with the applicable portion of interest received on such refund, determined on an after-tax basis, and all legal costs for which the Purchased Entities are reimbursed by the Taxation Authority as a result of such successful objection or appeal.

6.8 Indemnification Claims

The parties agree that Article 6 sets out the sole and exclusive manner by which the Purchasers may seek monetary compensation from the Vendor, or by which the Vendor may seek monetary compensation from the Purchasers, for any matter in respect of which the Purchasers or the Vendor may make a Claim under Sections 6.1 and 6.2 including, for greater certainty, any matter that could be made a Claim under those sections but for Section 6.6.1.

6.9 Duty to Mitigate

The Indemnified Party shall have a duty to mitigate its loss in accordance with Applicable Laws.

6.10 Right of Indemnification Not Affected by Knowledge

59

ARTICLE 7 CLOSING

7.1 Location and Time of the Closing

The Closing shall take place at the Time of Closing on the Closing Date at the offices of Purchasers' Counsel, or such other place as may be agreed upon orally or in writing by the Vendor and the Purchasers.

7.2 Deliveries at the Closing

- 7.2.1 At the Closing, the Vendor shall deliver or cause to be delivered to the Purchasers, as applicable, the following documents and payments:
 - (a) an assignment and transfer of the Purchased Shares in a form acceptable to ADS, including the share certificates representing the Purchased Shares duly endorsed in blank for transfer;
 - (b) evidence satisfactory to LMGC of the transfer of the Purchased Partnership Interest;
 - (c) the minute books, share certificate books and corporate seals of each of the Purchased Entities;
 - (d) a certificate of compliance or good standing in respect of the Vendor and each of the Purchased Entities;
 - (e) a corporate certificate of the Secretary or other officer of the Vendor in form and substance satisfactory to the Purchasers acting reasonably dealing with the due incorporation of the Vendor, the authorization by the Vendor of this Agreement and each of the Ancillary Agreements to which the Vendor is a party and the performance of its obligations hereunder and thereunder, and incumbency;
 - (f) a corporate certificate of the Secretary or other officer of each of the Purchased Entities in form and substance satisfactory to the Purchasers acting reasonably dealing with the due incorporation or formation of each such entity and incumbency;
 - (g) the resignation of, and a release from, each of the directors and officers of the Purchased Entities as contemplated under Section 5.1; and
 - (h) such other documents as are required or contemplated to be delivered by the Vendor or the Vendor's Counsel pursuant to this Agreement.
- 7.2.2 At the Closing, the Purchasers shall deliver or cause to be delivered to the Vendor the following documents and payments:
 - (a) the Purchase Price;
 - (b) a certificate of compliance or good standing in respect of each of the Purchasers;
 - (c) a corporate certificate of the Secretary or other officer of each of the Purchasers in form and substance satisfactory to the Vendor acting reasonably dealing with due incorporation of each of the Purchasers, the authorization by the Purchasers of this Agreement and each of the Ancillary Agreements and the performance of its obligations hereunder and thereunder, and incumbency; and
 - (d) such other documents as are required or contemplated to be delivered by the Purchasers or the Purchasers' Counsel pursuant to this Agreement.

60

ARTICLE 8 GENERAL MATTERS

8.1 Confidentiality

The Confidentiality Agreement shall and be of no further effect as of the Closing and the provisions of Section 5.12 shall apply.

8.2 Public Notices

No press release or other announcement concerning the transaction contemplated by this Agreement shall be made by the Vendor or by the Purchaser without the prior written consent of the other (such consent not to be unreasonably withheld) provided, however, that any party may, without such consent, make such disclosure if the same is required by any stock exchange on which any of the securities of such party or any of its Affiliates are listed or by any securities commission or other similar regulatory authority having jurisdiction over such party or any of its Affiliates, and if such disclosure is required, the party making

the disclosure shall use commercially reasonable efforts to give prior oral or written notice to the other, and if such prior notice is not possible, to give such notice immediately following the making of such disclosure.

8.3 Expenses

- 8.3.1 Each of the Vendor and the Purchaser shall be responsible for the expenses (including fees and expenses of legal advisers, accountants and other professional advisers) incurred by them, respectively, in connection with the negotiation and settlement of this Agreement and the completion of the transaction contemplated hereby. For greater certainty, the parties acknowledge and agree that the Purchased Entities shall not have any liability whatsoever for any of the expenses referred to in this Section 8.3.1.
- 8.3.2 The Vendor shall reimburse the Purchased Entities for fifty percent (50%) of all of the costs and expenses associated with the audit of the Financial Statements, provided that the aggregate costs and expenses associated with such audit are less than or equal to \$100,000. In the event that the aggregate costs and expenses associated with the audit of the Financial Statements exceeds \$100,000, the Vendor shall reimburse the Purchased Entities for such costs and expenses to the extent they exceed \$50,000.

8.4 Assignment

No party may assign its rights or benefits under this Agreement without the prior written consent of the other parties.

8.5 Notices

Any notice or other communication required or permitted to be given hereunder shall be in writing and shall be given by prepaid first-class mail, by facsimile or other means of electronic communication or by hand-delivery as hereinafter provided. Any such notice or other communication, if mailed by prepaid first-class mail at any time other than during a general discontinuance of postal service due to strike, lockout or otherwise, shall be deemed to have been received on the fourth Business Day after the post-marked date thereof, or if sent by facsimile or other means of electronic communication, shall be deemed to have been received on the Business Day following the sending, or if delivered by hand shall be deemed to have been received at the time it is delivered to the applicable address noted below either to the individual designated below or to an individual at such address having apparent authority to accept deliveries on behalf of the addressee. Notice of change of address

61

shall also be governed by this section. In the event of a general discontinuance of postal service due to strike, lock-out or otherwise, notices or other communications shall be delivered by hand or sent by facsimile or other means of electronic communication and shall be deemed to have been received in accordance with this section. Notices and other communications shall be addressed as follows:

(a) if to the Vendor:

Duke Energy Gas Transmission Canada 1333 West Georgia, Vancouver, British Columbia V6E 3K9

Attention: General Counsel Fax: (604) 488-8014

with a copy to the Vendor's Counsel at:

Torys LLP Maritime Life Tower Suite 3000, PO Box 270 Toronto-Dominion Centre Toronto, Ontario M5K 1N2

Attention: Mr. Philip Mohtadi Fax: (416) 865-7380

(b) if to the Purchaser:

ADS Alliance Data Systems, Inc. 17655 Waterview Parkway Dallas, Texas 75252

Attention: General Counsel Fax: (972) 348-5150

with a copy to the Purchasers' Counsel at:

Blake, Cassels & Graydon LLP Box 25, Commerce Court West 199 Bay Street Toronto, Ontario M5L 1A9

Attention: Mr. Frank Guarascio Fax: (416) 863-2653

The failure to send or deliver a copy of a notice to the Purchasers' Counsel or the Vendor's Counsel, as the case may be, shall not invalidate any notice given under this section.

8.6 Time of Essence

Time is of the essence of this Agreement.

62

8.7 Further Assurances

Each of the parties shall promptly do, make, execute, deliver, or cause to be done, made, executed or delivered, all such further acts, documents and things as the other party hereto may reasonably require from time to time for the purpose of giving effect to this Agreement and shall use reasonable efforts and take all such steps as may be reasonably within its power to implement to their full extent the provisions of this Agreement.

8.8 Counterparts

This Agreement may be signed in counterparts and each such counterpart shall constitute an original document and such counterparts, taken together, shall constitute one and the same instrument.

63

IN WITNESS WHEREOF the parties hereto have executed this Agreement as of the date first above written.

WESTCOAST ENERGY INC., carrying on business as DUKE ENERGY GAS TRANSMISSION

By: /s/ GREGORY L. EBEL

Name: Gregory L. Ebel Title: Vice President

By: /s/ LINDSAY A. HALL

Name: Lindsay A. Hall

Title: Vice President, Finance and Treasurer

ADS ALLIANCE DATA SYSTEMS, INC.

By: /s/ MICHAEL A. BELTZ

Name: Michael A. Beltz Title: Executive Vice President

LOYALTY MANAGEMENT GROUP CANADA INC.

By: /s/ MICHAEL KLINE

Name: Michael Kline

Title: Senior Vice President and Legal Counsel

64

QuickLinks

TABLE OF CONTENTS

PURCHASE AND SALE AGREEMENT

ARTICLE 1 INTERPRETATION

ARTICLE 2 PURCHASE AND SALE

ARTICLE 3 REPRESENTATIONS AND WARRANTIES

ARTICLE 4 CONDITIONS

ARTICLE 5 ADDITIONAL COVENANTS AND AGREEMENTS OF THE PARTIES

ARTICLE 6 INDEMNIFICATION
ARTICLE 7 CLOSING
ARTICLE 8 GENERAL MATTERS

INDEPENDENT AUDITORS' CONSENT

We consent to the incorporation by reference in Registration Statements (No. 333-65556 and 333-68134) of Alliance Data Systems Corporation on Form S-8 of our report dated September 4, 2002 on the combined financial statements of Enlogix Group as of and for the years ending December 31, 2000 and 2001, appearing in this Current Report on Form 8-K of Alliance Data Systems Corporation dated September 5, 2002.

/s/ Deloitte & Touche LLP Deloitte & Touche LLP

Toronto, Canada September 5, 2002

QuickLinks

INDEPENDENT AUDITORS' CONSENT

INDEPENDENT AUDITORS' REPORT

To the Board of Directors:

We have audited the accompanying combined balance sheets of Enlogix Inc. and Enlogix CIS L.P. and Intralynx L.P. (collectively "Enlogix"), all of which are under common ownership and common management, as of December 31, 2001 and 2000, and the related combined statements of loss, equity interests and of cash flows for the years then ended. These financial statements are the responsibility of Enlogix' management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with both Canadian generally accepted auditing standards and auditing standards generally accepted in the United States of America. Those standards require that we plan and perform an audit to obtain reasonable assurance whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such financial statements present fairly, in all material respects, the combined financial position of Enlogix as of December 31, 2001 and 2000 and the combined results of its operations and its combined cash flows for the years then ended in conformity with Canadian generally accepted accounting principles.

/s/ Deloitte & Touche LLP Deloitte & Touche LLP Toronto, Canada September 4, 2002

Comments by Auditor on Canada-United States of America Reporting Difference

In the United States of America, reporting standards for auditors require the addition of an explanatory paragraph when the financial statements are affected by conditions and events that cast substantial doubt on the entity's ability to continue as a going concern, such as those described in Note 1 to the financial statements. Although we conducted our audits in accordance with both Canadian generally accepted auditing standards and auditing standards generally accepted in the United States of America, our report to the Board of Directors dated September 4, 2002 is expressed in accordance with Canadian reporting standards which do not permit a reference to such conditions and events in the auditors' report when these are adequately disclosed in the financial statements.

F-1

ENLOGIX GROUP COMBINED BALANCE SHEETS (CANADIAN DOLLARS)

		December 31,			
		2000		2001	
ASSETS					
Cash and cash equivalents	\$	6,240,088	\$	2,624,946	
Accounts receivable, net of allowance for doubtful accounts of \$438,000 and \$nil,					
respectively		1,754,843		3,334,509	
Due from related parties (Note 3)		3,141,486		3,899,180	
Prepaids and other assets		24,306		137,890	
Sublicensing rights (Note 4)		865,000		1,064,000	
Total current assets		12,025,723		11,060,525	
Fixed assets, net (Note 5)		30,685,361		32,019,589	
Sublicensing rights (Note 4)		7,549,837		7,350,837	
Total assets	\$	50,260,921	\$	50,430,951	
LIABILITIES AND EQUITY INTERE	STS				
Accounts payable and accrued liabilities	\$	16,018,014	\$	8,852,908	
Due to related parties (Note 3)		2,587,778		2,570,641	
Deferred revenue		_		752,684	
Current portion of long-term obligations (Note 6)		879,667		3,590,150	
Total current liabilities		19,485,459		15,766,383	
Long-term obligations (Note 6)		2,357,274		6,395,803	
Long-term employee benefit obligation (Note 10)		31,000		233,000	
Total liabilities		21,873,733		22,395,186	

Commitments and contingencies (Notes 1, 13 and 15)			
Equity interests:			
Contributed capital		97,526,131	112,221,158
Deficit		(69,138,943)	(84,191,416)
Cumulative translation adjustment		_	6,023
Total equity interests		28,387,188	28,035,765
Total liabilities and equity interests	\$	50,260,921 \$	50,430,951
Total national and equity mercoto	Ψ	30,200,821 ψ	23, 180,881

See accompanying notes

F-2

ENLOGIX GROUP COMBINED STATEMENTS OF LOSS (CANADIAN DOLLARS)

	For the years end	For the years ended December 31			
	2000	2001			
Service revenue (Note 3)	\$ 26,702,006	\$ 29,282,028			
Operating expenses					
Operating, selling and general	43,784,853	35,345,369			
Depreciation and amortization	8,549,910	7,924,650			
Total operating expenses	52,334,763	43,270,019			
Operating loss	(25,632,757)	(13,987,991)			
Foreign exchange gain (loss)	(63,750)	80,786			
	(25,696,507)	(13,907,205)			
Interest expense:					
Short-term	410,081	21,387			
Long-term	290,497	1,122,523			
Loss before income taxes	(26,397,085)	(15,051,115)			
Income tax expense—current (Note 7)	1,298	1,358			
Net loss	\$ (26,398,383)	\$ (15,052,473)			

See accompanying notes

F-3

ENLOGIX GROUP COMBINED STATEMENTS OF EQUITY INTERESTS FOR THE YEARS ENDED DECEMBER 31, 2000 and 2001 (CANADIAN DOLLARS)

Balance—December 31, 1999	\$	19,647,538
Datable December 32, 1883	•	15,6 17,550
Contributions		35,138,033
Net loss		(26,398,383)
Balance—December 31, 2000		28,387,188
Contributions		14,695,027
Net loss		(15,052,473)

Foreign currency translation	6,023
Balance—December 31, 2001	\$ 28,035,765

See accompanying notes

F-4

ENLOGIX GROUP COMBINED STATEMENTS OF CASH FLOWS (CANADIAN DOLLARS)

	For the years ended December 31			
	2000		2001	
Operating activities				
Net loss	\$ (26,398,383)	\$	(15,052,473)	
Add (deduct) items to reconcile to net cash from operating activities:				
Depreciation and amortization	8,549,910		7,924,650	
Other operating	125,530		970,684	
Non-cash working capital changes (Note 8)	(8,986,240)		(7,674,819)	
Net cash used by operating activities	(26,709,183)		(13,831,958)	
Investing activities				
Additions to fixed assets	(440,124)		(1,024,603)	
Proceeds on disposal of fixed assets	_		21,000	
		_		
Net cash used by investing activities	(440,124)		(1,003,603)	
Financing activities				
Decrease in bank indebtedness	(1,485,621)		_	
Payments on long-term obligations	(463,364)		(2,474,608)	
Partnership contributions	35,138,033		13,695,027	
		_		
Net cash provided by financing activities	33,189,048		11,220,419	
Increase (decrease) in cash & cash equivalents	6,039,741		(3,615,142)	
Cash & cash equivalents, beginning of year	200,347		6,240,088	
		_		
Cash & cash equivalents, end of year	\$ 6,240,088	\$	2,624,946	

See accompanying notes

F-5

ENLOGIX GROUP NOTES TO COMBINED FINANCIAL STATEMENTS DECEMBER 31, 2000 AND 2001 (CANADIAN DOLLARS)

1. BASIS OF PRESENTATION

These financial statements reflect the operations of Enlogix Inc., its wholly owned subsidiaries Enlogix CIS Inc., Enlogix USA Inc. and Intralynx Inc., and two limited partnerships, Enlogix CIS L.P. and Intralynx L.P (collectively "Enlogix"). Enlogix CIS L.P. and Intralynx L.P. (the "limited partnerships") are owned approximately 1% by Enlogix Inc. and 99%, directly and indirectly, by Westcoast Energy Inc. ("Westcoast"). In March 2002, a reorganization of Enlogix resulted in the transfer of assets to and assumption of liabilities of Intralynx L.P. by Enlogix CIS L.P., whereupon Intralynx L.P. was wound up. Subsequently, a vertical short form amalgamation of Intralynx Inc., Enlogix Inc. and Enlogix CIS Inc. was completed with the amalgamated company continuing as Enlogix Inc. Enlogix Inc. is a wholly owned subsidiary of Westcoast. Because a direct ownership relationship does not exist among the various business units comprising Enlogix, equity interest in Enlogix is shown in lieu of shareholder's equity in the combined financial statements.

Enlogix was principally conceived from an internal project within Westcoast in 1996. The initial project was eventually expanded resulting in the formation of Enlogix in 1998. Enlogix is a North American provider of billing and customer information services to the energy and utility sectors, providing 13 clients with billing services for their approximately 3 million customers, approximately 1.2 million of which are customers of affiliated companies. Enlogix' system supports clients in the regulated and unregulated gas, electric, water, and energy services sectors across North America.

While the financial statements have been prepared on the basis of accounting principles applicable to a going concern, the significant operating losses incurred since inception and the working capital deficiency as at December 31, 2001 cast substantial doubt upon the validity of this assumption. The ability of

Enlogix to continue operations is dependent on the continued financial support of its shareholder or other source of financing, until such time as the operations and business of Enlogix become self-sustaining. The success of Enlogix is ultimately dependent on its ability to market commercially viable business solutions in the billing and customer information services market. Management is of the opinion that sufficient working capital will be obtained from operations and external financing to execute its business plan, and meet its liabilities and commitments as they become due. If the going concern assumption were not appropriate for these financial statements, then adjustments would be necessary in the carrying values of assets and liabilities, the reported net loss and the balance sheet classifications used.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The financial statements of Enlogix have been prepared in accordance with Canadian generally accepted accounting principles using accounting principles applicable to a going concern, which assumes that Enlogix will continue in operation for a reasonable period of time and will be able to realize its assets and discharge its liabilities in the normal course of operations.

Principles of Combination

The accompanying combined financial statements of Enlogix include Enlogix Inc. and its wholly owned subsidiaries and the assets and liabilities of the limited partnerships. All material intergroup transactions and balances have been eliminated.

F-6

The accompanying combined financial statements reflect the operations of Enlogix as at and for the years ended December 31, 2001 and 2000 and are presented in Canadian dollars.

Revenue Recognition

Revenues are primarily derived from the following sources: customer billing and bill presentment/remittance processing, custom modifications and project implementation fees. Enlogix' services are generally sold as part of a contract and the terms of the contract, taken as a whole, determine the appropriate revenue recognition method. Enlogix recognizes revenue when persuasive evidence of an arrangement exists, when the service is performed in accordance with all terms and conditions of customer contracts, when the fee is fixed or determinable, and when collection is reasonably assured.

Customer billing and bill presentment/remittance processing service revenues are recognized over the life of the contract as services are provided.

Custom modification revenue is recognized when the service has been rendered.

In instances where project implementation fees are collected in advance, Enlogix will defer and amortize the advance payment over the life of the contract. Any incremental expense to provide the implementation service is also recognized over the life of the contract.

Income Taxes

Enlogix provides for income taxes using the liability method of tax allocation. Under this method, deferred income tax assets and liabilities are determined based on differences between the financial reporting and tax bases of assets and liabilities, and measured using the substantively enacted tax rates and laws that will be in effect when the differences are expected to reverse.

A significant amount of Enlogix' operations are conducted within the limited partnerships, which are taxed at the partner level. The ownership of the limited partnerships resides largely outside of Enlogix' ownership structure and, accordingly, the results of these operations are not tax effected within these combined financial statements.

Employee Benefit Plans

For pension and post-retirement benefits other than pension, Enlogix recognizes the liability and expense for the period when the benefits are earned.

Enlogix accrues its obligations under employee benefit plans and the related costs, net of plan assets. The plan assets are valued at fair value. The calculation of the expected return on assets is based on the market-related value of assets.

Past service costs from plan amendments are amortized on a straight-line basis over the average remaining service period of employees active at the date of amendment.

The excess of the net unamortized cumulative actuarial gain or loss over 10 per cent of the greater of the benefit obligation and the market-related value of plan assets at the beginning of the year is amortized over the average remaining service period of the active employees.

F-7

For defined contribution plans maintained by Enlogix, contributions payable by Enlogix are expensed as pension costs.

Cash and Cash Equivalents

Cash and cash equivalents consist of cash on hand and balances with banks.

Sublicensing Rights

Sublicensing rights are valued at the lower of cost and net realizable value. Sublicensing rights are relieved in accordance with the contract with the software supplier from whom the sublicensing rights were purchased.

Fixed Assets

Fixed assets are recorded at cost. Interest costs incurred during development are capitalized as part of the cost of the asset.

Furniture, fixtures, computer equipment, software and leasehold improvements are amortized on a straight-line basis using estimated lives ranging from 3 to 6 years.

Costs of software development, including enhancements and custom modifications to existing products for resale, are expensed as incurred until the technological feasibility of the software product has been established. Thereafter, software development costs are capitalized as incurred and reported at the lower of unamortized cost and net realizable value. Costs of software development are amortized on a straight-line basis over an 8-year period based on length of current contracts or benefit period.

Foreign Currency

Enlogix USA Inc. primarily operates and maintains its accounts in United States dollars. This business is operationally and functionally self-sustaining and accordingly, the assets and liabilities are translated into Canadian dollars at the year-end exchange rate, and revenues and expenses are translated into Canadian dollars at the average exchange rate for the year. The resulting unrealized cumulative translation gains or losses are deferred as a separate component of equity interests.

Other current assets and liabilities denominated in United States dollars have been translated into Canadian dollars at the year-end exchange rate. Any resulting gain or loss is reflected in income.

Deferred Revenue

Deferred revenue consists of amounts billed to customers for which services will be performed over future periods.

F-8

Use of Estimates

The preparation of financial statements in conformity with Canadian generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, expenses and disclosure of contingent assets and liabilities. These estimates include the allowance for uncollectible accounts receivable, valuation of sublicensing rights and asset valuations. The recorded amounts for such items are based on management's best information and judgment and, accordingly, actual results could differ from those estimates.

3. RELATED PARTY TRANSACTIONS

Enlogix transactions with its parent company and other companies related through common control are as follows:

	For the years ended December 31			
	2000			2001
Westcoast Energy Inc.				
Administration fees and corporate charges paid	\$	83,932	\$	391,435
Union Gas Limited				
Customer billing charges received		7,678,570		11,246,028
Rental lease payments		1,099,606		1,241,862
Administration fees paid		641,812		359,069
Centra Gas British Columbia Inc.				
Customer billing charges received		854,405		895,995
Bill presentment/remittance processing charges received		499,484		589,732
Pacific Northern Gas Ltd.				
Customer billing charges received		524,118		327,402
Bill presentment/remittance processing charges received		174,192		252,691
Union Energy Inc. (to October 31, 2001)				
Customer billing charges received		3,168,635		2,747,898
Bill presentment charges received		4,455,804		2,280,675

These transactions are in the normal course of operations and are recorded at exchange amounts established and agreed between the related parties.

		2000	2001	
ties				
	\$	7,516	\$ 5,596	
		48,784	2,433,371	
с.			1,000,000	
tish Columbia Inc.		308,417	342,966	
ern Gas Ltd.		217,878	117,247	
		2,558,891		
	\$	3,141,486	\$ 3,899,180	
nergy Inc. imited	\$	2,360,087	\$ 2,467,990	
		227,691	102,651	
	\$	2,587,778	\$ 2,570,641	

These amounts arise in the normal course of operations and are subject to normal credit terms.

4. SUBLICENSING RIGHTS

Sublicensing rights were purchased from a third party software supplier in 1997 and 2000. In December 2000, Enlogix entered into an agreement with the software supplier for the exclusive rights to sell the sublicensing rights in Canada for a five-year term and on a non-exclusive basis thereafter for an additional five years.

F-10

5. FIXED ASSETS

		December 31			
		2000		2001	
sets					
are	\$	11,465	\$	4,439,608	
rare		7,906,344		1,815,415	
e development		26,051,686		26,051,686	
under capital lease		1,461,350		9,921,481	
		964,078		1,633,973	
		36,394,923		43,862,163	
ulated depreciation re		3,830		2,748,720	
vare		2,420,739		539,268	
development		3,169,404		7,233,902	
under capital lease		68,614		1,106,864	
ements		46,975		213,820	
		5,709,562		11,842,574	
	\$	30,685,361	\$	32,019,589	
	_				

Software development costs consist of the Banner Customer Information Service ("CIS") System, the recovery of which is uncertain. In accordance with current accounting standards, management uses estimated expected future net cash flows to measure the recoverability of its investment in these assets. The estimation of expected future net cash flows is inherently uncertain and relies to a considerable extent on assumptions regarding current and future economic market conditions. If, in future periods, there are changes in the estimates or assumptions incorporated into the impairment review analysis, the changes could result in an adjustment to the carrying amount of the Banner CIS System.

6. LONG-TERM OBLIGATIONS

In June 2000, Enlogix entered into a three-year capital lease arrangement with IBM Canada Ltd. ("IBM") for hardware and equipment. The monthly lease payments were \$43,894 paid in advance. In June 2001, Enlogix terminated its existing capital lease arrangement and entered into a new four-year capital lease with IBM, expiring in 2005, for hardware equipment providing an increase in system capacity and performance. The quarterly lease payments are \$805,622 paid in advance with an imputed discount rate of 15%.

Enlogix is also obligated under a long-term commitment extending to July 2005. Monthly payments of \$50,000 are payable in arrears and bear an imputed discount rate of 15%.

The following is a schedule of future minimum payments under the long-term obligations together with the balance outstanding:

2002	\$ 3,822,490
2003	3,822,490
2004	3,822,490
2005	1,155,623
Total minimum payments	12,623,093
Less: Amount representing interest	(2,637,140)
Balance of obligation	9,985,953
Less: Current portion	(3,590,150)
Long-term portion	\$ 6,395,803

7. INCOME TAXES

At December 31, 2001, Enlogix has loss carryforwards of \$694,777 for income tax purposes that expire as follows: 2003—\$28,497; 2004—\$362,938; and thereafter—\$303,342. For financial reporting purposes a deferred tax asset has not been recognized in respect of these loss carryforwards as the criteria for recognition has not been met.

8. SUPPLEMENTAL CASH FLOW INFORMATION

Non-cash working capital changes:

	 For the years ended December 31				
	2000		2001		
Accounts receivable	\$ 605,511	\$	(1,579,666)		
Sublicensing rights and prepaids	(6,441,805)		(113,584)		
Accounts payable and accrued liabilities	6,418,739		(6,206,738)		
Due from/to related parties	(9,568,685)		225,169		
Attributable to operating activities	\$ (8,986,240)	\$	(7,674,819)		

Interest and tax payments:

	I	For the years en	led De	cember 31
		2000		2001
	\$	424,368	\$	691,387
				_

F-12

9. STOCK-BASED COMPENSATION

Stock Option Plan

Under its Long Term Incentive Share Option Plan 1989 ("1989 Plan"), Westcoast has granted regular, key employee retention, and performance-based stock options to certain executives of Enlogix. Stock options are granted at an exercise price that equals the market price as defined in the 1989 Plan of Westcoast's shares on the date of grant.

The 1989 Plan also provides for share appreciation rights under which the holder of a stock option may, in lieu of exercising the option, exercise the share appreciation right. Enlogix does not record any expense for these rights.

At December 31, 2001, 98,240 common shares of Westcoast were under option at prices ranging from \$20.48 to \$34.45 per share to certain executives of Enlogix, of which 95,340 are eligible for share appreciation rights that allow the holder to receive 50% of the appreciated value in cash and the balance in common shares of Westcoast.

10. EMPLOYEE BENEFITS

Accrued benefit asset (liability)

Enlogix has defined benefit pension plans, defined contribution pension plans and defined benefit plans providing retirement and post-employment health and life insurance benefits for most employees. Information about the defined benefit plans, in aggregate, is as follows:

	For the years ended December 31									
		Pension B	enefit Plans		Other Be	lans				
		2000	2001		2000		2001			
Accrued benefit obligations										
Balance, beginning of year	\$	1,826,000	\$ 2,057,0	00	\$ 280,000	\$	332,000			
Current service cost		99,000	194,0	00	30,000		32,000			
Interest cost		132,000	169,0	00	22,000		25,000			
Prior service costs		_	20,0	00	_		_			
Actuarial loss (gain)		_	101,0	00			(69,000)			
Balance, end of year		2,057,000	2,541,0	00	332,000		320,000			
				_						
Plan assets										
Fair value, beginning of year		1,360,000	1,587,0	00	_		_			
Actual return on plan assets		41,000	(605,0	00)	_					
Employer contributions		173,000	162,0	00	_		_			
Employees' contributions		13,000	14,0	00						
Fair value, end of year		1,587,000	1,158,0	00	_		_			
				_						
Funded status—plan deficit		(470,000)	(1,383,0		(332,000)		(320,000)			
Unamortized net actuarial loss (gain)		68,000	653,0		_		(69,000)			
Unamortized prior service costs		_	20,0		_					
Unamortized transitional obligation		439,000	609,0	00	264,000		248,000			
Contributions remitted after measurement date		_	-		_		9,000			

The non-pension defined benefit plans are unfunded. The accrued benefit obligations and fair value of plan assets of defined benefit pension plans with accrued benefit obligations in excess of plan assets are \$1,963,000 and \$575,000, respectively (December 31, 2000—\$2,057,000 and \$1,587,000, respectively).

F-14

37,000

The following is a summary of the weighted average significant actuarial assumptions used in measuring Enlogix' accrued benefit obligations:

	For t	For the years ended December 31						
	Pension Bene	efit Plans	Other Benefit Plans					
	2000	2001	2000	2001				
Discount rate	7%	7.25%	7%	7.25%				
Expected long-term rate of return on plan assets	7.5	8.5	_	_				
Rate of compensation increase	3.25	3.25	3.25	3.25				

In addition, in determining the expected cost of health care benefit plans, it is assumed that the health care inflation will decrease gradually to 5% in 2003 and remain level thereafter.

Enlogix' net benefit plan expense for defined benefit plans is as follows:

For the years	ended December	31
---------------	----------------	----

(101,000) \$

(68,000) \$

(132,000)

	Pension Benefit Plans			Pension Benefit Plans Other Bene			ans		
	2000 2001 2000		2000 2001		2000		2000 20		2001
rrent service cost	\$ 87,000	\$	180,000	\$	30,000	\$	32,000		
erest cost	132,000		169,000		22,000		25,000		
pected return on plan assets	(109,000)		(131,000)		_		_		

Amortization of transitional obligation	27,000	46,000	16,000	16,000
Net benefit plan expense	\$ 137,000	\$ 264,000	\$ 68,000	\$ 73,000

The defined contribution pension expense for the years ended December 31, 2000 and 2001 was \$200,000 and \$0, respectively.

11. RECONCILIATION OF GENERALLY ACCEPTED ACCOUNTING PRINCIPLES

Enlogix prepares its accounts in accordance with Canadian generally accepted accounting principles (Canadian GAAP), which in the main parallel United States generally accepted accounting principles (US GAAP). The following tables reflect the major differences in accounting principles.

Combined net loss under US GAAP would be:

		For the years ended December 31					
		2000		2001			
Net loss under Canadian GAAP	\$	(26,398,383)	\$	(15,052,473)			
Adjustments							
Pension and post-retirement benefits other than pensions		(60,000)		(65,000)			
Stock-based compensation		(610,926)		(393,080)			
	_		_				
Net loss under US GAAP	\$	(27,069,309)	\$	(15,510,553)			

F-15

Combined comprehensive loss under US GAAP would be:

		For the years ended December 31					
		2000		2001			
Net loss under US GAAP	\$	(27,069,309)	\$	(15,510,553)			
Change in the Cumulative Translation Adjustment		_		6,023			
	_						
Comprehensive loss under US GAAP	\$	(27,069,309)	\$	(15,504,530)			
	_						

Under Canadian GAAP, a combined statement of comprehensive translation adjustment is recorded as a separate component of equity interests.

 $After \ adjusting \ for \ certain \ differences, selected \ combined \ balance \ sheet \ items \ under \ US \ GAAP \ would \ be:$

	December 31					
		2000		2001		
Long-term obligations (representing pension and post-retirement benefits other than						
pension and share appreciation rights)	\$	821,926	\$	1,482,006		
Accumulated other comprehensive income		_		6,023		
Deficit		(69,929,869)		(85,440,422)		

Pensions and Post-Retirement Benefits Other than Pensions

Effective January 1, 2000, accounting for pension and post-retirement benefits other than pension under Canadian GAAP was modified and is similar to the requirements under US GAAP. Pension fund assets are measured at current market values and the accrued pension plan obligations are discounted using current interest rates. Post-retirement benefits other than pension are recorded on an accrual basis. The reconciliation between Canadian GAAP and US GAAP relates to the difference in the amortization of the net transition obligation for Canadian GAAP and US GAAP purposes, respectively.

Income Taxes

Canadian GAAP requires accounting for income taxes using the liability method of tax allocation, similar to the requirements under US GAAP. However, the only significant difference is that under Canadian GAAP, substantively enacted tax rates are used to measure deferred tax assets and liabilities whereas under US GAAP, current legislated tax rates are used to measure the deferred tax assets and liabilities. The impact on deferred income taxes under US GAAP is negligible.

Westcoast has one stock-based compensation plan in which certain Enlogix executives participate. Under Canadian GAAP, no compensation expense is recognized in the results of Enlogix for this plan when the stock options are issued. In accordance with US GAAP, stock-based payments to employees of a subsidiary based on the parent's stock are accounted for in the results of the subsidiary using either Accounting Principles Board Opinion No. 25 ("APB 25"), Accounting for Stock Issued to Employees, or Statement of Financial Accounting Standards No. 123 ("SFAS 123"), Accounting for Stock-Based Compensation. As permitted by US GAAP, Enlogix has elected to follow APB 25 and accordingly has not recognized compensation expense for its participation in the Westcoast stock option plan. Share appreciation rights (SARs) and similar awards to be settled in cash are accounted for by measuring the amount by which the quoted market price exceeds the option price at the balance sheet date. Under Canadian GAAP, no compensation expense is recognized for SARs.

12. FINANCIAL INSTRUMENTS

The carrying values of financial assets recorded on the balance sheets of Enlogix, including accounts receivable, current and long-term liabilities and amounts due to and from related parties, approximate their fair values. Enlogix has not entered into derivative instruments for trading or hedging purposes.

13. COMMITMENTS

Enlogix has entered into leases extending to 2006 for certain office facilities, equipment and vehicles under noncancellable operating leases. Future annual minimum rental payments required under noncancellable operating leases, some of which contain renewal options, as of December 31, 2001 are:

2002	\$ 1,255,677
2003	1,255,677
2004	1,244,164
2005	1,241,862
2006	1,138,368

Operating lease payments for the years ended December 31, 2000 and 2001 were \$1,455,200 and \$1,691,624, respectively.

14. SEGMENTS

Enlogix operates in one segment being the design, development and sale of billing and customer information services to the energy and utility sectors. Substantially all of Enlogix' revenue is derived from sales to customers in Canada. Substantially all assets of Enlogix are also located in Canada. In 2001, one customer constituted approximately 38% of combined revenue (2000—approximately 29%).

F-17

15. CONTINGENCIES

In the normal course of business, Enlogix is subject to proceedings, lawsuits and other claims. In the opinion of management, these matters will not have a material impact on the combined financial position or results of operations.

16. SUBSEQUENT EVENTS

In September 2002, Westcoast entered into an agreement to sell Enlogix for cash proceeds of approximately US\$13.5 million, subject to post-closing adjustments.

F-18

QuickLinks

INDEPENDENT AUDITORS' REPORT

ENLOGIX GROUP COMBINED BALANCE SHEETS (CANADIAN DOLLARS)

ENLOGIX GROUP COMBINED STATEMENTS OF LOSS (CANADIAN DOLLARS)

ENLOGIX GROUP COMBINED STATEMENTS OF EQUITY INTERESTS FOR THE YEARS ENDED DECEMBER 31, 2000 and 2001 (CANADIAN DOLLARS)

ENLOGIX GROUP COMBINED STATEMENTS OF CASH FLOWS (CANADIAN DOLLARS)

ENLOGIX GROUP NOTES TO COMBINED FINANCIAL STATEMENTS DECEMBER 31, 2000 AND 2001 (CANADIAN DOLLARS)

UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL INFORMATION

The following unaudited pro forma consolidated financial information is based on the historical financial statements of Alliance Data Systems Corporation ("ADSC") and Enlogix Group ("Enlogix"). The unaudited pro forma adjustments are based upon certain assumptions that we believe are reasonable. The unaudited pro forma consolidated financial information and accompanying notes should be read in conjunction with the historical financial statements of Alliance Data Systems Corporation and Enlogix Group, and the respective notes to those statements.

The pro forma adjustments give effect to the acquisition of Enlogix Group, accounted for under the purchase method of accounting, as if that acquisition had been consummated on January 1, 2001.

The unaudited pro forma consolidated financial information does not purport to be indicative of the results that would have been obtained had the transactions been completed as of the assumed dates and for the periods presented or that may be obtained in the future.

1

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS (in thousands, except per share data)

For the year ended December 31, 2001

ADSC	1	Enlogix(1)		stments	I	Pro Forma		
\$ 777,351	\$	18,887	\$	<u> </u>	\$	796,238		
603,493		22,798				626,291		
45,431						45,431		
30,698		5,111		(3,671)(2)		32,138		
43,506		_		1,395 (3)		44,901		
	_							
723,128		27,909		(2,276)		748,761		
54,223		(9,022)		2,276		47,477		
5,000		52				5,052		
15,131		_				15,131		
30,097		738				30,835		
11,612		1		—(4)		11,613		
\$ (7,617)	\$	(9,813)	\$	2,276	\$	(15,154)		
\$ (0.17)					\$	(0.29)		
64,555						64,555		
,								
2								
\$	\$ 777,351 603,493 45,431 30,698 43,506 723,128 54,223 5,000 15,131 30,097 11,612 \$ (7,617) \$ (0.17)	\$ 777,351 \$ 603,493 45,431 30,698 43,506 723,128 54,223 5,000 15,131 30,097 11,612 \$ (7,617) \$ \$ (0.17)	ADSC Enlogix(1) \$ 777,351 \$ 18,887 603,493 22,798 45,431 30,698 5,111 43,506 — 723,128 27,909 54,223 (9,022) 5,000 52 15,131 — 30,097 738 11,612 1 \$ (7,617) \$ (9,813) \$ (0.17)	ADSC Enlogix(1) Adju \$ 777,351 \$ 18,887 \$ 603,493 22,798 45,431 30,698 5,111 43,506 — 723,128 27,909 54,223 (9,022) 5,000 52 15,131 — 30,097 738 11,612 1 \$ (7,617) \$ (9,813) \$ \$ (0.17)	ADSC Enlogix(1) Adjustments \$ 777,351 \$ 18,887 \$ — 603,493 22,798 45,431 30,698 5,111 (3,671)(2) 43,506 — 1,395 (3) 723,128 27,909 (2,276) 54,223 (9,022) 2,276 5,000 52 15,131 — 30,097 738 11,612 1 — (4) \$ (7,617) \$ (9,813) \$ 2,276 \$ (0.17)	ADSC Enlogix(1) Adjustments F \$ 777,351 \$ 18,887 \$ — \$ 603,493 22,798 45,431 30,698 5,111 (3,671)(2) 43,506 — 1,395 (3) — 723,128 27,909 (2,276) — 54,223 (9,022) 2,276 5,000 52 — — 30,097 738 — — 43,506 —		

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS (in thousands, except per share data)

For the six months ended June 30, 2002

\$	Pro Forma 425,154
\$	425 154
	423,134
	332,076
	24,898
	19,886
	13,895
_	
	390,755
	34,399
)	

			(0)		(0)
Other non-operating (income) expense			(6)		(6)
Fair value loss on interest rate derivative		5,260	_		5,260
Interest expense		11,294	441		11,735
Income tax expense (benefit)		7,998	(6)	513 (4)	8,505
Income (loss) from continuing operations	\$	8,393	\$ (365)	\$ 877	\$ 8,905
Income per share from continuing operations					
basic	\$	0.11			\$ 0.12
diluted	\$	0.11			\$ 0.12
Weighted average shares					
basic		74,040			74,040
	_				
diluted		76,690			76,690
	_				
		3			

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED BALANCE SHEET (in thousands)

As of June 30, 2002

		1.15 of Julie 30, 2002						
	ADSC		Enlogix(1)		Adjustments(6)		Pro Forma	
Cash and cash equivalents	\$	80,995	\$	896	\$	(13,500)	\$	68,391
Due from card associations		118,868						118,868
Trade receivables, net		86,816		4,770				91,586
Credit card receivables and seller's interest,								
net		131,354						131,354
Other current assets		89,827		663				90,490
Total current assets		507,860		6,329		(13,500)		500,689
Redemption settlement assets, restricted		162,411						162,411
Property and equipment, net		116,758		19,554		(12,354)(2)		123,958
Other non-current assets		43,270		4,551		(4,551)(5)		43,270
Due from securitizations		177,640		,		())(-)		177,640
Goodwill, net		432,107				7,319 (3)		439,426
Other intangible assets		82,152				5,000 (3)		87,152
Total assets	\$	1,522,198	\$	30,434	\$	(18,086)	\$	1,534,546
	_						_	
Accounts payable	\$	145,970	\$	6,398	\$		\$	152,368
Merchant settlement obligations		154,638						154,638
Other liabilities		32,568						32,568
Debt, current portion		153,826		2,521				156,347
Total current liabilities		487,002		8,919				495,921
Other liabilities		17,495		129				17,624
Deferred revenue		350,859						350,859
Long-term and subordinated debt		145,800		3,300				149,100
Total liabilities		1,001,156		12,348				1,013,504
Total stockholders' equity		521,042		18,086		(18,086)		521,042
Total liabilities and stockholders' equity	<u> </u>	1,522,198	\$	30,434	\$	(18,086)	\$	1,534,546

NOTES TO UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS

The Unaudited Pro Forma Consolidated Statements of Operations for the year ended December 31, 2001 and the six months ended June 30, 2002 and the Unaudited Pro Forma Balance Sheet as of June 30, 2002 reflect the following adjustments:

- (1) Represents operating activity for Enlogix Group for the year ended December 31, 2001 and the six months ended and as of June 30, 2002. Prior to its acquisition, Enlogix Group had no significant transactions other than normal operations.
- (2) Represents pro forma adjustments to property and equipment, net and related depreciation and amortization expense to reduce property and equipment to its estimated fair value resulting from the preliminary purchase accounting of the Enlogix Group acquisition.
- (3) Represents pro forma adjustments to goodwill and other purchased intangibles and related amortization resulting from the preliminary purchase accounting of the Enlogix Group acquisition. The preliminary amortization periods used were five years for purchased intangibles and 20 years for goodwill prior to the Company's adoption of SFAS 142 (for the year ended December 31, 2001 only).
- (4) Realization of any tax benefit arising from Enlogix Group losses for the year ended December 31, 2001 is uncertain. Pro forma tax expense for the six months ended June 30, 2002 reflects Canadian tax expense on the Enlogix Group incremental income.
- (5) Represents pro forma adjustments to other non-current assets to reduce such asset to its estimated fair value resulting from the preliminary purchase accounting of the Enlogix Group acquisition.
- (6) The Company's preliminary purchase allocation for its acquisition of Enlogix Group, subject to post-acquisition due diligence, is as follows:

Cash consideration	\$	13,500
Net tangible asssets acquired		(1,181)
Customer contracts acquired		(5,000)
Excess purchase price allocated to goodwill	\$	7,319

5

QuickLinks

UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL INFORMATION

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS (in thousands, except per share data)

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED STATEMENT OF OPERATIONS (in thousands, except per share data)

ALLIANCE DATA SYSTEMS CORPORATION UNAUDITED PRO FORMA CONSOLIDATED BALANCE SHEET (in thousands)

ALLIANCE DATA SYSTEMS CORPORATION NOTES TO UNAUDITED PRO FORMA CONSOLIDATED FINANCIAL STATEMENTS